As filed with the Securities and Exchange Commission on August 14, 2000.

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

(Mark One)

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 or 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2000

OF

[] TRANSITION REPORT PURSUANT TO SECTION 13 or 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

Commission File No. 0-15279

GENERAL COMMUNICATION, INC. (Exact name of registrant as specified in its charter)

STATE OF ALASKA
(State or other jurisdiction of incorporation or organization)

2550 Denali Street

92-0072737 (I.R.S. Employer Identification No.)

Suite 1000
Anchorage, Alaska
(Address of principal executive offices)

99503 (Zip Code)

Registrant's telephone number, including area code: (907) 265-5600

Former name, former address and former fiscal year, if changed since last report $\ensuremath{\mathsf{P}}$

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes X No .

The number of shares outstanding of the registrant's classes of common stock, as of July 31, 2000 was:

48,090,828 shares of Class A common stock; and 3,908,148 shares of Class B common stock.

1

<TABLE>

INDEX

GENERAL COMMUNICATION, INC.

FORM 10-Q

FOR THE SIX MONTHS ENDED JUNE 30, 2000

<CAPTION>

			Page No.
<s> Caution</s>	ary Statemer	nt Regarding Forward-Looking Statements	<c></c>
PART I.	FINANCIAL	INFORMATION	
	Item l.	Consolidated Balance Sheets as of June 30, 2000 (unaudited) and December 31, 1999	5
		Consolidated Statements of Operations for the three and six months ended June 30, 2000 (unaudited) and 1999 (unaudited)	7
		Consolidated Statements of Stockholders' Equity for the six months ended June 30, 2000 (unaudited) and 1999 (unaudited)	8

		Consolidated Statements of Cash Flows for the six months ended June 30, 2000 (unaudited) and 1999 (unaudited)9
		Notes to Interim Condensed Consolidated Financial Statements
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations
	Item 3.	Quantitative and Qualitative Disclosures About Market Risk
ART II.	OTHER INF	ORMATION
	Item 1.	Legal Proceedings
	Item 4.	Submission of Matters to a Vote of Securities Holders
	Item 6.	Exhibits and Reports on Form 8-K
	Other item	s are omitted, as they are not applicable.
IGNATUR		35

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

You should carefully review the information contained in this Quarterly Report, but should particularly consider any risk factors that we set forth in this Quarterly Report and in other reports or documents that we file from time to time with the Securities and Exchange Commission. In this Quarterly Report, in addition to historical information, we state our beliefs of future events and of our future operating results, financial position and cash flows. In some cases, you can identify those so-called "forward-looking statements" by words such as "may," "will," "should," "expects," "plans," "anticipates," "believes," "estimates," "predicts," "potential," or "continue" or the negative of those words and other comparable words. You should be aware that those statements are only our predictions and are subject to risks and uncertainties. Actual events or results may differ materially. In evaluating those statements, you should specifically consider various factors, including those outlined below. Those factors may cause our actual results to differ materially from any of our forward-looking statements. For these statements, we claim the protection of the safe harbor for forward-looking statements provided by the Private Securities Litigation Reform Act of 1995.

- Material adverse changes in the economic conditions in the markets we serve;
- The efficacy of the rules and regulations to be adopted by the Federal Communications Commission ("FCC") and state public regulatory agencies to implement the provisions of the Telecommunications Act of 1996; the outcome of litigation relative thereto; and the impact of regulatory changes relating to access reform;
- Our responses to competitive products, services and pricing, including pricing pressures, technological developments, alternative routing developments, and the ability to offer combined service packages that include local, cable and Internet services; the extent and pace at which different competitive environments develop for each segment of our business; the extent and duration for which competitors from each segment of the telecommunications industry are able to offer combined or full service packages prior to our being able to do so; the degree to which we experience material competitive impacts to our traditional service offerings prior to achieving adequate local service entry; and competitor responses to our products and services and overall market acceptance of such products and services;
- The outcome of our negotiations with incumbent local exchange carriers ("ILECs") and state regulatory arbitrations and approvals with respect to interconnection agreements; and our ability to purchase unbundled network elements or wholesale services from ILECs at a price sufficient to permit the profitable offering of local exchange service at competitive rates;
- Success and market acceptance for new initiatives, many of which are untested; the level and timing of the growth and profitability of new initiatives, particularly local access services, Internet (consumer and business) services and wireless services; start-up costs associated with entering new markets, including advertising and promotional efforts; successful deployment of new systems and applications to support new initiatives; and local conditions and obstacles;
- Uncertainties inherent in new business strategies, new product launches and development plans, including local access services, Internet services, wireless services, digital video services, cable modem services, and transmission services;
- Rapid technological changes;
- Development and financing of telecommunication, local access, wireless, Internet and cable networks and services;

- Future financial performance, including the availability, terms and deployment of capital; the impact of regulatory and competitive developments on capital outlays, and the ability to achieve cost savings and realize productivity improvements;
- Availability of qualified personnel;
- Changes in, or failure, or inability, to comply with, government regulations, including, without limitation, regulations of the FCC, the Regulatory Commission of Alaska ("RCA"), and adverse outcomes from regulatory proceedings;
- Uncertainties in federal military spending levels and military base closures in markets in which we operate;
- Other risks detailed from time to time in our periodic reports filed with the Securities and Exchange Commission.

.3

These forward-looking statements (and such risks, uncertainties and other factors) are made only as of the date of this report and we expressly disclaim any obligation or undertaking to disseminate any updates or revisions to any forward-looking statement contained in this document to reflect any change in our expectations with regard to those statements or any other change in events, conditions or circumstances on which any such statement is based. Readers are cautioned not to put undue reliance on such forward-looking statements.

4

<TABLE>

PART I. FINANCIAL INFORMATION

ITEM 1. CONSOLIDATED FINANCIAL STATEMENTS

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS

<CAPTION>

<caption></caption>		(Unaudited)	December 31,
ASSETS		2000	1999
			n thousands)
<\$>	<c></c>		<c></c>
Current assets:			
Cash and cash equivalents	\$	12,609	13,734
Receivables:			
Trade		45,857	48,145
Other		1,302	269
Other			
		47 , 159	48,414
Less allowance for doubtful receivables		3 , 452	2,833
Net receivables		43,707	45,581
Definedable demonstr			0.100
Refundable deposit			9,100
Prepaid and other current assets		1,851 3,033	2,224
Deferred income taxes, net			
Inventories		4,602	3,754
Property held for sale		10,877	
Notes receivable with related parties		478	449
Total current assets		77,157	77,814
Property and equipment in service, net of depreciation		344,789	302 , 762
Construction in progress		4,287	2,898
Net property and equipment		349,076	305,660
Cable franchise agreements, net of amortization of \$18,928,000 and \$16,347,000 at June 30, 2000 and December 31, 1999, respectively Goodwill, net of amortization of \$5,321,000 and \$4,563,000 at		187,564	190,145
June 30, 2000 and December 31, 1999, respectively Other intangible assets, net of amortization of \$490,000 and		40,633	41,391
\$269,000 at June 30, 2000 and December 31, 1999, respectively		4,379	4,402
Property held for sale		1,550	10,877
Deferred loan and senior notes costs, net of amortization		8,861	8,863
Notes receivable with related parties		2,441	2,067
Other assets, at cost, net of amortization		1,431	1,932
Total other assets			259,677
Total assets	\$	673,092	

See accompanying notes to interim condensed consolidated financial statements. </TABLE>

5 (Continued)

<TABLE>

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (Continued)

<CAPTION>

		(Unaudited)	
		June 30,	December 31,
LIABILITIES AND STOCKHOLDERS' EQUITY		2000	1999
		(Amounts in	thousands)
<\$>	<c></c>	(<c></c>
Current liabilities:			
Current maturities of obligations under capital leases	\$	1.714	574 25,321 7,985 8,601
Accounts payable		25,674	25.321
Accrued interest		9.256	7.985
Accrued payroll and payroll related obligations		8 323	8 601
Deferred revenue		8,180	8,173
Accrued liabilities		3 030	2 152
		1 5/7	8,173 3,152 1,314
Subscriber deposits and other current liabilities		1,54/	1,314
Total current liabilities		58,632	55,120
Long-term debt, excluding current maturities		329,400	339,400
Obligations under capital leases, excluding current maturities		47,580	747
Obligations under capital leases due to related party, excluding		•	
current maturities		274	353
Deferred income taxes, net of deferred income tax benefit		25,127	30,861
Other liabilities		4,117	353 30,861 4,210
Total liabilities			430,691
outstanding 20,000 shares at March 31, 2000 and December 31, 1999; and convertible into Class A common stock at \$5.55 per share of Class A common stock, redemption price at March 31, 2000 of \$1,080		21,658	19,912
Stockholders' equity:			
Common stock (no par):			
Class A. Authorized 100,000,000 shares; issued and outstanding			
47,951,024 and 46,869,671 shares at June 30, 2000 and			
December 31, 1999, respectively		180 582	176,740
becomber 31, 1333, respectively		100,302	170,740
Class B. Authorized 10,000,000 shares; issued and outstanding 3,908,148 and 4,048,480 shares at June 30, 2000 and December			
31, 1999, respectively; convertible on a share-per-share basis into Class A common stock		3,303	3,422
THEO CLASS A COMMON SCOCK		3,303	5,422
Less cost of 357,958 and 347,958 Class A common shares held in			
treasury at June 30, 2000 and December 31, 1999, respectively		(1,659)	(1,607)
Paid-in capital		6 , 720	6,343
Notes receivable issued upon stock option exercise		(2,539)	(2,167)
Retained earnings (deficit)		(103)	9,817
Total stockholders' equity		186,304	192,548
Commitments and contingencies			
Total liabilities and stockholders' equity	\$	673,092	643,151
4 4	. ==:		

See accompanying notes to interim condensed consolidated financial statements. </TABLE>

<TABLE>

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS

<CAPTION>

(Unaudited) (Unaudited) Three Months Ended Six Months Ended June 30, June 30, 2000 1999 2000 1999

(Unaudited)

(Amounts in thousands, except per share amounts)

<S>

Revenues	\$	71,426	83,659	139,703	144,997
Cost of sales and services Selling, general and administrative Depreciation and amortization		25,733	34,342 25,236 11,426	59,295 50,387 25,594	48,774
Operating income		3 , 550	12,655	4,427	12,287
Interest expense Interest income	_	9,398 183	8,992 832	19,412 358	952
Interest expense, net	_	9,215	8,160	19,054	15,120
Net income (loss) before income taxes and cumulative effect of a change in accounting principle		(5,665)	4,495	(14,627)	(2,833)
Income tax (expense) benefit		2,139	(2,004)	5,603	803
	-				
Net income (loss) before cumulative effect of a change in accounting principle		(3,526)	2,491	(9,024)	(2,030)
Cumulative effect of a change in accounting principle, net of income tax benefit of \$245					344
Net income (loss)	\$		2,491	(9,024)	(2,374)
Basic and diluted net income (loss) per common share: Income (loss) before cumulative effect of a					
change in accounting principle Cumulative effect of a change in accounting	\$	(.08)	.04	(.19)	(.04)
principle	_				.01
Net income (loss)	\$ =	, ,	.04	(.19)	(.05)

</TABLE>

See accompanying notes to interim condensed consolidated financial statements.

7

<TABLE>

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY SIX MONTHS ENDED JUNE 30, 2000 AND 1999

<CAPTION>

		Class A	Class B	Class A		Notes	
Retained Earnings	(Unaudited)	Common	Common	Shares Held	Paid-in	Receivable	
(Deficit	(Amounts in thousands)	Stock	Stock	in Treasury	Capital	Issued	
	•						
	<pre><s> Balances at December 31, 1998</s></pre>	<c> \$ 172,708</c>	<c> 3,432</c>	<c> (1,607)</c>	<c> 5,609</c>	<c> (637)</c>	<c></c>
20,502	Net loss						
(2,374)	Tax effect of excess stock compensation expense for tax purposes over amounts recognized for financial reporting purposes				93		
	Shares issued and issuable under stock						
	option plan	5			109		
	Shares issued under officer stock option agreements and notes issued upon officer stock option exercise	38				(104)	-

	Shares issued to Employee Stock Purchase Plan	1,153					
	Warrants issued				67		
(289)	Preferred stock dividends						
17,839 ======	Balances at June 30, 1999	·	3,432	(1,607)	5 , 878	(741)	
9 , 817	Balances at December 31, 1999	\$ 176,740	3,422	(1,607)	6,343	(2,167)	
(9,024)	Net loss						
(3,024)	Tax effect of excess stock compensation expense for tax purposes over amounts recognized for financial reporting purposes				192		
	Class B shares converted to Class A	119	(119)				
	Shares issued and issuable under stock option plan	1,406			185		
	Shares issued under officer stock option agreements and notes issued upon officer stock option exercise	450				(372)	_
	Shares issued to Employee Stock Purchase Plan	1,867					
	Purchase of treasury stock			(52)			
(896)	Preferred stock dividends						
(103)	Balances at June 30, 2000	\$ 180,582	3,303	(1,659)	6,720	(2,539)	

</TABLE>
See accompanying notes to interim condensed consolidated financial statements.

<TABLE>

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS

8

<CAPTION>

June 30, 2000

(Unaudited) Six Months Ended

			(Amounts in	
thousand	s)			
<s></s>		<c></c>		<c></c>
	flows from operating activities:			
Net	loss	\$	(9,024)	
(2,374)				
A	djustments to reconcile net loss to net cash provided (used) by operating			
	activities:			
	Depreciation and amortization		25,594	
21,724				
	Amortization charged to selling, general and administrative		495	
894				
	Deferred income tax benefit		(5 , 603)	
(1,048)				
	Deferred compensation and compensatory stock options		376	
417				
	Non-cash cost of sales			
3,703	B 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		61.0	
1 506	Bad debt expense, net of write-offs		619	
1,596				
	Employee Stock Purchase Plan expense funded with General Communication, Inc.		1 201	
1 150	Class A common stock issued and issuable		1,391	
1,153				

	Write-off of capitalized interest	1,955	-
589	Write-off of unamortized start-up costs		
	Write-off of deferred debt issuance costs upon modification of Senior Holdings Loan		
472	Warrants issued		
67	Other noncash income and expense items	(160)	
(33)		4,217	
(4,660)	Change in operating assets and liabilities		
			-
22,500	Net cash provided by operating activities	19,860	
			-
Cash f	lows from investing activities:		
Pur (23,227)	chases of property and equipment, including construction period interest	(18,873)	
Ref	und of deposit	8,806	-
Res (3,978)	tricted cash investment		
	chase of property held for sale	(1,550)	
	chases of other assets	(145)	
	es receivable with related parties issued	(829)	
_	ments received on notes receivable with related parties	582	
149			-
	Net cash used in investing activities	(12,009)	
(27,646)			-
	lows from financing activities: -term borrowings - bank debt		
13,776	yments of long-term borrowings and capital lease obligations	(10,280)	
(20,223)	eeds from preferred stock issuance		
20,000	erred stock offering issuance costs		
(78)	ent of debt issuance costs and loan commitment fees	(128)	
(495)			
(104)	s receivable with related parties issued upon stock option exercise	(372)	
43	eeds from common stock issuance	1,856	
Purc	hase of treasury stock	(52)	-
			-
12,919	Net cash provided (used) by financing activities	(8,976)	
			-
	Net increase (decrease) in cash and cash equivalents	(1,125)	
7,773		\-,,	
12 000	Cash and cash equivalents at beginning of period	13,734	
12,008			
		4	
19,781	Cash and cash equivalents at end of period	\$ 12,609	
		===========	

</TABLE:

See accompanying notes to interim condensed consolidated financial statements.

(1) General

(a) Business

General Communication, Inc. ("GCI"), an Alaska corporation, was incorporated in 1979. We offer the following services:

- Long-distance telephone service between Anchorage,
 Fairbanks, Juneau, and other communities in Alaska and the remaining United States and foreign countries
- Cable television services throughout Alaska
- Facilities-based competitive local access services in Anchorage, Alaska
- Internet access services
- Termination of traffic in Alaska for certain common carriers
- Interstate and intrastate private line services
- Managed services to certain commercial customers
- Sales and service of dedicated communications systems and related equipment
- Private network point-to-point data and voice transmission services between Alaska and the western contiguous United States
- Own and lease capacity on two undersea fiber optic cables used in the transmission of interstate and intrastate private line, switched message long-distance and Internet services between Alaska and the remaining United States and foreign countries

(b) Principles of Consolidation

The consolidated financial statements include the accounts of GCI, GCI's wholly-owned subsidiary GCI, Inc., GCI, Inc.'s wholly-owned subsidiary GCI Holdings, Inc., GCI Holdings, Inc.'s wholly-owned subsidiaries GCI Communication Corp., GCI Cable, Inc. and GCI Transport Co., Inc., GCI Transport Co., Inc.'s wholly-owned subsidiaries GCI Satellite Co., Inc., GCI Fiber Co., Inc. and Fiber Hold Company, Inc. and GCI Fiber Co., Inc.'s and Fiber Hold Company, Inc. and GCI Fiber Co., Inc.'s and Fiber Hold Company, Inc.'s wholly-owned partnership Alaska United Fiber System Partnership ("Alaska United"). GCI Communication Services, Inc. and its wholly owned subsidiary GCI Leasing Co. were merged into GCI Communication Corp. effective January 1, 2000. GCI Cable/Fairbanks, Inc. and GCI Cable/Juneau, Inc. were merged into GCI Cable, Inc. effective January 1, 2000.

All significant intercompany balances and transactions have been eliminated in consolidation.

(c) Net Income (Loss) Per Common Share

<TABLE>

Net loss used to calculate basic and diluted net loss per common share is increased by preferred stock dividends of \$455,000 for the three months ended June 30, 2000 and \$896,000 and \$289,000 for the six months ended June 30, 2000 and 1999, respectively. Net income used to calculate basic and diluted net income per common share is decreased by preferred stock dividends of \$289,000 for the three months ended June 30, 1999. Shares used to calculate net income (loss) per common share consist of the following (amounts in thousands):

<CAPTION>

CAPIION					
		Three Mon		Six Month	
		June	30,	June 30,	
		2000	1999	2000	1999
	-				
	<\$>	<c></c>	<c></c>	<c></c>	<c></c>
	Weighted average common shares outstanding	51,418	49,562	51,341	49,133
	Common equivalent shares outstanding		4,169		
	-				
		51,418	53,731	51,341	49,133
	=				:

</TABLE>

<TABLE>

Common equivalent shares outstanding which are anti-dilutive for purposes of calculating the net loss per common share for the three months ended June 30, 2000 and six months ended June 30,

Notes to Interim Condensed Consolidated Financial Statements (Unaudited)

2000 and 1999, are not included in the diluted net loss per share calculation, and consist of the following (amounts in

	Three Montl June :		Six Months Ended June 30,		
	2000	1999	2000	1999	
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	
Common equivalent shares outstanding	450	n/a 	559 	527	

</TABLE>

<TABLE>

Weighted average shares associated with outstanding stock options for the three and six months ended June 30, 2000 and 1999 which have been excluded from the diluted income (loss) per share calculations because the options' exercise price was greater than the average market price of the common shares consist of the following (amounts in thousands):

<CAPTION>

	Three Months Ended June 30,		Six Months Ended June 30,		
	2000	1999	2000	1999	
<pre><s> Weighted average shares associated with</s></pre>	<c></c>	<c></c>	<c></c>	<c></c>	
outstanding stock options	2 , 579	2,107 ======	2,359 =======	2,213	

</TABLE>

(d) Preferred and Common Stock

<TABLE>

Following is the statement of preferred and common stock at June 30, 2000 and 1999 (shares, in thousands):

<CAPTION>

	Preferred	Common	Stock
	Stock	Class A	Class B
<s></s>	<c></c>	<c></c>	<c></c>
Balances at December 31, 1998		45 , 895	4,061
Class B shares converted to Class A		7	(7)
Shares issued under stock option plan Shares issued under officer stock option		1	
agreements		50	
Shares issued to Employee Stock Purchase Plan Shares issued under Preferred Stock		280	
Agreement	20		
Balances at June 30, 1999	20	46,233 ========	4,054
Balances at December 31, 1999	20	46,870	4,048
Class B shares converted to Class A		140	(140)
Shares issued under stock option plan		181	
Shares issued to Employee Stock Purchase Plan		335	
Warrant exercise		425	
Balances at June 30, 2000	20	47,951	3,908
•	========		========

</TABLE>

(e) Cumulative Effect of a Change in Accounting Principle The American Institute of Certified Public Accountants issued Statement of Position ("SOP") 98-5, "Reporting on the Costs of Start-Up Activities", which provides guidance on the financial reporting of start-up costs and organization costs and requires costs of start-up activities and organization costs to be expensed as incurred. A one-time expense of \$344,000 (net of income tax benefit of \$245,000) associated with the write-off of unamortized start-up costs was recognized in the first quarter of 1999 upon adoption of SOP 98-5.

(Continued)

11 GENERAL COMMUNICATION, INC. Notes to Interim Condensed Consolidated Financial Statements (Unaudited)

Reclassifications have been made to the 1999 financial statements to make them comparable with the 2000 presentation.

(q) Other

The accompanying unaudited interim condensed consolidated financial statements have been prepared in accordance with generally accepted accounting principles for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. The interim condensed consolidated financial statements include the consolidated accounts of GCI and its wholly owned subsidiaries with all significant intercompany transactions eliminated. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. Operating results for the three-month and six-month periods ended June 30, 2000 are not necessarily indicative of the results that may be expected for the year ended December 31, 2000. For further information, refer to the financial statements and footnotes thereto included in our annual report on Form 10-K for the year ended December 31, 1999.

(2) Consolidated Statements of Cash Flows Supplemental Disclosures $\langle {\sf TABLE} \rangle$

Changes in operating assets and liabilities consist of (amounts in thousands):

<CAPTION>

Six-month periods ended June 30,

\(\sigma\)
(Increase) decrease in receivables
Decrease in income tax receivable
(Increase) decrease in prepaid and other current assets
Increase in inventory
increase (decrease) in accounts payable
Increase in accrued liabilities
Increase in accrued payroll and payroll related obligations
ncrease (decrease) in accrued interest
Increase in subscriber deposits and other current liabilities
increase (decrease) in deferred revenues
Increase (decrease) in other long-term liabilities

	2000	1999
<c></c>		<c></c>
\$	1,282	(3,702)
		1,965
	264	(1,235)
	(743)	(504)
	353	(2,028)
	786	1,005
	198	111
	1,271	(105)
	232	411
	7	(97)
	567	(481)
\$	4,217	(4,660)
==:		= ========

</TABLE>

We paid no income taxes during the six-month periods ended June 30, 2000 and 1999. We received income tax refunds of \$0 and \$1,965,000 during the six-month periods ended June 30, 2000 and 1999, respectively.

We paid interest totaling \$16,987,000 and \$16,239,000 during the six-month periods ended June 30, 2000 and 1999, respectively.

We recorded \$192,000 and \$93,000 during the six-month periods ended June 30, 2000 and 1999, respectively, in paid-in capital in recognition of the income tax effect of excess stock compensation expense for tax purposes over amounts recognized for financial reporting purposes.

During the six-month periods ended June 30, 2000 and 1999 we funded the employer matching portion of Employee Stock Purchase Plan contributions by issuing GCI Class A Common Stock valued at \$1,406,000 and \$1,153,000, respectively.

We financed the purchase of satellite transponders pursuant to a long-term capital lease arrangement with a leasing company during the six-month period ended June 30, 2000 at a cost of \$48.2 million.

12 (Continued)

GENERAL COMMUNICATION, INC.

Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

(3) Redeemable Preferred Stock
We issued 20 000 shares of

We issued 20,000 shares of convertible redeemable accreting preferred stock ("Preferred Stock") on April 30, 1999. Proceeds totaling \$20 million (before payment of expenses of \$88,000) were used for general corporate purposes, to repay outstanding indebtedness, and to provide additional liquidity. Our amended Senior Holdings Loan facilities limit use of such proceeds. The Preferred Stock contains a \$1,000 per share liquidation preference, plus accrued but unpaid dividends and

fees. Prior to the four-year anniversary following closing, dividends are payable semi-annually at the rate of 8.5%, plus accrued but unpaid dividends, at our option, in cash or in additional fully-paid shares of Preferred Stock. Dividends earned after the four-year anniversary of closing are payable semi-annually in cash only. Dividends of \$1,746,000 have been accrued at June 30, 2000 and will be paid in additional fully-paid shares of Preferred Stock. Additional dividends totaling \$309,000, or \$15.00 per share, are accrued at June 30, 2000 and the determination of whether they will be paid in cash or additional fully-paid shares of Preferred Stock will be made at the next semi-annual payment date. Mandatory redemption is required 12 years from the date of closing.

(4) Industry Segments Data

Our reportable segments are business units that offer different products. The reportable segments are each managed separately because they manage and offer distinct products with different production and delivery processes.

We have four reportable segments as follows:

Long-distance services. We offer a full range of common-carrier long-distance services to commercial, government, other telecommunications companies and residential customers, through our networks of fiber optic cables, digital microwave, and fixed and transportable satellite earth stations and our SchoolAccess(TM) offering to rural school districts and a similar offering to rural hospitals and health clinics.

Cable services. We provide cable television services to residential, commercial and government users in the State of Alaska. Our cable systems serve 26 communities and areas in Alaska, including the state's three largest urban areas, Anchorage, Fairbanks and Juneau. Cable plant upgrades in 1999 and 1998 enabled us to offer digital cable television services in Anchorage and retail cable modem service (through our Internet services segment) in Anchorage, Fairbanks and Juneau, complementing our existing service offerings. We plan to expand our product offerings as plant upgrades are completed in other communities in Alaska.

Local access services. We offer facilities based competitive local exchange services in Anchorage and plan to provide similar competitive local exchange services in Alaska's other major population centers.

Internet services. We began offering wholesale and retail Internet services in 1998. Deployment of the new undersea fiber optic cable allowed us to offer enhanced services with high-bandwidth requirements.

Included in the "Other" segment in the tables that follow are our managed services, product sales, cellular telephone services, and management services for Kanas Telecom, Inc., a company that owns and operates a fiber optic cable system constructed along the trans-Alaska oil pipeline corridor extending from Prudhoe Bay to Valdez, Alaska. None of these business units have ever met the quantitative thresholds for determining reportable segments. Also included in the Other segment in 1999 is a \$19.5 million sale of undersea fiber optic cable system capacity, and corporate related expenses including marketing, customer service, management information systems, accounting, legal and regulatory, human resources and other general and administrative expenses.

We evaluate performance and allocate resources based on (1) earnings or loss from operations before depreciation, amortization, net interest expense and income taxes, and (2) operating income or loss. The accounting policies of the reportable segments are the same as those described in the summary of

13 (Continued)

GENERAL COMMUNICATION, INC.

Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

significant accounting policies included in our annual report on Form 10-K at December 31, 1999. Intersegment sales are recorded at cost plus an agreed upon intercompany profit.

We earn all revenues through sales of services and products within the United States of America. All of our long-lived assets are located within the United States of America.

Summarized financial information for our reportable segments follows for the six months ended June 30, 2000 and 1999 follows (amounts in thousands):

<caption> Total</caption>		D	Long- istance ervices	Cable Services		Internet Services	Other
	<\$>			<c></c>	<c></c>	<c></c>	<c></c>
C>	2000	\C>		\C >	\C >	\C>	\C>
	Revenues:						
1,908	Intersegment	\$	6,742	739	3,146	1,281	
.39,703	External		88,475	32,590	9,309	3,731	5 , 598
51,611	Total revenues	\$	95,217	33,329	12,455	5,012	5 , 598
 0,133	Earnings (loss) from operations before depreciation, amortization, net interest expense and income taxes				1,320	(4,706)	(17,006
	Operating income (loss)				(1,246)	(5,548)	(18,835
	1999 Revenues:	====		=====			
,705	Intersegment	\$	4,124	1,159	1,422		
44,997	External		79,187	29 , 971	7,478	2,151	26,210
51 , 702	Total revenues		·		8 , 900	2,151	26,210
4,406	Earnings (loss) from operations before depreciation, amortization, net interest expense, income taxes and cumulative effect of a change in accounting principle				(166)	(3,557)	(6 , 599
2,682	Operating income (loss)				(2,763)	(4,088)	(10,825
====== /TABLE> TABLE>	A reconciliation of total segment revention follows:				es		
CAPTION>	Six-months ended June 30,				2000	199	9
	<pre><s> Total segment revenues Less intersegment revenues elimin</s></pre>	ated	in consol	idation	<c></c>	<c> 151,</c>	702
	Consolidated revenues				\$ 139,703		
·/manip.							

14 (Continued)

</TABLE>

A reconciliation of total segment earnings from operations before depreciation, amortization, net interest expense, income taxes and cumulative effect of a change in accounting principle to consolidated net loss before income taxes and cumulative effect of a change in accounting principle follows:

<CAPTION>

Six-months ended June 30,		2000	1999
<pre><s> Total segment earnings from operations before depreciation, amortization, net interest expense, income taxes and cumulative effect of a change in</s></pre>	<c></c>		<c></c>
accounting principle Less intersegment contribution eliminated in consolidation	\$	30,133 (112)	34,406 (395)
Consolidated earnings from operations before depreciation, amortization, net interest expense, income taxes and cumulative effect of a change in accounting principle Depreciation and amortization		30,021 25,594	34,011 21,724
Consolidated operating income Interest expense, net		4,427 19,054	•
Consolidated net loss before income taxes and cumulative effect of a change in accounting principle	\$	(14,627)	(2,833)

</TABLE>

A reconciliation of total segment operating income to consolidated net loss before income taxes and cumulative effect of a change in accounting principle follows:

<CAPTION>

Six-months ended June 30,		2000	1999
<\$>	<c></c>		<c></c>
Total segment operating income	\$	4,539	12,682
Less intersegment contribution eliminated in consolidation			
		(112)	(395)
Consolidated operating income	_	4,427	12,287
Interest expense, net		19,054	15,120
Consolidated net loss before income taxes and cumulative effect of a change in accounting	-		
principle	\$	(14,627)	(2,833)
	=		

</TABLE>

(5)

Commitments and Contingencies

Satellite Transponders Capital Lease

We entered into a purchase and lease-purchase option agreement in August 1995 for the acquisition of satellite transponders to meet our long-term satellite capacity requirements. The satellite was successfully launched in January 2000 and delivered to us on March 5, 2000. In March 2000 we agreed to finance the satellite transponders pursuant to a long-term capital lease arrangement with a leasing company. The base term of the lease is one year from the closing date with the option for eight one-year lease term renewals. The capital lease includes certain covenants requiring maintenance of specific levels of operating cash flow to indebtedness and limitations on additional indebtedness.

We took ownership of the satellite transponders on April 1, 2000. The satellite transponders are recorded at a cost of \$48.2\$ million and will be depreciated over nine years with a remaining residual value of \$14.3\$ million.

15 (Continued)

GENERAL COMMUNICATION, INC.
Notes to Interim Condensed Consolidated Financial Statements

otes to Interim Condensed Consolidated Financial Statements (Unaudited)

Future Fiber Capacity Sale We entered into an agreement effective July 1999 for a second \$19.5 million sale of fiber capacity. The agreement requires that the buyer acquire the capacity during the 18-month period following the effective date of the contract. Costs associated with the capacity to be sold have been classified as Property Held for Sale in the accompanying interim condensed consolidated financial statements at June 30, 2000.

Deferred Compensation Plan

Our non-qualified, unfunded deferred compensation plan provides a means by which certain employees may elect to defer receipt of designated percentages or amounts of their compensation and provides a means for certain other deferrals of compensation. We may, at our discretion, contribute matching deferrals equal to the rate of matching selected by us. Participants immediately vest in all elective deferrals and all income and gain attributable thereto. Matching contributions and all income and gain attributable thereto vest over a six-year period. Participants may elect to be paid in either a single lump sum payment or annual installments over a period not to exceed 10 years. Vested balances are payable upon termination of employment, unforeseen emergencies, death and total disability. Participants are our general creditors with respect to deferred compensation plan benefits. Compensation deferred pursuant to the plan totaled approximately \$30,000 and \$60,000 during the six-month periods ended June 30, 2000 and 1999, respectively.

Self-Insurance

We are self-insured for losses and liabilities related primarily to health and welfare claims up to predetermined amounts above which third party insurance applies. A reserve of \$630,000 and \$600,000 was recorded at June 30, 2000 and December 31, 1999, respectively, to cover estimated reported losses, estimated unreported losses based on past experience modified for current trends, and estimated expenses for investigating and settling claims. Actual losses will vary from the recorded reserve. While management uses what it believes is pertinent information and factors in determining the amount of reserves, future additions to the reserves may be necessary due to changes in the information and factors used.

Litigation and Disputes

We are routinely involved in various lawsuits, billing disputes, legal proceedings and regulatory matters that have arisen in the normal course of business. While the ultimate results of these items cannot be predicted with certainty, management does not expect at this time the resolution of them to have a material adverse effect on our financial position, results of operations or liquidity.

Cable Service Rate Reregulation

Effective March 31, 1999, the rates for cable programming services (service tiers above basic service) are no longer regulated. This regulation ended pursuant to provisions of the Telecommunications Act of 1996 and the regulations adopted pursuant thereto by the FCC. Federal law still permits regulation of basic service rates. However, Alaska law provides that cable television service is exempt from regulation by the RCA unless 25% of a system's subscribers request such regulation by filing a petition with the RCA. At June 30, 2000, only the Juneau system is subject to RCA regulation of its basic service rates. No petition requesting regulation has been filed for any other system. (The Juneau system serves 7.9% of our total basic service subscribers at June 30, 2000.) On July 27, 2000 the RCA approved in full a requested rate increase for the Juneau system, to be effective October 1, 2000.

16

PART I. ITEM 2.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

In the following discussion, General Communication, Inc. and its direct and indirect subsidiaries are referred to as "we," "us" and "our." $\,$

The following discussion and analysis should be read in conjunction with our Interim Condensed Consolidated Financial Statements and the notes thereto. See - Cautionary Statement Regarding Forward-Looking Statements.

OVERVIEW

We have experienced significant growth in recent years through expansion and development of our new and existing businesses and products. We have historically met our cash needs for operations through our cash flows from operating activities. Cash requirements for acquisitions and capital expenditures have been provided largely through our financing activities.

Long-distance services. Our provision of interstate and intrastate long-distance services to residential, commercial and governmental customers and to other common carriers (principally WorldCom, Inc. ("WorldCom") and Sprint Corporation ("Sprint")), and provision of private line and leased dedicated capacity services accounted for 97.1% of our total long-distance services revenues during the second quarter of 2000. Factors that have the greatest impact on

year-to-year changes in long-distance services revenues include the rate per minute charged to customers and usage volumes, usually expressed as minutes of use.

Revenues from private line and other data services sales increased 15.6% to \$6.7 million during the second quarter of 2000 as compared to the second quarter of 1999 due primarily to increased system capacity and increasing demand for data services by Internet service providers ("ISP"), commercial and governmental customers, and others. Demand for data services to and from the lower 48 states previously exceeded the available supply of capacity, however such demand is beginning to be filled with uncompressed fiber optic capacity on our Alaska United fiber optic cable system.

Our long-distance cost of sales and services has consisted principally of direct costs of providing services, including local access charges paid to LECs for originating and terminating long-distance calls in Alaska, and fees paid to other long-distance carriers to carry calls terminating in areas not served by our network. Calls terminating in the lower 49 states are carried over Worldcom's network and calls terminating in international locations are carried principally over Sprint's network. During the second quarter of 2000, local access charges accounted for 66.4% of long-distance cost of sales and services, fees paid to other long-distance carriers represented 28.2%, satellite transponder lease and undersea fiber maintenance costs represented 4.6%, and other costs represented 0.8% of long-distance cost of sales and services.

Our long-distance selling, general, and administrative expenses have consisted of operating and engineering, customer service, sales and communications, management information systems, general and administrative, and legal and regulatory expenses. Most of these expenses consist of salaries, wages and benefits of personnel and certain other indirect costs (such as rent, travel, utilities, insurance and property taxes). A significant portion of long-distance selling, general, and administrative expenses, 37.3% during the second quarter of 2000, represents operating and engineering costs.

Long-distance services face significant competition from AT&T Alascom, Inc., long-distance resellers, and from local telephone companies that have entered the long-distance market. The total number of active long-distance residential, commercial and small business customers increased 5.6% at June 30, 2000 as compared to June 30, 1999, and increased 2.5% as compared to December 31, 1999. We believe our approach to developing, pricing, and providing long-distance services and bundling different business segment services will continue to allow us to be competitive in providing those services.

17 (Continued)

Revenues derived from other common carriers increased 17.3% to \$17.5 million in the second quarter of 2000 as compared to the second quarter of 1999. The increase is due primarily to a 24.7% increase to 170.1 million minutes carried for other common carriers offset by a change in the mix of wholesale minutes carried for such customers, which reduced the average rate charged 6.0%. We secured contract amendments during the second quarter of 1999 with Worldcom and Sprint. The amendments provided, among other things, for a three-year contract term extension for Sprint. The Worldcom contract expires in 2001. Other common carrier traffic routed to us for termination in Alaska is largely dependent on traffic routed to Worldcom and Sprint by their customers. Pricing pressures, new program offerings and market consolidation continue to evolve in the markets served by Worldcom and Sprint. If, as a result, their traffic is reduced, or if their competitors' costs to terminate or originate traffic in Alaska are reduced, our traffic will also likely be reduced, and our pricing may be reduced to respond to competitive pressures. We are unable to predict the effect on us of such changes, however given the materiality of other common carrier revenues to us; a significant reduction in traffic or pricing could have a material adverse effect on our financial position, results of operations and liquidity.

Services included in the Other segment as described in note 4 to the accompanying interim condensed consolidated financial statements are included in the long-distance services segment for purposes of this Management's Discussion and Analysis.

Cable services. During the second quarter of 2000, cable television revenues represented 23.3% of consolidated revenues. The cable systems serve 26 communities and areas in Alaska, including the state's three largest population centers, Anchorage, Fairbanks and Juneau.

We generate cable services revenues from three primary sources: (1) programming services, including monthly basic or premium subscriptions and pay-per-view movies or other one-time events, such as sporting events; (2) equipment rentals or installation; and (3) advertising sales. During the second quarter of 2000 programming services generated 85.4% of total cable services revenues, equipment rental and installation fees accounted for 8.1% of such revenues, advertising sales accounted for 3.6% of such revenues, cable modem services accounted for 2.1% of such revenues and other services accounted for the remaining 0.8% of total cable services revenues. The primary factors that contribute to year-to-year changes in cable services revenues are average monthly subscription

and pay-per-view rates, the mix among basic, premium and pay-per-view services, the average number of subscribers during a given reporting period, and revenues generated from new product offerings.

The cable systems' cost of sales and selling, general and administrative expenses have consisted principally of programming and copyright expenses, labor, maintenance and repairs, marketing and advertising and rental expense. During the second quarter of 2000 programming and copyright expenses represented 45.5% of total cable cost of sales and selling, general and administrative expenses, and general and administrative costs represented 48.9% of such total. Marketing and advertising costs represented approximately 5.7% of such total expenses.

Cable services face competition from alternative methods of receiving and distributing television signals and from other sources of news, information and entertainment. We believe our cable television services will continue to be competitive by providing, at reasonable prices, a greater variety of programming and other communication services than are available off-air or through other alternative delivery sources and upon superior technical performance and customer service.

Local access services. We generate local access services revenues from three primary sources: (1) business and residential basic dial tone services; (2) business private line and special access services; and (3) business and residential features and other charges, including voice mail, caller ID, distinctive ring, inside wiring and subscriber line charges. Effective March 1999 we transitioned to the "bill and keep" cost settlement method for termination of traffic on our facilities and on other's facilities. Local exchange services revenues totaled \$4.8 million representing 6.7% of consolidated revenues in the second quarter of 2000. The primary factors that contribute to year-to-year changes in local access services revenues are the average number of business and residential subscribers to our services during a given reporting period and the average monthly rates charged for non-traffic sensitive services.

18 (Continued)

Operating and engineering expenses represented approximately 3.6% of total local access services cost of sales and selling, general and administrative expenses during the second quarter of 2000. Marketing and advertising costs represented approximately 7.4% of such total expenses, customer service and general and administrative costs represented approximately 42.1% of such total expenses, and local access cost of sales represented approximately 46.9% of such total expenses.

Our local access services segment faces significant competition in Anchorage from Alaska Communications Systems, Inc. ("ACS") and AT&T Alascom, Inc. We believe our approach to developing, pricing, and providing local access services will allow us to be competitive in providing those services.

Internet services. We began offering Internet services in several markets in Alaska during 1998. We generate Internet services revenues from two primary sources: (1) access product services, including commercial DIAS, ISP DIAS, and retail dial-up service revenues, and (2) network management services. Internet services segment revenues totaled \$2.0 million representing 2.8% of total revenues in the second quarter of 2000. The primary factors that contribute to year-to-year changes in Internet services revenues are the average number of subscribers to our services during a given reporting period, the average monthly subscription rates, and the number and type of additional premium features selected

Operating and general and administrative expenses represented approximately 50.2% of total Internet services cost of sales and selling, general and administrative expenses during the second quarter of 2000. Internet cost of sales represented approximately 37.1% of such total expenses and marketing and advertising represented approximately 12.7% of such total expenses.

Marketing campaigns continue to be deployed featuring bundled residential and commercial Internet products. Additional bandwidth was made available to our Internet segment resulting from completion of our Alaska United undersea fiber optic cable project. The new Internet offerings are coupled with our long-distance and local access services offerings and provide free basic Internet services or discounted premium Internet services if certain long-distance or local access services plans are selected. Value-added premium Internet features are available for additional charges.

We compete with a number of Internet service providers in our markets. We believe our approach to developing, pricing, and providing Internet services will allow us to be competitive in providing those services.

Other services, other expenses and net loss. Telecommunications services revenues reported in the Other segment as described in note 4 to the accompanying interim condensed consolidated financial statements include corporate network management contracts, telecommunications equipment sales and service, management services for Kanas Telecom, Inc., a company that owns and

operates a fiber optic cable system constructed along the trans-Alaska oil pipeline corridor extending from Prudhoe Bay to Valdez, Alaska, and other miscellaneous revenues (including revenues from cellular resale services, from prepaid and debit calling cards sales, and installation and leasing of customer's very small aperture terminal ("VSAT") equipment).

Other services segment revenues during the second quarter of 2000 include network solutions and outsourcing revenues totaling \$2.1 million, communications equipment sales totaling \$371,000 and cellular resale and other revenues totaling \$670,000.

During the second quarter of 1999 we completed a \$19.5 million sale of long-haul capacity in our Alaska United undersea fiber optic cable system ("fiber capacity sale") in a cash transaction. The sale includes both capacity within Alaska, and between Alaska and the lower 49 states. We announced in July 1999 that an agreement pertaining to a second \$19.5 million sale of fiber capacity had been executed. The agreement requires that the buyer acquire additional capacity during the 18-month period following the effective date of the contract.

We have invested approximately \$2.2 million in our PCS license at June 30, 2000. During second quarter 2000 we deployed fixed wireless service in the Anchorage area. We have incurred expenditures totaling \$315,000 in the deployment at June 30, 2000 and we expect to incur approximately \$200,000 in additional expenditures during the remainder of 2000.

(Continued)

Depreciation, amortization and net interest expense on a consolidated basis increased \$2.1 million in the second quarter of 2000 as compared to the second quarter of 1999 resulting primarily from additional depreciation on 1999 and 2000 capital expenditures, increased interest rates, and additional average outstanding capital lease obligation balances.

RESULTS OF OPERATIONS

<TABLE>

The following table sets forth selected Statement of Operations data as a percentage of total revenues for the periods indicated and the percentage changes in such data as compared to the corresponding prior year period:

(Underlyi <caption></caption>	ng data rounded to the nearest thousands)					
		Thr	ee Months E June 30,		Si	x Months Ende June 30,	d
Percentag	re			Percentage			
Change (1)				Change (1)			
2000 vs.				2000 vs.			
(Un 1999	audited)	2000	1999	1999	2000	1999	
							-
<s></s>	renues	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
11.7%	Long-distance services	62.8%	48.7%	10.2%	63.3%	54.6%	
8.7%	Cable services	23.3%	17.8%	11.7%	23.3%	20.7%	
24.5%	Local access services	6.7%	4.5%	27.2%	6.7%	5.1%	
73.5%	Internet services	2.8%	1.3%	82.0%	2.7%	1.5%	
(78.6%)	Other services	4.4%	27.7%	(86.6%)	4.0%	18.1%	
(3.7%)	Total revenues	100.0%	100.0%	(14.6%)	100.0%	100.0%	
Cos	t of sales and services	41.5%	41.1%	(13.7%)	42.4%	42.9%	
	ling, general and administrative expenses	36.0%	30.2%	2.0%	36.1%	33.6%	
	reciation and amortization	17.5%					
				454 00)			
(64.0%)	Operating income	5.0%	15.1%	(71.9%)	3.2%	8.5%	
416.3%	Net income (loss) before income taxes and cumulative effect of a change in accounting principle	(7.9%)	5.4%	(226.0%)	(10.5%)	(2.0%)	

	Net income (loss) before cumulative effect of a change in					
	accounting principle	(4.9%)	3.0%	(241.5%)	(6.5%)	(1.4%)
344.6%						
	Net income (loss)	(4.9%)	3.0%	(241.5%)	(6.5%)	(1.6%)
280.1%						
Cabl (18.3%)	r Operating Data (2): e services operating income (3)	22.3%	27.7%	(10.1%)	21.1%	28.1%
	l services operating loss (4)	(3.3%)	(45.7%)	(90.9%)	(13.4%)	(36.9%)
(54.9%)		44.05.00.		4.5 00	(4.40 =0.)	4400 401
Inte 35.7%	rnet services operating loss (5)	(137.3%)	(101.6%)	145.9%	(148.7%)	(190.1%)

- (1) Percentage change in underlying data.
- (2) Includes customer service, marketing and advertising costs.
- (3) Computed as a percentage of total cable services revenues.
- (4) Computed as a percentage of total local services revenues.
- (5) Computed as a percentage of total Internet services revenues.

</FN></TABLE>

20 (Continued)

THREE MONTHS ENDED JUNE 30, 2000 ("2000") COMPARED TO THREE MONTHS ENDED JUNE 30, 1999 ("1999")

Revenues. Total revenues decreased 14.6% from \$83.7 million in 1999 to \$71.4 million in 2000. The decrease is due to the \$19.5 million fiber capacity sale in 1999 as previously described. Excluding the 1999 fiber capacity sale revenue, total revenues grew \$7.2 million, or 11.2%, in 2000 as compared to 1999.

Long-distance revenues from commercial, residential, governmental, and other common carrier customers increased 10.2% to \$44.9 million in 2000. The long-distance revenue increase in 2000 was largely due to the following:

- An increase of 5.7% in the number of active residential, small business and commercial customers billed from 88,100 at June 30, 1999 to 93,100 at June 30, 2000
- An increase of 14.9% in total minutes of use to 257.2 million minutes
- An increase of 15.5% in private line and private network transmission services revenues from \$5.8 million in 1999 to \$6.7 million in 2000 due to an increased number of customers
- An increase of 17.4% in revenues from other common carriers (principally Worldcom and Sprint), from \$14.9 million in 1999 to \$17.5 million in 2000

Long-distance revenue increases were offset by a 11.7% reduction in our average rate per minute on long-distance traffic from \$0.137 per minute in 1999 to \$0.121 per minute in 2000. The decrease in rates resulted primarily from a new category of wholesale minutes carried on our network at a reduced rate per minute. Decreased rates are also attributed to our promotion of and customers' enrollment in calling plans offering discounted rates and length of service rebates, such plans being prompted in part by our primary long-distance competitor, AT&T Alascom, reducing its rates, and the entry of LECs into long-distance markets served by us.

Cable revenues increased 11.7% to \$16.7 million in 2000. Programming services revenues increased 14.2% to \$14.2 million in 2000 resulting from an increase of approximately 4,500 basic subscribers served and increased pay-per-view and premium service revenues. New facility construction efforts during the last half of 1999 and the first half of 2000 resulted in approximately 3,700 additional homes passed at June 30, 2000 which contributed to additional subscribers and revenues in 2000. Revenue per average basic subscriber per month increased \$4.22, or 11.4%, from 1999 to 2000 due to rate increases in certain markets and continued growth of new premium products, such as the 300.6% increase in digital subscribers to 8,900 in 2000. The cable segment's share of cable modem revenue increased \$198,000 to \$341,000 in 2000 after the introduction of such services in the first quarter of 1999.

Local access services revenues increased 27.2% in 2000 to \$4.8 million. At June 30, 2000 approximately 54,600 lines were in service and approximately 1,400 additional lines were awaiting connection as compared to approximately 38,000 lines in service and approximately 1,800 additional lines awaiting connection at June 30, 1999.

Internet services revenues increased from \$1.6 million in 1999 to \$2.0 million in 2000 primarily due to growth in the number of customers served. We have approximately 59,000 active residential, commercial and small business retail dial-up Internet subscribers at June 30, 2000 as compared to approximately 32,200 at June 30, 1999.

Cost of sales and services. Cost of sales and services totaled \$34.3 million in 1999 and \$29.6 million in 2000. As a percentage of total revenues, cost of sales and services increased from 41.1% in 1999 to 41.5% in 2000. The increase in cost

of sales and services as a percentage of revenues is primarily attributed to the impact of the fiber capacity sale, increased cable services cost of sales as a percentage of cable services revenues and changes in our product mix due to the growth of the local access services and Internet services product lines. Off-setting these increases was a decrease in costs resulting from reassigning long-distance and other traffic from leased to owned satellite facilities.

Long-distance cost of sales and services decreased from \$21.1 million in 1999 to \$19.4 million in 2000. Long-distance cost of sales as a percentage of long-distance revenues decreased from 52.6% in 1999 to 43.2% in 2000 primarily due to the effect of reassigning traffic carried by satellite transponders from leased to owned

21 (Continued)

capacity and reductions in access costs due to distribution and termination of our traffic on our own local services network instead of paying other carriers to distribute and terminate our traffic. Offsetting the 2000 decrease as compared to 1999 is a decrease in the average rate per minute billed to customers without a comparable decrease in access charges paid by us. We expect increased cost savings as traffic carried on our own facilities continues to grow. Additional capacity between Alaska and the lower 48 states now available on our Alaska United fiber optic cable system has allowed us to carry significant additional amounts of data services traffic on our own facilities rather than paying other carriers for leased capacity.

Cable cost of sales and services as a percentage of cable revenues, which is less as a percentage of revenues than are long-distance, local access and Internet services cost of sales and services, increased from 25.1% in 1999 to 26.2% in 2000. Cable services rate increases did not keep pace with increases in programming and copyright costs in 2000. Programming costs increased for most of our cable services offerings, and we incurred additional costs on new programming introduced in 1999 and 2000.

Local access services cost of sales and services as a percentage of local access services revenues increased from 46.8% in 1999 to 61.9% in 2000 primarily due to accruals recorded for disputed billings.

Internet services cost of sales and services increased \$124,000 from 1999 to 2000. Internet services costs of sales as a percentage of Internet services revenues totaled 56.6% and 51.2% in 1999 and 2000, respectively. The Internet services costs of sales decrease as a percentage of Internet services revenues is primarily due to a \$830,000 increase in Internet's portion of cable modem revenue. As Internet revenues have increased, economies of scale and more efficient network utilization have resulted in reduced Internet cost of sales and services as a percentage of Internet revenues.

Selling, general and administrative expenses. Selling, general and administrative expenses increased 2.0% to \$25.7 million in 2000. In 2000 we accrued a Company-wide success sharing bonus totaling \$800,000. Success sharing is a bonus paid to all employees when our earnings before interest, depreciation, amortization and taxes reach new highs. No accrual was recorded in the three-month period ended June 30, 1999. The effect of the success sharing accrual was off-set by a \$840,000 decrease in advertising expense. Selling, general and administrative expenses, as a percentage of total revenues, increased from 30.2% in 1999 to 36.0% in 2000 primarily as a result of the fiber capacity sale.

Depreciation and amortization. Depreciation and amortization expense increased 9.5% to \$12.5 million in 2000. The increase is attributable to our \$36.6 million investment in equipment and facilities placed into service during 1999 for which a full year of depreciation will be recorded during the year ended December 31, 2000, the acquisition of a satellite transponder asset (as discussed in note 5 to the interim condensed consolidated financial statements) for which depreciation began in 2000 and the \$18.9 million investment in other equipment and facilities during 2000 for which a partial year of depreciation will be recorded during 2000.

Interest expense, net. Interest expense, net of interest income, increased 12.2% to \$9.2 million in 2000. This increase resulted primarily from increases in our average outstanding indebtedness resulting primarily from the capital lease of satellite transponder capacity, construction of new long-distance and Internet facilities, expansion and upgrades of cable television facilities, investment in local access services equipment and facilities, and higher interest rates on outstanding indebtedness. We charged \$470,000 of deferred financing costs to interest expense in the second quarter of 1999 resulting from the amendment to the Holdings Loan Facilities which reduced borrowing capacity (see Liquidity and Capital Resources).

Income tax (expense) benefit. Income tax (expense) benefit decreased from (\$2.0) million in 1999 to \$2.1 million in 2000 due to an increased net loss before income taxes in 2000 as compared to 1999. Our effective income tax rate decreased from 44.6% in 1999 to 37.8% in 2000 due to the increased net loss and the proportional amount of items that are nondeductible for income tax purposes.

At June 30, 2000, we have (1) tax net operating loss carryforwards of approximately \$105.8 million that will begin expiring in 2008 if not utilized, and (2) alternative minimum tax credit carryforwards of approximately \$2.5 million available to offset regular income taxes payable in future years. Our utilization of remaining net operating loss carryforwards is subject to certain limitations pursuant to Internal Revenue Code section 382.

22 (Continued)

Tax benefits associated with recorded deferred tax assets are considered to be more likely than not realizable through future reversals of existing taxable temporary differences and future taxable income exclusive of reversing temporary differences and carryforwards. The amount of deferred tax asset considered realizable, however, could be reduced in the near term if estimates of future taxable income during the carryforward period are reduced. We estimate that our effective income tax rate for financial statement purposes will be approximately 38% in 2000.

SIX MONTHS ENDED JUNE 30, 2000 ("2000") COMPARED TO SIX MONTHS ENDED JUNE 30, 1999 ("1999")

Revenues. Total revenues decreased 3.7% from \$145.0 million in 1999 to \$139.7 million in 2000. The decrease is due to the \$19.5 million fiber capacity sale in 1999 as previously described. Excluding the 1999 fiber capacity sale revenue, total revenues grew \$14.2 million, or 11.3%, in 2000 as compared to 1999.

Long-distance revenues from commercial, residential, governmental, and other common carrier customers increased 11.7% to \$88.5 million in 2000. The long-distance revenue increase in 2000 was largely due to the following:

- An increase of 5.6% in the number of active residential, small business and commercial customers billed from 88,100 at June 30, 1999 to 93,000 at June 30, 2000
- An increase of 23.4% in total minutes of use to 512.5 million minutes
- An increase of 23.1% in private line and private network transmission services revenues from \$10.4 million in 1999 to \$12.8 million in 2000 due to an increased number of customers
- An increase of 14.4% in revenues from other common carriers (principally Worldcom and Sprint), from \$29.8 million in 1999 to \$34.1 million in 2000

Long-distance revenue increases were offset by a 18.2% reduction in our average rate per minute on long-distance traffic from \$0.148 per minute in 1999 to \$0.121 per minute in 2000. The decrease in rates resulted primarily from a new category of wholesale minutes carried on our network at a reduced rate per minute. Decreased rates are also attributed to our promotion of and customers' enrollment in calling plans offering discounted rates and length of service rebates, such plans being prompted in part by our primary long-distance competitor, AT&T Alascom, reducing its rates, and the entry of LECs into long-distance markets served by us.

Cable revenues increased 8.7% to \$32.6 million in 2000. Programming services revenues increased 9.5% to \$28.0 million in 2000 resulting from an increase of approximately 4,500 basic subscribers served and increased pay-per-view and premium service revenues. New facility construction efforts in the last half of 1999 and first half of 2000 resulted in approximately 3,700 additional homes passed which contributed to additional subscribers and revenues in 2000. Revenue per average basic subscriber per month increased \$4.22, or 11.4%, from 1999 to 2000 due to rate increases in certain markets and continued growth of new premium products, such as the 300.6% increase in digital subscribers to 8,900 in 2000. The cable segment's share of cable modem revenue increased \$562,000 to \$705,000 in 2000 after the introduction of cable modem services in the first quarter of 1999.

Local access services revenues increased 24.5% in 2000 to \$9.3 million. At June 30, 2000 approximately 54,600 lines were in service and approximately 1,400 additional lines were awaiting connection as compared to approximately 38,000 lines in service and approximately 1,800 additional lines awaiting connection at June 30, 1999.

Internet services revenues increased from \$2.2 million in 1999 to \$3.7 million in 2000 primarily due to growth in the number of customers served. We have approximately 59,000 active residential, commercial and small business retail dial-up Internet subscribers at June 30, 2000 as compared to approximately 32,200 at June 30, 1999.

Cost of sales and services. Cost of sales and services totaled \$62.2 million in 1999 and \$59.3 million in 2000. As a percentage of total revenues, cost of sales and services decreased from 42.9% in 1999 to 42.4% in 2000. The decrease in cost of sales and services as a percentage of revenues is primarily due to the effect of reassigning long-distance traffic carried by satellite transponders from leased to owned capacity off-set by the

percentage of cable services revenues and changes in our product mix due to the growth of the local access services and Internet services product lines.

Long-distance cost of sales and services decreased from \$40.7 million in 1999 to \$39.9 million in 2000. Long-distance cost of sales as a percentage of long-distance revenues decreased from 51.4% in 1999 to 45.1% in 2000 primarily due to reassigning traffic carried by satellite transponders from leased to owned capacity and reductions in access costs due to distribution and termination of our traffic on our own local services network instead of paying other carriers to distribute and terminate our traffic. Offsetting the 2000 decrease as compared to 1999 is a decrease in the average rate per minute billed to customers without a comparable decrease in access charges paid by us. We expect increased cost savings as traffic carried on our own facilities continues to grow. Additional capacity between Alaska and the lower 48 states now available on our Alaska United fiber optic cable system has allowed us to carry significant additional amounts of data services traffic on our own facilities rather than paying other carriers for leased capacity.

Cable cost of sales and services as a percentage of cable revenues, which is less as a percentage of revenues than are long-distance, local access and Internet services cost of sales and services, increased from 24.9% in 1999 to 26.6% in 2000. Cable services rate increases did not keep pace with increases in programming and copyright costs in 2000. Programming costs increased for most of our cable services offerings, and we incurred additional costs on new programming introduced in 1999 and 2000.

Local access services cost of sales and services as a percentage of local access services revenues increased from 49.7% in 1999 to 57.8% in 2000 primarily due to accruals recorded for disputed billings.

Internet services cost of sales and services increased \$782,000 from 1999 to 2000. Internet services costs of sales as a percentage of Internet services revenues totaled 61.6% and 56.5% in 1999 and 2000, respectively. The decrease of Internet services costs of sales as a percentage of Internet services revenues is primarily due to a \$1.4 million increase in Internet's portion of cable modem revenue. As Internet revenues have increased, economies of scale and more efficient network utilization have resulted in reduced Internet cost of sales and services as a percentage of Internet revenues.

Selling, general and administrative expenses. Selling, general and administrative expenses increased 3.3% to \$50.4 million in 2000. The 2000 increase resulted from:

- Internet services operating, engineering, sales, customer service and administrative cost increases from \$2.5 million in 1999 to \$3.5 million in 2000. Increased costs were necessary to provide the operations, engineering, customer service and support infrastructure necessary to accommodate expected growth in our Internet services customer base.
- Accrual of a Company-wide success sharing bonus totaling \$800,000 in 2000. Success sharing is a bonus paid to all employees when our earnings before interest, depreciation, amortization and taxes reach new highs. No accrual was recorded in the six-month period ended June 30, 1999.
- Reduced long-distance services capitalized labor due to completion of the fiber optic cable system construction effort.

The increases above are off-set by a \$1.3 million decrease in advertising expense and a \$720,000 decrease in bad debt expense.

Selling, general and administrative expenses, as a percentage of total revenues, increased from 33.6% in 1999 to 36.1% in 2000 primarily as a result of the fiber capacity sale.

Depreciation and amortization. Depreciation and amortization expense increased 17.8% to \$25.6 million in 2000. The increase is attributable to our \$36.6 million investment in equipment and facilities placed into service during 1999 for which a full year of depreciation will be recorded during the year ended December 31, 2000, the acquisition of a satellite transponder asset (as discussed in note 5 to the interim condensed consolidated financial statements) for which depreciation began in the second quarter of 2000, the \$18.9 million investment in other equipment and facilities during 2000 for which a partial year of depreciation will be recorded during 2000, and a charge of \$1.7 million in first quarter resulting from a change in the estimated remaining lives of assets that will be replaced in the future.

24 (Continued)

Interest expense, net. Interest expense, net of interest income, increased 26.5% from \$15.1 million in 1999 to \$19.1 million in 2000. This increase resulted primarily from a charge of \$2.0 million to interest expense in first quarter to write-off previously capitalized interest expense, increases in our average outstanding indebtedness resulting primarily from the capital lease of satellite transponder capacity, construction of new long-distance and Internet facilities, expansion and upgrades of cable television facilities, investment in local access services equipment and facilities, and higher interest rates on outstanding indebtedness. We charged \$470,000 of deferred financing costs to interest expense in the second quarter of 1999 resulting from the amendment to the Holdings Loan Facilities which reduced borrowing capacity (see Liquidity and

Income tax benefit. Income tax benefit increased from \$803,000 in 1999 to \$5.6 million in 2000 due to an increased net loss before income taxes in 2000 as compared to 1999. Our effective income tax rate increased from 28.3% in 1999 to 38.3% in 2000 due to the increased net loss and the proportional amount of items that are nondeductible for income tax purposes.

At June 30, 2000, we have (1) tax net operating loss carryforwards of approximately \$105.8 million that will begin expiring in 2008 if not utilized, and (2) alternative minimum tax credit carryforwards of approximately \$2.5 million available to offset regular income taxes payable in future years. Our utilization of remaining net operating loss carryforwards is subject to certain limitations pursuant to Internal Revenue Code section 382.

Tax benefits associated with recorded deferred tax assets are considered to be more likely than not realizable through future reversals of existing taxable temporary differences and future taxable income exclusive of reversing temporary differences and carryforwards. The amount of deferred tax asset considered realizable, however, could be reduced in the near term if estimates of future taxable income during the carryforward period are reduced. We estimate that our effective income tax rate for financial statement purposes will be approximately 38% in 2000.

25 (Continued)

FLUCTUATIONS IN QUARTERLY RESULTS OF OPERATIONS <TABLE>

The following chart provides selected unaudited statement of operations data from our quarterly results of operations during 2000 and 1999:

	_	(Amount	ts in thousar	nds, except	per share am 	ounts)
		First Quarter	Second Quarter	Third Quarter	Fourth Quarter	Tota Yea
<s></s>	<c></c>		<c></c>	<c></c>	<c></c>	<c></c>
2000						
Revenues:	\$	43,620	44,855			88,4
Long-distance services Cable services	ş \$	15,930	16,660			32,5
Local access services	ş \$		·			
Internet services	ş \$	4,520 1,713	4,789 2,018			9,3 3,7
	\$		•			
Other services	Ş	2,494 	3,104 			5,5
Total revenues	\$	68 , 277	71,426			139,
Operating income	\$	877	3,550			4,
Net loss before income taxes	\$	(8,962)	(5,665)			(14,
Net loss	\$	(5,498)	(3,526)			(9,0
Basic and diluted net loss per	Ψ.	(3) 130)	(3/320)			(5)
common share (1)	\$	(0.12)	(.08)			(0
1999						
Revenues:						
Long-distance services	\$	38,469	40,697	13 276	41,601	164,0
Cable services	\$	15,062	•	15 210	15,957	61,
Local access services	\$	3,714	•	3,845	4,220	15,5
Internet services	\$	1,042	1,109	1,151	1,497	4,
Other services	\$	3,051	23,180	3,850	3,567	33,
	_					
Total revenues	\$ \$	61,338	83,659	67,340	66,842	279,
Operating income (loss) Net income (loss) before income taxes and cumulative effect of a change in	Ş	(368)	12 , 655	1,908	1 , 555	15,7
accounting principle	\$	(7,328)	4,495	(5 , 702)	(6,331)	(14,8
Net income (loss) before cumulative effect		•		. ,	•	
of a change in accounting principle	\$	(4,521)	2,491	(3,537)	(3,616)	(9,1
Net income (loss)	\$	(4,865)	2,491	(3,537)	(3,616)	(9,5
Basic and diluted net income (loss) per common share:						
Net income (loss) before cumulative effect of a change in accounting						
principle (1)	\$	(0.09)	0.04	(0.08)	(0.08)	(0)
Cumulative effect of a change in accounting principle	\$	0.01				0
Net income (loss) (1)	\$	(0.10)	0.04	(0.08)	(0.08)	(0)

1 Due to rounding, the sum of quarterly loss per common share amounts may not agree to year-to-date loss per common share amounts.

</FN>

</TABLE>

Revenues. Total revenues for the quarter ended June 30, 2000 ("second quarter") were \$71.4 million, representing a 4.5% increase from \$68.3 million for the quarter ended March 31, 2000 ("first quarter"). The second quarter increase resulted from:

- A 2.8% increase in long-distance services revenue to \$44.9 million in second quarter primarily due to a 4.9% increase in revenues from other common carriers to \$17.5 million, and a 9.4% increase in private line revenues to \$6.7 million. Long distance minutes increased 1.0% to 257.2 million minutes, due to a

26 (Continued

- 2.2% increase in OCC minutes (principally Worldcom and Sprint) off-set by a 2.0% decrease in non-OCC minutes of traffic carried. The long-distance average rate per minute was \$.121 in the first and second quarters.
- A 4.6% increase in cable services revenue to \$16.7 million in second quarter due to 3.3% increase in programming services revenues to \$14.2 million generated from new product offerings and a 44.9% increase in advertising sales to \$605,000.

Long-distance revenues have historically been highest in the summer months as a result of temporary population increases attributable to tourism and increased seasonal economic activity such as construction, commercial fishing, and oil and gas activities. Cable television revenues, on the other hand, are higher in the winter months because consumers spend more time at home and tend to watch more television during these months. Local service operations are not expected to exhibit significant seasonality. Internet access services are expected to reflect seasonality trends similar to the cable television segment. Our ability to implement construction projects is also hampered during the winter months because of cold temperatures, snow and short daylight hours.

Cost of sales and services. Cost of sales and services decreased from \$29.7 million in the first quarter to \$29.6 million in the second quarter. As a percentage of revenues, second and first quarter cost of sales and services totaled 41.5% and 43.4%, respectively. The decrease in the cost of sales and services as a percentage of revenues is primarily due to the transfer of our traffic from a satellite leased under an operating lease to a satellite owned by us and financed via a capital lease, and avoidance of access charges resulting from distribution and termination of our traffic on our own network instead of paying other carriers to distribute and terminate our traffic.

Selling, general and administrative expenses. Selling, general and administrative expenses increased \$1.1 million in the second quarter as compared to the first quarter. As a percentage of revenues, second quarter selling, general and administrative expenses were 36.0% as compared to 36.1% for the first quarter. The second quarter decrease as a percentage of sales is primarily a result of increased revenues in second quarter without a corresponding proportional increase in support costs off-set by a \$800,000 increase in expenses associated with an accrual for the Company-wide success sharing program in the second quarter. Success sharing is a bonus paid to all employees when our earnings before interest, depreciation, amortization and taxes reach new highs. No such accrual was made in the first quarter.

Net loss. We reported a net loss of \$3.5 million for the second quarter as compared to a net loss of \$5.5 million for the first quarter. The decrease in the net loss is primarily due to a non-recurring charge of \$2.0 million to interest expense in first quarter to write-off previously capitalized interest expense and a non-recurring charge of \$1.7 million in the first quarter resulting from a change in the estimated remaining lives of assets that will be replaced in the future.

LIQUIDITY AND CAPITAL RESOURCES

Cash flows from operating activities totaled \$19.9 million in the six-month period ended June 30, 2000 ("2000") as compared to \$22.5 million in the six-month period ended June 30, 1999 ("1999"), net of changes in the components of working capital. Other sources of cash during 2000 include the refund of a \$8.8 million deposit. Our expenditures for property and equipment, including construction in progress, totaled \$18.9 million and \$23.2 million in 2000 and 1999, respectively. Other uses of cash during 2000 included repayment of \$10.3 million of long-term borrowings and capital lease obligations and purchases of \$1.6 million of property held for sale.

Net receivables decreased \$1.9 million from December 31, 1999 to June 30, 2000 primarily due to decreased OCC trade receivables.

Working capital totaled \$18.5 million at June 30, 2000, a \$4.2 million decrease from working capital of \$22.7 million as of December 31, 1999. The decrease in working capital is primarily attributed to our use of current assets to purchase

long-term capital assets and repay long-term debt.

The Holdings \$200,000,000 (\$150,000,000 as amended) and \$50,000,000 credit facilities mature June 30, 2005. The Holdings Loan facilities were amended in April 1999 (see below) and bear interest, as amended, at

27 (Continued)

either Libor plus 1.00% to 2.50%, depending on the leverage ratio of Holdings and certain of its subsidiaries, or at the greater of the prime rate or the federal funds effective rate (as defined) plus 0.05%, in each case plus an additional 0.00% to 1.375%, depending on the leverage ratio of Holdings and certain of its subsidiaries. \$77.7 million and \$87.7 million were drawn on the credit facilities as of June 30, 2000 and December 31, 1999, respectively.

On April 13, 1999, we amended the Holdings credit facilities. These amendments contained, among other things, provisions for payment of a one-time amendment fee of 0.25% of the aggregate commitment, an increase in the commitment fee by 0.125% per annum on the unused portion of the commitment, and an increase in the interest rate of 0.25%. The amended facilities reduce the aggregate commitment by \$50 million to \$200 million, and limit capital expenditures to \$35 million in 1999 and \$35 million in 2000 with no limits thereafter (excluding amounts paid for the Alaska United fiber optic cable system and the capital lease of the satellite transponder asset). Pursuant to the Financial Accounting Standards Board Emerging Issues Task Force Issue 98-14, "Debtor's Accounting for Changes in Line-of-Credit or Revolving Debt Arrangements," we recorded as additional interest expense \$470,000 of deferred financing costs in the second quarter of 1999 resulting from the reduced borrowing capacity. In connection with the April 1999 amendment, we agreed to pay all fees and expenses of our lenders, including an amendment fee of 0.25% of the aggregate commitment, totaling \$530,000.

Holding's credit facilities and GCI, Inc.'s senior notes contain restrictions on our operations and activities, including requirements that we comply with certain financial covenants and financial ratios. Under the amended Holding's credit facility, Holdings may not permit the ratio of senior debt to annualized operating cash flow (as defined) of Holdings and certain of its subsidiaries to exceed 2.75 to 1.0 through September 30, 2000 and 2.50 to 1.0 from October 1, 2000 to December 31, 2000, total debt to annualized cash flow to exceed 5.50 times, and annualized operating cash flow to interest expense to be less than 2.0 to 1.0 from April 1, 2000 and thereafter. Certain of the foregoing ratios decrease in specified increments during the life of the credit facility. The credit facility requires Holdings to maintain a ratio of annualized operating cash flow to debt service of Holdings and certain of its subsidiaries of at least 1.25 to 1.0, and annualized operating cash flow to fixed charges of at least 1.0 to 1.0 effective January 1, 2001 (which adjusts to 1.05 to 1.0 in April, 2003 and thereafter). The senior notes impose a requirement that the leverage ratio of GCI, Inc. and certain of its subsidiaries not exceed 6.0 to 1.0 on an incurrence basis, subject to the ability of GCI, Inc. and certain of its subsidiaries to incur specified permitted indebtedness without regard to such ratios.

On January 27, 1998 Alaska United closed a \$75 million project finance facility ("Fiber Facility") to construct a fiber optic cable system connecting Anchorage, Fairbanks, Valdez, Whittier, Juneau and Seattle. At June 30, 2000 and December 31, 1999 \$71.7 million was borrowed under the facility. The Fiber Facility is a 10-year term loan that is interest only for the first 5 years. The facility can be extended an additional two years at any time between the second and fifth anniversary of closing the facility if we can demonstrate projected revenues from certain capacity commitments will be sufficient to pay all operating costs, interest, and principal installments based on the extended maturity. The Fiber Facility bears interest at either Libor plus 3.0%, or at the lender's prime rate plus 1.75%. The interest rate will decline to Libor plus 2.5%-2.75%, or, at our option, the lender's prime rate plus 1.25%-1.5% after the project completion date and when the loan balance is \$60 million or less.

The Fiber Facility contains, among others, covenants requiring certain intercompany loans and advances in order to maintain specific levels of cash flow necessary to pay operating costs, interest and principal installments. All of Alaska United's assets, as well as a pledge of the partnership interests' owning Alaska United, collateralize the Fiber Facility. Construction of the fiber facility was completed and the facility was placed into service on February 4, 1999. The project was completed on budget.

We expect to use approximately one-half of the Alaska United system capacity in addition to our existing owned and leased facilities to carry our own traffic. One of our large commercial customers signed agreements in the first quarter of 1999 for the immediate lease of three DS3 circuits on Alaska United facilities within Alaska, and between Alaska and the lower 48 states. The lease agreements provide for three-year terms, with renewal options for additional terms. In the second quarter of 1999 we completed a sale of capacity in our Alaska United system in a \$19.5 million cash transaction. The sale includes both capacity within Alaska, and between Alaska and the lower 48 states. An agreement was executed in July 1999 for a second \$19.5 million sale of fiber capacity. The agreement requires that the buyer acquire additional capacity

28 (Continued)

during the 18-month period following the effective date of the contract. We continue to pursue opportunities for sale or lease of additional capacity on our system.

We entered into a purchase and lease-purchase option agreement in August 1995 for the acquisition of satellite transponders to meet our long-term satellite capacity requirements. The satellite was successfully launched in January 2000 and delivered to us on March 5, 2000. In March 2000 we agreed to finance the satellite transponders pursuant to a long-term capital lease arrangement with a leasing company. At June 30, 2000 \$48.2 million was financed under this capital lease. The base term of the lease is one year from the closing date with the option for eight one-year lease term renewals. The capital lease includes certain covenants requiring maintenance of specific levels of operating cash flow to indebtedness and limitations on additional indebtedness.

Our expenditures for property and equipment, including construction in progress, totaled \$9.8 million and \$23.2 million during 2000 and 1999, respectively. Planned capital expenditures over the next five years include those necessary for continued expansion of our long-distance, local exchange and Internet facilities, the development and construction of a PCS network and continued upgrades to our cable television plant. Sources of funds for these planned capital expenditures are expected to include internally generated cash flows and borrowings under our credit facilities.

Our ability to invest in discretionary capital and other projects will depend upon our future cash flows and access to borrowings under our credit facilities. Management anticipates that cash flow generated by us and our borrowings under our credit facilities will be sufficient to fund capital expenditures and our working capital requirements. Should cash flows be insufficient to support additional borrowings, such investment in capital expenditures will likely be reduced.

We issued 20,000 shares of convertible redeemable accreting preferred stock ("Preferred Stock") on April 30, 1999. Proceeds totaling \$20 million (before payment of expenses) were used for general corporate purposes, to repay outstanding indebtedness, and to provide additional liquidity. Prior to the four-year anniversary following closing, dividends are payable semi-annually at the rate of 8.5%, plus accrued but unpaid dividends, at our option, in cash or in additional fully-paid shares of Preferred Stock. Dividends earned after the four-year anniversary of closing are payable semi-annually in cash only. Dividends of \$1,746,000 have been accrued at June 30, 2000 and will be paid in additional fully-paid shares of Preferred Stock. Additional dividends totaling \$309,000, or \$15.00 per share, are accrued at June 30, 2000 and the determination of whether they will be paid in cash or additional fully-paid shares of Preferred Stock will be made at the next semi-annual payment date. Mandatory redemption is required 12 years from the date of closing.

The long-distance, local access, cable, Internet and wireless services industries are experiencing increasing competition and rapid technological changes. Our future results of operations will be affected by our ability to react to changes in the competitive environment and by our ability to fund and implement new technologies. We are unable to determine how competition, technological changes and our net operating losses will affect our ability to obtain financing.

We believe that we will be able to meet our current and long-term liquidity and capital requirements, including fixed charges and Preferred Stock dividends, through our cash flows from operating activities, existing cash, cash equivalents, short-term investments, credit facilities, and other external financing and equity sources.

NEW ACCOUNTING PRONOUNCEMENTS

SFAS No. 133. In June 1998, the Financial Accounting Standards Board issued SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities". Among other provisions, SFAS No. 133, as amended by SFAS No. 138, "Accounting for Certain Derivative Instruments and Certain Hedging Activities and Amendment of SFAS No. 133", requires that entities recognize all derivatives as either assets or liabilities in the statement of financial position and measure those instruments at fair value. Gains and losses resulting from changes in the fair values of those derivatives would be accounted for depending on the use of the derivative and whether it qualifies for hedge accounting. The effective date of this standard was delayed via the issuance of SFAS No. 137. The effective date for SFAS No. 133 is now for fiscal years beginning after June 15, 2000, though earlier adoption is encouraged and retroactive application is prohibited. This means

9 (Continued)

FASB Interpretation No. 44. In March 2000, the Financial Accounting Standards Board issued FASB Interpretation No. 44, "Accounting for Certain Transactions Involving Stock Compensation". This interpretation clarifies the application of APB Opinion No. 25, "Accounting for Stock Issued to Employees", for certain issues including the definition of employee for purposes of applying APB Opinion No. 25, the criteria for determining whether a plan qualifies as a noncompensatory plan, the accounting consequence of various modifications to the terms of a previously fixed stock option or award and the accounting for an exchange of stock compensation awards in a business combination. This interpretation is effective July 1, 2000, but certain conclusions in the interpretation cover specific events that occur after either December 15, 1998, or January 12, 2000. To the extent that this interpretation covers events occurring during the period after December 15, 1998, or January 12, 2000, but before the effective date of July 1, 2000, the effects of applying it are recognized on a prospective basis from July 1, 2000. We do not expect the adoption of this standard to have a material impact on our results of operations, financial position or cash flows.

SEC Staff Accounting Bulletin No. 101. SEC Staff Accounting Bulletin No. 101, "Revenue Recognition in Financial Statements", summarizes certain of the SEC staff's views in applying generally accepted accounting principles to revenue recognition in financial statements. This bulletin is effective October 1, 2000, we believe the adoption will not have a material impact on our results of operations, financial position or cash flows.

Year 2000 Costs

We did not defer any critical information technology projects because of our Year 2000 program efforts. At June 30, 2000 we have an estimated \$200,000 in remaining incremental remediation costs.

ALASKA ECONOMY

We offer voice and data telecommunication and video services to customers primarily throughout Alaska. As a result of this geographic concentration, growth of our business and of our operations depend upon economic conditions in Alaska. The economy of Alaska is dependent upon the natural resource industries, and in particular oil production, as well as investment earnings, tourism, government, and United States military spending. Any deterioration in these markets could have an adverse impact on us. Oil revenues are now the third largest source of state revenues, following investment income and federal funds. Alaska's investment earnings will supply 34% of the state's projected revenues in fiscal 2001, with federal funding comprising 25% and oil revenues 27% of the total. Much of the investment income and all of the federal funding is restricted or dedicated for specific purposes, however, leaving oil revenues as the primary funding source (75%) of general operating expenditures.

The volume of oil transported by the TransAlaska Oil Pipeline System ("TAPS") over the past 20 years has been as high as 2.0 million barrels per day in fiscal 1988. Production has begun to decline in recent years and is presently down 40% from the fiscal 1988 level, and down 25% from the fiscal 1997 level. The two largest producers of oil in Alaska (the primary users of the TAPS) continue to explore, develop and produce new oil fields and to enhance recovery from existing fields to offset the decline in production from the Prudhoe Bay field. Both companies have invested large sums of money in developing and implementing oil recovery techniques at the Prudhoe Bay field and other nearby fields. The state now forecasts a temporary reversal of the production rate decline and a slight increase in the production rate in 2005. This forecasted increase is attributed to new developments at the Alpine, Liberty and Northstar fields, as well as new production from Prudhoe Bay and other fields.

Market prices for North Slope oil declined to below \$10 per barrel in 1998, and averaged \$12.70 in fiscal 1999, well below the average price used by the state to budget its oil related revenues. The prices have since increased to a 10-year high of \$32.30 on March 7, 2000, with a fiscal 2000 average price per barrel of \$23.27.

The July 2000 update to the state's spring 2000 forecast for fiscal 2001 forecasts the price for North Slope crude averaging \$26.77 and then declining to \$22.32 in fiscal 2002 and \$18.53 over the following few years. Recent higher prices are largely due to the March 1999 OPEC agreement to cut production to force prices

30 (Continued)

higher. The OPEC agreement called for production cuts from January 1999 levels of a little more than 2 million barrels per day. At its March 27, 2000 meeting, nine of the eleven OPEC members agreed to increase production quotas by a total of 1.452 million barrels per day. Iran did not agree to an official quota but has been quoted as saying it would increase production sufficient to maintain its market share. Iraq is not subject to an OPEC quota. Based on estimates of current production, the new production quotas for the nine members would represent about a 450,000 barrels per day increase. At its June 21, 2000 meeting, OPEC members agreed to increase production quotas by an additional 500,000 barrels per day. History suggests that market forces lead to lower

prices when oil sells for more than \$20 per barrel. What is uncertain is when and how fast the correction will occur. The response of non-OPEC production to higher prices is uncertain. The production policy of OPEC and its ability to continue to act in concert represents a key uncertainty in the state's revenue forecast.

The state of Alaska maintains the Constitutional Budget Reserve Fund that is intended to fund budgetary shortfalls. The state withdrew \$256 million from the Constitutional Budget Reserve Fund in fiscal 2000 and, based on the state's oil price and production forecasts, and considering the state's other revenues, the Alaska Department of Revenue expects to draw about \$122 million in fiscal 2001 to balance the state's budget, down substantially from the \$413 million fiscal 2001 draw expected in their spring 2000 forecast. If the state's current projections are realized, the Constitutional Budget Reserve Fund will be depleted in 2004. If the fund is depleted, aggressive state action will be necessary to increase revenues and reduce spending in order to balance its budget. The Governor of the state of Alaska and the Alaska Legislature are pursuing cost cutting and revenue enhancing measures.

Oil companies and service providers announced cost cutting measures to offset a portion of the declining oil revenues in 1999, resulting in a reduction of oil industry jobs of over 1,400. Projects that are underway are reportedly not affected by the cutbacks, however BP (previously BP Amoco) did notify state officials that it would delay its exploration of the Genesee test site east of Prudhoe.

Although oil prices have a substantial effect on Alaska's economy, analysts believe that tourism, air cargo, and service sectors are strong enough to offset a portion of the expected downturn. These industries have helped offset the prevailing pattern of oil industry downsizing that has occurred during much of the last several years. Two other factors that support Alaska's economy are the healthy national economy and low inflation. Economists expect construction to remain strong over the next few years. \$1.69 billion of federal money is expected to be distributed to the State of Alaska for highways and other federally supported projects in fiscal 2001.

Effective March 1997, the State of Alaska passed new legislation relaxing state oil royalties with respect to marginal oil fields that the oil companies claim would not be economic to develop otherwise. No assurance can be given that oil companies doing business in Alaska will be successful in discovering new fields or further developing existing fields which are economic to develop and produce oil with access to the pipeline or other means of transport to market, even with the reduced level of royalties.

Should new discoveries or developments not materialize or the price of oil return to its prior depressed levels, the long term trend of continued decline in oil production from the Prudhoe Bay field area is inevitable with a corresponding adverse impact on the economy of the state, in general, and on demand for telecommunications and cable television services, and, therefore, on us, in particular.

BP and Atlantic Richfield Company ("ARCO") announced April 13, 2000 that they received clearance from the Federal Trade Commission for the combination of their companies, which was completed April 18, 2000.

BP, Exxon Mobil Corporation ("ExxonMobil"), ARCO and Phillips Petroleum ("Phillips") announced April 13, 2000 that they reached an agreement to resolve outstanding issues relating to the ownership and operation of the Prudhoe Bay and Point Thomson Units in Alaska. The agreement reportedly is intended to optimize operations, reduce costs and facilitate new oil and gas development in the state of Alaska. The agreement aligns the respective equity interests of BP Exploration (Alaska), ExxonMobil and Phillips in the Prudhoe Bay Unit, and provides for a single operator at that unit. In addition, the agreement resolves issues relating to North Slope preferential rights and field operatorship. The companies stated that the agreement will not only help ensure the efficient and long-term production of the fields, but will also facilitate future Alaska development, including gas commercialization.

31 (Continued)

The aligned oil and gas interests among the major owners will be 26.7% for BP Exploration (Alaska), 36.8% for ExxonMobil and 36.5% for Phillips. BP Exploration (Alaska), current operator of the Western Operating Area in the Prudhoe Bay Unit, will become the single operator. ExxonMobil and BP Exploration (Alaska) Inc. have also agreed to work towards alignment in the Point Thomson field area with respective interests of 45% for BP Exploration and 55% for ExxonMobil

Phillips became a major new operator of the North Slope Kuparuk and Alpine fields, following Federal Trade Commission approval and final closing of the ARCO Alaska acquisition August 1, 2000.

We have, since our entry into the telecommunication marketplace, aggressively marketed our services to seek a larger share of the available market. The customer base in Alaska is limited, however, with a population of approximately

620,000 people. 42% of the State's population are located in the Anchorage area, 14% are located in the Fairbanks area, 5% are located in the Juneau area, and the rest are spread out over the vast reaches of Alaska. No assurance can be given that the driving forces in the Alaska economy, and in particular, oil production, will continue at levels to provide an environment for expanded economic activity.

No assurance can be given that oil companies doing business in Alaska will be successful in discovering new fields or further developing existing fields which are economic to develop and produce oil with access to the pipeline or other means of transport to market, even with the reduced level of royalties. We are not able to predict the effect of changes in the price and production volumes of North Slope oil or the acquisition of ARCO by BP and Phillips on Alaska's economy or on us.

SEASONALITY

Long-distance revenues have historically been highest in the summer months as a result of temporary population increases attributable to tourism and increased seasonal economic activity such as construction, commercial fishing, and oil and gas activities. Cable television revenues, on the other hand, are higher in the winter months because consumers tend to watch more television, and spend more time at home, during these months. Our local access services revenues are not expected to exhibit significant seasonality. Our Internet access services are expected to reflect seasonality trends similar to the cable television segment. Our ability to implement construction projects is reduced during the winter months because of cold temperatures, snow and short daylight hours.

INFLATION

We do not believe that inflation has a significant effect on our operations.

32

PART I.

ITEM 3. QUANTITATIVE AND QUALITIATIVE DISCLOSURES ABOUT MARKET RISK

We are exposed to various types of market risk in the normal course of business, including the impact of interest rate changes. We do not hold derivatives for trading purposes.

Our Senior Holdings Loan carries interest rate risk. Amounts borrowed under this Agreement bear interest at either Libor plus 1.0% to 2.5%, depending on the leverage ratio of Holdings and certain of its subsidiaries, or at the greater of the prime rate or the federal funds effective rate (as defined) plus 0.05%, in each case plus an additional 0.0% to 1.375%, depending on the leverage ratio of Holdings and certain of its subsidiaries. Should the Libor rate, the lenders' base rate or the leverage ratios change, our interest expense will increase or decrease accordingly. As of June 30, 2000, we have borrowed \$77.7 million subject to interest rate risk. On this amount, a 1% increase in the interest rate would cost us \$777,000 in additional gross interest cost on an annualized basis.

Our Fiber Facility carries interest rate risk. Amounts borrowed under this Agreement bear interest at either Libor plus 3.0%, or at our choice, the lender's prime rate plus 1.75%. The interest rate will decline to Libor plus 2.5%-2.75%, or at our choice, the lender's prime rate plus 1.25%-1.5% after the project completion date and when the loan balance is \$60,000,000 or less. Should the Libor rate, the lenders' base rate or the leverage ratios change, our interest expense will increase or decrease accordingly. As of June 30, 2000, we have borrowed \$71.7 million subject to interest rate risk. On this amount, a 1% increase in the interest rate would cost us \$717,000 in additional gross interest cost on an annualized basis.

Our Satellite Transponder Capital Lease carries interest rate risk. Amounts borrowed under this Agreement bear interest at Libor plus 3.25%. Should the Libor rate change, our interest expense will increase or decrease accordingly. As of June 30, 2000, we have borrowed \$48.2 million subject to interest rate risk. On this amount, a 1% increase in the interest rate would cost us \$482,000 in additional gross interest cost on an annualized basis.

PART II. OTHER INFORMATION ITEM 1. LEGAL PROCEEDINGS

Information regarding pending legal proceedings to which we are a party is included in Note 5 of Notes to Interim Condensed Consolidated Financial Statements and is incorporated herein by reference.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

- (a) Date of the meeting: June 8, 2000
 Purpose of meeting: Annual shareholders meeting
- (b) Name of each director elected at the meeting and the name of each other director whose term of office as a director

continued after the meeting:

Name	Votes for	Votes withheld
Ronald A. Duncan	71,459,435	6,517,100
Paul S. Lattanzio	73,440,440	4,536,095
Stephen R. Mooney	73,440,689	4,535,846
Larry E. Romrell	71,198,901	6,777,634

Directors, in addition to those listed above, whose term of office as director continued after the meeting:

Ronald R. Beaumont Donne F. Fisher William P. Glasgow Carter F. Page James M. Schneider Robert M. Walp

(c) Other matters voted upon:

Amendment to the Company's Revised 1986 Stock Option Plan to increase the number of shares of the Company's common stock allocated to the plan by 1.5 million shares of Class A common stock, ratifying an amendment to the plan deleting the exceptions to Board of Director authority to amend the Stock Option Plan without shareholder approval and adding a new exception to that authority, and approval of several administrative amendments to the plan approved by the Board of Directors.

Votes

For: 77,991,849 Against: 6,232,045 Abstain: 75,389 Result: Passed

Amendments to the Company's Restated Articles of Incorporation generally relating to the terms under which the Company's board of directors may approve issuance of Company preferred stock, and approving action by the board canceling and otherwise deleting a statement of stock designation as issued and filed with the State of Alaska and relating to a 1991 offer of preferred stock which is no longer outstanding.

Votes

For: 62,177,486 Against: 10,859,570 Abstain: 119,970 Result: Passed

(d) Not applicable

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) Exhibits:

Exhibit 10.82 - Lease Intended for Security between GCI Satellite Co., Inc. and General Electric Capital Corporation *

Exhibit 27 - Financial Data Schedule *

(b) Reports on Form 8-K filed during the quarter ended June 30, 2000 - None

* Filed herewith.

34

<TABLE>

GENERAL COMMUNICATION, INC.

<caption> Signature</caption>	Title	Date
<s></s>	<c></c>	<c></c>
/s/ Ronald A. Duncan	President and Director (Principal Executive Officer)	August 10, 2000
/s/ John M. Lowber	Senior Vice President, Chief Financial Officer, Secretary and Treasurer (Principal Financial Officer)	August 10, 2000
/s/ 	Vice President, Chief Accounting Officer (Principal Accounting Officer)	August 10, 2000

</TABLE>

LEASE INTENDED FOR SECURITY

Dated as of March 31, 2000

among

GCI SATELLITE CO., INC.,

as Lessee,

STATE STREET BANK AND TRUST COMPANY OF CONNECTICUT, NATIONAL ASSOCIATION,

not in its individual capacity, except as expressly set forth herein, but solely as Owner Trustee,

and

GENERAL ELECTRIC CAPITAL CORPORATION,

as Owner Participant

<table></table>	
(1110000)	TABLE OF CONTENTS
<caption></caption>	Page
<s></s>	- age
<c></c>	
	1
Introduction	1
Section 1.	Procurement, Delivery and Acceptance2
Section 2.	Term, Rent and Payment9
Section 3.	Warranties
Section 4.	Possession, Assignment, Sublease, Use and Maintenance
Section 5.	General Tax Indemnity
Section 6.	Increased Costs, Etc
Section 7.	Risk of Loss; Replacement; Waiver and Indemnity17
Section 8.	[Reserved.]
Section 9.	Default
Section 10.	Return of Units24
Section 11.	Distributions to Owner Participants24
Section 12.	Limitation of Liability; Compensation26
Section 13.	Assignments
Section 14.	Ownership, Grant of Security Interest and Further Assurances27
Section 15.	Late Payments, Etc
Section 16.	Effect of Waiver
Section 17.	Survival of Covenants
Section 18.	APPLICABLE LAW
Section 19.	Effect and Modification of Lease29
Section 20.	Reports; Confidentiality
Section 21.	Notices
Section 22.	Counterparts32
Section 23. Section 24.	Severability
Section 24.	Successors and Assigns; Merger
Section 26.	Jury Trial
Section 27.	Captions; Table of Contents
Section 28.	Schedules and Exhibits
Section 29.	Certain Definitions

	Schedule I	List of Owner Participants
Schedule II	A. Description of Units	
	B. Aggregate Purchase Price	
	C. Rental	
	D. Interest on Late Payments	
	E. Purchase/Remarketing Provision	
	Annex I Description of Units	
	Annex II Rental Payments	
	Annex III Return Requirements	
Exhibit A	Form of Bill of Sale (Section 1.2(a))	
Exhibit B	Form of Acceptance Certificate (Section 1.3(o))	
Exhibit C-1	Form of Opinion of In-House Counsel to Lessee	
Exhibit C-2	Form of Opinion of FCC Counsel	
Exhibit C-3	Form of Opinion of Special Counsel	
Exhibit D-1	Form of Assignment	
Exhibit D-2	Form of PanAmSat Consent and Agreement	
Exhibit E	Form of Closing Notice	

LEASE INTENDED FOR SECURITY

This LEASE INTENDED FOR SECURITY (as amended, modified, restated or supplemented from time to time, this "Lease") dated as of March 31, 2000 (the "Execution Date"), is among GCI SATELLITE CO., INC., an Alaska corporation ("Lessee"), STATE STREET BANK AND TRUST COMPANY OF CONNECTICUT, NATIONAL ASSOCIATION, a national banking association ("State Street"), not in its individual capacity, except as otherwise set forth herein, but solely in its capacity as owner trustee pursuant to the Trust Agreement hereinafter referred to (in such capacity, the "Owner Trustee"); and GENERAL ELECTRIC CAPITAL CORPORATION, a New York corporation ("GECC"), as the sole initial Owner Participant.

RECITALS

- A. GCI Communication Corp., an Alaska corporation and an Affiliate of Lessee ("GCI Communication"), entered into the Galaxy X Transponder Purchase Agreement dated as of August 24, 1995 (as heretofore or hereafter amended or supplemented, the "Purchase Agreement"), with Hughes Communications Galaxy, Inc., who subsequently assigned its rights and obligations thereunder to PanAmSat Corporation, a Delaware corporation ("PanAmSat"), pursuant to which GCI Communication agreed to buy six Primary C-Band Transponders and one Ku-Band Transponder (as more fully described on Annex I to Schedule II, individually a "Unit" and collectively the "Units"). GCI Communication assigned its rights and obligations under the Purchase Agreement to Lessee.
- B. Owner Trustee agrees, for the account of Owner Participants, to acquire from PanAmSat and lease to Lessee and Lessee agrees to lease from Owner Trustee the Units, upon the terms and conditions hereinafter set forth. In connection with such acquisition, Lessee has assigned to Owner Trustee all of its rights and claims under the Purchase Agreement and the Services Agreement.
- C. Lessee intends to sublease the Units to GCI Holdings, Inc., an Alaska corporation ("Sublessee"), on the Delivery Date pursuant to a Sublease Agreement, dated as of the date hereof, in substantially the form attached hereto as Exhibit F (as amended, supplemented or otherwise modified from time to time, the "Sublease").
- D. To secure Lessee's obligations under this Lease, Lessee will grant to Owner Trustee a security interest in the Units, the Sublease and the other Collateral.

 Section 1. Procurement, Delivery and Acceptance.
- 1.1. Subject to the terms and conditions of this Lease, including the satisfaction or waiver of the conditions set forth in Section 1.2, below, (i) Owner Trustee agrees that, on a Business Day specified by the Lessee in the Closing Notice, which date shall be on or prior to March 31, 2000 (the "Delivery Date"), it shall (a) purchase the Units from PanAmSat and pay certain transaction costs and warranty costs related to the Units and the transactions contemplated hereby, with funds provided by the Owner Participants, for the Aggregate Purchase Price, as provided herein, and (ii) each Owner Participant, severally and for itself only, hereby agrees to transfer on the Delivery Date such Owner Participant's ratable share, in accordance with the percentage set forth opposite its name in Schedule I hereto, of the Aggregate Purchase Price for the Units; provided that the Aggregate Purchase Price shall not exceed Fifty-Three Million Six Hundred Thirty One Thousand Dollars (\$53,631,000). Lessee acknowledges and agrees that its acceptance of the Units on the Delivery Date shall be conclusive proof that, as of the Delivery Date, each such Unit is in compliance with all requirements of this Lease. On the Delivery Date, Owner Trustee shall lease and Lessee shall accept each Unit and each component part thereof "AS IS" at that time.
- 1.2. The obligation of Owner Trustee and each Owner Participant to purchase from, and lease to, Lessee each Unit is subject to the following conditions, each of which conditions (i) unless waived as provided below, must be met on or prior to the Delivery Date (unless Owner Participants, Lessee and Owner Trustee agree to another date with respect to any given condition) in form and substance satisfactory to Owner Participants and Owner Trustee, and (ii) is for the sole benefit of the Owner Participants and may be waived (but only by all Owner Participants) in their sole discretion:
 - (a) PanAmSat shall have executed and delivered to Owner Trustee a bill of sale (a "Bill of Sale") effective as of the Delivery Date with respect to the Units to be sold by it to Owner Trustee on the Delivery Date, substantially in the form of Exhibit A hereto;
 - (b) Owner Participants and Owner Trustee shall have received an appraisal (the "Appraisal") of the Units from the Appraiser, which

appraisal shall state as of the Delivery Date, (i) the appraised value (the "Appraised Value") of each Unit, which Appraised Value is at least equal to the Purchase Price of such Unit, and (ii) the estimated fair market value of each Unit at the end of the Base Term and each Renewal Term, and which appraisal shall cover such other matters as Owner Participants may request;

(c) Lessee shall have delivered to Owner Trustee a certified copy of the Purchase Agreement;

2

- (d) Lessee and Sublessee shall have entered into the Sublease, and Lessee shall have delivered the original executed chattel paper copy thereof to the Owner Trustee;
- (e) There shall exist no Event of Default or Default that has not been remedied or waived;
- (f) Except as provided in the report on Form 10-K of General Communications, Inc., an Alaska corporation ("GCI"), dated December 31, 1998 (the "Disclosure Document"), there shall have been no material adverse change in the financial condition, operations, business, assets or prospects of GCI, Lessee, Sublessee or any other GCI Entity, and no other event has occurred that has had, or could reasonably be expected to have, a Material Adverse Effect;
- (g) Owner Participants and Owner Trustee shall have received (i) a certificate of the Secretary or an Assistant Secretary of each of Lessee and Sublessee as to resolutions of the Board of Directors of such company (or other evidence satisfactory to Owner Participants and Owner Trustee of the due authorization by such company of the execution, delivery and performance of each Pertinent Document to which it is a party), such company's corporate charter and bylaws, copies of all of which shall be attached to such certificate and certified as true, all evidencing such company's corporate authority to enter into and perform its obligations under each Pertinent Document to which it is a party and (ii) good standing certificates for Lessee and Sublessee from the Secretary of State of Alaska;
- (h) Owner Participants and Owner Trustee shall have received a certificate of the Secretary or an Assistant Secretary of each of Lessee and Sublessee as to the incumbency of the person or persons executing each Pertinent Document on behalf of Lessee and Sublessee, including the signatures of such persons;
- (i) Owner Participants and Owner Trustee shall have received a search of the UCC records against GCI Communication and Lessee from all applicable jurisdictions showing no tax or judgment liens, and no financing statements covering any part of the Collateral;
- (j) Owner Participants and Owner Trustee shall have received evidence that all UCC financing statements and other documents necessary to the satisfaction of the Owner Participant to cause Owner Trustee to have an enforceable and perfected security interest in the Collateral, subject to no liens other than Permitted Liens, have been filed in all appropriate jurisdictions, and all filing fees required to be paid in connection therewith shall have been paid;
- (k) Owner Participants and Owner Trustee shall have received an (i) opinion of in-house counsel to Lessee and Sublessee, substantially in the form of Exhibit C-1

3

hereto, (ii) opinion of special FCC counsel to Lessee and Sublessee, substantially in the form of Exhibit C-2 hereto and (iii) opinion of special counsel to Sublessee, substantially in the form of Exhibit C-3 hereto;

- (1) Lessee shall have delivered to Owner Trustee a certified copy of the Galaxy X Transponder Service Agreement between PanAmSat (as assignee of Hughes Communications Satellite Services, Inc.) and Lessee (as assignee of GCI Communications), dated as of August 24, 1995 (as heretofore or hereafter amended or supplemented, the "Services Agreement");
- (m) Lessee shall have delivered to Owner Trustee (i) a duly executed assignment by Lessee of its rights, claims, warranties and benefits under the Purchase Agreement and the Services Agreement to Owner Trustee in the form of Exhibit D-1 hereto and (ii) a duly executed consent and agreement of PanAmSat to such assignment in the form of Exhibit D-2 hereto;
 - (n) The execution, delivery and performance of this Lease

shall not constitute or result in a violation by Owner Trustee or any Owner Participant of any applicable foreign, federal, state or local law, rule, regulation, treaty or international convention or any judicial or administrative order or decree;

- (o) Final testing of the Satellite and the Units shall have been accomplished, the results of which shall have been satisfactory to Lessee and the Owner Trustee, the Units shall meet the standards set forth in Section 4.03 of the Purchase Agreement and Lessee shall have inspected to its satisfaction and accepted the Units by delivering to Owner Trustee an acceptance certificate (the "Acceptance Certificate") effective as of the Delivery Date, substantially in the form of Exhibit B hereto, whereupon the Units described therein shall immediately become subject to and be governed by all of the provisions of this Lease;
- (p) Lessee shall have delivered to Owner Trustee and each Owner Participant a notice of closing in substantially the form of Exhibit E hereto (the "Closing Notice");
- (q) Each of the representations and warranties set forth in Sections 1.3 and 1.4 shall be true and correct as of the Delivery Date;
- (r) Lessee shall have paid to GECC the structuring fee described in the letter dated March 8, 2000 from GECC to Sublessee and all expenses for which Lessee is responsible pursuant to Section 2.5 with respect to which Lessee has received an invoice;
- (s) Owner Trustee shall have received evidence satisfactory to it that all sales, use and similar taxes required to be paid in connection with the purchase of the Units, or

4

the execution and delivery of this Lease, the Sublease or any other Pertinent Document, have been paid in full; and

(t) Owner Participants and Owner Trustee shall have received such other documents as may be reasonably requested by any Owner Participant or Owner Trustee.

If any of the foregoing conditions in Section 1.2 shall not have been met or waived by all Owner Participants, with respect to all Units on or prior to the date hereof or the Delivery Date, respectively, neither Owner Participants nor Owner Trustee will have any obligation to Lessee to purchase any Unit or to lease any such Unit to Lessee pursuant to this Lease.

- 1.3. Lessee represents and warrants as of the Execution Date and as of the Delivery Date to Owner Participants and Owner Trustee that:
 - (a) To the best of Lessee's knowledge, information and belief, the Appraisal is based on factual assumptions which are reasonable, the information given to the Appraiser by Lessee or its Affiliates was true and correct when given, was not misleading in any material respect when given, and would be true and correct and not be misleading in any material respect if so given as of the Execution Date or the Delivery Date, as the case may be;
 - (b) Lessee is a corporation duly incorporated, validly existing and in good standing under the laws of Alaska, and is qualified and in good standing as a foreign corporation in each jurisdiction where the nature of its business or the location of its properties requires such qualification, except where the failure to be so qualified has not had, and could not reasonably be expected to have, a Material Adverse Effect;
 - (c) Lessee has all requisite corporate power and authority to execute, deliver, and perform its obligations under, each Pertinent Document to which it is a party;
 - (d) The execution and delivery by Lessee of each of the Pertinent Documents to which it is a party and the performance by Lessee of its obligations under each of such Pertinent Documents has been duly authorized by all necessary corporate action (including any necessary stockholder action) on its part, and such execution and delivery have not or do not, and such performance does not and will not, (i) violate any provision of any applicable foreign, federal, state or local law, rule, regulation, treaty or international convention presently in effect having applicability to Lessee or the Units or of any order, writ, judgment, decree, determination or award presently in effect having applicability to Lessee or the Units or of the charter or by-laws of Lessee or (ii) result in a breach of or constitute a default under the Credit Agreement, the Indenture or any other existing indenture, loan or credit agreement, or result in a breach of or constitute a default under the Purchase Agreement or the

Services Agreement, or any other existing agreement or instrument to which Lessee or any of its Affiliates is a party or by which Lessee or any of its Affiliates, respectively, or its properties may be bound or affected, or

.

- (iii) result in, or require, the creation or imposition of any lien of any nature upon or with respect to any of the properties now owned or hereafter acquired by Lessee (other than the lien contemplated by this Lease), and Lessee is not in default under or in violation of its charter or by-laws;
- (e) Each of the Pertinent Documents to which Lessee is a party has been duly executed and delivered by Lessee, and constitutes the legal, valid and binding obligation of Lessee, enforceable against Lessee in accordance with its terms, except as enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law);
- (f) No authorization, consent, approval, license or formal exemption from, nor any filing, declaration or registration with, any court, governmental agency or regulatory authority (Federal, state, local or foreign), including, without limitation, the Federal Communications Commission, the Securities and Exchange Commission, or with any securities exchange, is or will be required in connection with the execution and delivery by Lessee of the Pertinent Documents to which it is a party, or the performance by Lessee of its obligations under the Pertinent Documents to which it is a party;
- (g) Neither any Owner Participant nor Owner Trustee will, solely by reason of its execution, delivery and performance of this Lease, be subject to the regulation or control of the Federal Communications Commission of the United States of America or of any public utility commission or other regulatory agency, which has or may assert its jurisdiction, in whole or in part, over any Unit;
- (h) The chief executive office and principal place of business of Lessee is 2550 Denali Street, Suite 1000, Anchorage, Alaska 99503, or such other office as Lessee shall notify the Owner Participants and Owner Trustee in writing at least 30 days prior to relocation;
- (i) All information regarding the Lessee, any other GCI Entity or the Units heretofore or contemporaneously produced and furnished by or on behalf of Lessee or any other GCI Entity to any Owner Participant or Owner Trustee in connection with the Pertinent Documents or any transaction contemplated thereby, including such information contained in the Disclosure Document, was true and accurate (when taken together as a whole with all other such information furnished to each of the Owner Participants) in all material respects on the date as of which such information is dated or certified and was not incomplete by omitting to state any material fact necessary to make such information (when taken together as a whole with all other such information furnished to each of the Owner Participants) in all material respects not misleading at such time;

6

- (j) The Owner Trustee and Owner Participants have received a complete executed copy of the Purchase Agreement and the Services Agreement, including all attachments, amendments, waivers and side letters or agreements affecting the terms thereof, each of which is in full force and effect and enforceable against the parties thereto in accordance with its terms;
- (k) Except as shown in the Disclosure Document, there is not litigation, proceeding, arbitration or investigation pending or, to the best of Lessee's knowledge, threatened, by or before any court of other governmental authority against Lessee, Sublessee or any other Affiliate of Lessee that questions the validity or enforceability of any Pertinent Document or, if adversely determined, could reasonably be expected to have a Material Adverse Effect;
- (1) Lessee is not engaged in the business of extending credit for the purpose of purchasing or carrying margin stock (within the meaning of Regulation U issued by the Board of Governors of the Federal Resume System); and
- (m) Lessee has filed all federal, state and other tax returns that are required to be filed, and has paid all taxes shown on such returns as well as all other taxes due and payable, except to the extent payment thereof is being contested in good faith and for which adequate reserves have been established in accordance with generally

accepted accounting principles.

- 1.4. Lessee represents and warrants as of the Delivery Date to Owner Participants and Owner Trustee that:
 - (a) Upon payment to PanAmSat of \$44,310,000.00, Owner Trustee will have good and marketable title to each Unit, free from all claims, liens, security interests and encumbrances other than Permitted Liens;
 - (b) No filing, recordation or registration is necessary in order to perfect Owner Trustee's security interest in the Collateral in any jurisdiction except for the filing of UCC financing statements with the UCC Central Office, Department of Natural Resources of the State of Alaska; upon the filing referred to above in this Paragraph (b), Owner Trustee will have an enforceable and perfected security interest in the Collateral subject to no liens other than Permitted Liens;
 - (c) There is sufficient fuel on the satellite (the "Satellite") in which the Units are contained to provide for the orbit described in Section 1.4(e) for a period of at least 12 years from the Delivery Date;
 - (d) The Lessee has made all of the tariff or other regulatory filings, and has all requisite Federal Communications Commission and other licenses and permits, if any (the "Permits"), necessary for the use of the Units as currently contemplated. The Owner

7

Trustee (and all successors and assigns thereof) and the Owner Participants may own the record and/or beneficial interest in and to the Units without any Permits;

- (e) The Satellite has reached its geosynchronous orbit at an orbital position of 123(Degree) West Longitude, or such other orbital slot as the Federal Communications Commission has approved or required;
- (f) Each Unit is operating in a satisfactory manner and all of the Units are operable simultaneously during a period of eclipse in a commercially usable manner (as contemplated in the Purchase Agreement);
- (g) Each Unit is usable for the reception, frequency conversion, amplification and transmission to the ground of communication signals;
- (h) The period of time over which there is predicted to be sufficient solar array power available to operate all Units simultaneously in a commercially usable manner (as contemplated in the Services Agreement and the Purchase Agreement) is not less than 12 years;
- (i) The Lessee and its Affiliates have complied in all material respects with all applicable foreign, federal, state and local laws, rules, regulations, orders, ordinances and other laws of the United States of America and all foreign, state and local governments, agencies, courts or other governmental bodies of any of the foregoing, and all treaties and international conventions, pertaining to the use of Units: and
- (j) Except with respect to permits and agreements related only to the operation and maintenance of earth stations, the Services Agreement is the only agreement, and the Permits are the only licenses and permits, necessary for the use of the Units as they are currently contemplated to be used by Lessee in this Lease and the Services Agreement.
- 1.5. Lessee hereby covenants that:
- (a) it will not amend, modify or waive any provision of the Services Agreement, the Purchase Agreement or the Sublease, without Owner Trustee's prior written consent; it will timely enforce all of its rights and claims, and will take all such action as the Owner Trustee shall request to timely enforce all of its rights and claims, under the Services Agreement, the Purchase Agreement and the Sublease;
- (b) it shall comply with all laws, rules, regulations and orders applicable to it;
- (c) it shall maintain its corporate existence and all qualifications, licenses and permits (including the Permits) necessary for the operation of its business (including the operation of the Units) and the ownership of its properties;

- (d) it will promptly pay and discharge all taxes imposed on it or its properties, unless Lessee is contesting the validity thereof by appropriate proceedings and has established adequate reserves therefor in accordance with generally accepted accounting principles; and
- (e) it will provide Owner Trustee and each Owner Participant certified copies of any amendment, modification or change to the terms of the Services Agreement or the Purchase Agreement promptly upon the execution thereof. Lessee shall duly perform and observe in all material respects each of its obligations under the Services Agreement and Purchase Agreement and Lessee shall diligently defend and enforce its rights under each such agreement.

Section 2. Term, Rent and Payment.

- 2.1. The base term (the "Base Term") of this Lease as to each Unit shall commence on the Delivery Date and continue for a period of one year from the Delivery Date. So long as no Default or Event of Default has occurred and is continuing, Lessee may renew this Lease for all, but not less than all, of the Units at the end of the Base Term and each previous Renewal Term for a period of one year (each, a "Renewal Term", and together with the Base Term, the "Term"), provided that in no event shall the Term exceed nine (9) years from the Delivery Date. Lessee may exercise its option to extend this Lease for a Renewal Term by delivering written notice of such exercise to the Owner Trustee not less than three (3) months prior to the then scheduled termination date, which notice shall be irrevocable. At the end of the Term, Lessee shall fulfill its obligations set forth in Paragraph E in Schedule II.
- 2.2. Lessee shall pay to Owner Trustee for the account of Owner Participants rental for each Unit in the amounts and at the times set forth in Paragraph C of Schedule II hereto.
- 2.3. Rent and all other sums due Owner Trustee hereunder shall be paid in immediately available funds at the office of Owner Trustee specified for receipt of notices in Section 21 (the "Owner Trustee's Office"). Sums due directly to an Owner Participant hereunder shall be paid in immediately available funds: (i) if such sum is a payment under Section 5 or 7.2, to such Owner Participant at such office of that Owner Participant as it from time to time specifies to Lessee herein or in a notice pursuant to this Lease; and (ii) if such sum is a payment due under any other provision of this Lease, to Owner Trustee as provided in the foregoing sentence. Notwithstanding the foregoing, so long as GECC is the only Owner Participant, GECC and the Owner Trustee hereby direct Lessee to make payment of rent and other amounts due to the Owner Trustee at GECC's account set forth on Schedule I hereto.
- 2.4. This Lease is a net lease and Lessee's obligation to pay all rental, indemnity and other amounts payable hereunder shall be absolute and unconditional under any and all circumstances and, without limiting the generality of the foregoing, Lessee shall not be entitled to any abatement or reduction of rent or any setoff against rent, indemnity or other amount,

q

whether arising by reason of any past, present or future claims of any nature by Lessee against any Owner Participant or Owner Trustee or the registered holder of any interest in this Lease, or otherwise, or any right to delay the payment thereof. Except as otherwise expressly provided herein, this Lease shall not terminate, nor shall the obligations of Lessee be otherwise affected, (a) by reason of any defect in, damage to, or loss of possession or use, obsolescence or destruction, of any or all of the Units, however caused, or (b) by the taking or requisitioning of any or all of the Units or the Satellite by condemnation or otherwise, or (c) by the invalidity or unenforceability or lack of due authorization by any Owner Participant, Lessee or Owner Trustee or other infirmity of this Lease, or (d) by lack of power or authority of any Owner Participant or Owner Trustee to enter into this Lease, or (e) by the attachment of any lien, encumbrance, security interest or other right or claim of any third party to any Unit, or (f) by any prohibition or restriction of or interference with Lessee's use of any or all of the Units by any person or entity, or (g) by the insolvency of or the commencement by or against Lessee, Owner Trustee or any Owner Participant of any bankruptcy, reorganization or similar proceeding, or (h) by any act of God, force majeure or other cause, whether similar or dissimilar to the foregoing, any present or future law to the contrary notwithstanding. It is the intention of the parties that all rent, indemnities and other amounts payable by Lessee hereunder shall be payable in all events in the manner and at the times herein provided unless Lessee's obligations in respect thereof have been terminated or modified pursuant to the express provisions of this Lease. To the extent permitted by applicable law, Lessee waives any and all rights which it may now have or which may at any time be conferred upon it, by statute or otherwise, to terminate, cancel, quit or surrender this Lease, in whole or in part, except strictly in accordance with the express terms hereof. Each rental, indemnity or other payment made by Lessee hereunder shall be final, and Lessee shall not seek to recover (except as expressly provided in this Lease) all or any part of such payment from any Owner Participant or Owner Trustee for any reason whatsoever.

- 2.5. Lessee shall pay promptly upon receipt of an invoice therefor all reasonable out-of-pocket costs and expenses of each Owner Participant and Owner Trustee (including the reasonable fees and disbursements of Mayer, Brown & Platt, as special counsel to Owner Participant and the reasonable fees and disbursements of Shipman & Goodwin LLP, as special counsel to the Owner Trustee) in connection with
 - (a) the negotiation, preparation, execution and delivery of the Pertinent Documents and schedules, certificates and exhibits thereto (including, without limitation, appraisal expenses), or entering into, or giving or withholding of, any future amendments, waivers, consents, supplements or other modifications to any of the Pertinent Documents, whether or not such amendment, modification, consent or waiver is executed and delivered,
 - $% \left(h\right) =\left(h\right) =\left(h\right) +h^{2}$ (b) any Casualty Occurrence or early termination under this Lease,
 - (c) any Event of Default or the enforcement or preservation of any rights or remedies of any Owner Participant or Owner Trustee under the Pertinent Documents,

10

- $\,$ (d) the preparation and review of the form of any document or instrument relevant to the Pertinent Documents,
- (e) all other administrative fees and expenses payable to third parties relating to the Pertinent Documents, including the initial and on-going fees of the Owner Trustee, or
- (f) any transfer (i) by Owner Trustee of any of its interests in this Lease to any successor or (ii) by any Owner Participant of any of its interests in the Trust Estate if any Event of Default exists.

Lessee shall indemnify, defend and hold harmless Owner Trustee and each Owner Participant against any and all costs, expenses, damages and liabilities paid, incurred or asserted to be due for or in connection with any and all brokerage or finder's fees and commissions payable or claimed to be payable to any Person in connection with any of the transactions contemplated by the Pertinent Documents.

Section 3. Warranties.

- 3.1. LESSEE ACKNOWLEDGES AND AGREES THAT (a) EACH UNIT IS OF A SIZE, DESIGN, CAPACITY AND MANUFACTURE SELECTED BY LESSEE, (b) LESSEE IS SATISFIED THAT THE SAME IS SUITABLE FOR ITS PURPOSES, (c) NEITHER ANY OWNER PARTICIPANT NOR OWNER TRUSTEE IS A MANUFACTURER THEREOF OR A DEALER IN PROPERTY OF SUCH KIND, AND (D) NEITHER ANY OWNER PARTICIPANT NOR OWNER TRUSTEE HAS MADE, OR DOES OR WILL MAKE, (I) ANY REPRESENTATION OR WARRANTY OR COVENANT WITH RESPECT TO THE TITLE, MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, CONDITION, QUALITY, DESCRIPTION, DURABILITY OR SUITABILITY OF ANY SUCH UNIT IN ANY RESPECT OR IN CONNECTION WITH OR FOR THE PURPOSES AND USES OF LESSEE, OR (ii) ANY OTHER REPRESENTATION OR WARRANTY WHATSOEVER, EXPRESS OR IMPLIED, WITH RESPECT TO ANY UNIT, IT BEING AGREED THAT ALL RISKS, AS BETWEEN OWNER PARTICIPANTS AND OWNER TRUSTEE, ON THE ONE HAND, AND LESSEE, ON THE OTHER HAND, SHALL BE BORNE BY LESSEE.
- 3.2. Lessee acknowledges and agrees that neither any Owner Participant nor Owner Trustee has made any representations and warranties concerning the tax, accounting or legal characteristics of this Lease and that Lessee has obtained and relied on such tax, accounting and legal advice regarding this Lease as it deems appropriate.
- Section 4. Possession, Assignment, Sublease, Use and Maintenance.
- 4.1. Lessee shall not: (a) use, operate or maintain any Unit (i) except in accordance with the provisions of the Services Agreement, the Purchase Agreement and any other applicable contract or agreement and prudent industry practices or (ii) in violation of any applicable

11

insurance policy or law or regulation of any governmental authority; (b) except as required or permitted by Section 4.2, sublease or assign any Unit or permit the use thereof by anyone other than Lessee and Sublessee pursuant to the Sublease; or (c) except as set forth in Sections 4.2 and 24.2, sell, assign or transfer any of its rights hereunder or in any Unit, or directly or indirectly create, incur or suffer to exist any lien, claim, security interest or encumbrance of any kind on any of its rights hereunder or in any Unit, except for Permitted Liens. Lessee will defend the rights, title and interest of Owner Trustee in and to the Units against the claims or demands of all persons.

4.2. Lessee may assign and delegate all (but not less than all) of its

rights, obligation and interest under this Lease only with the prior written consent of Owner Trustee and all of the Owner Participants. Lessee may sublease the Units to Sublessee pursuant to the Sublease or to another Restricted Subsidiary pursuant to a sublease in substantially the form of the Sublease, in each case without the consent of the Owner Trustee or any Owner Participant (provided that Lessee shall provide prompt notice of any such sublease to Owner Trustee). So long as no Event of Default or Default has occurred and is continuing, Lessee may sublease, assign or transfer the use of one or more Units (or portions of a Unit) to persons other than Lessee, Sublessee or to a Restricted Subsidiary; provided that, in the case of a sublease, assignment or transfer of use of one or more Units (or portions of a Unit) to any person or entity (other than to Sublessee pursuant to the Sublease or a Restricted Subsidiary), the written consent of the Deciding Interest of Owner Participants is obtained by Lessee prior to such sublease, assignment or transfer, which consent shall not be unreasonably withheld; and provided, further, that in the case of a sublease, assignment or transfer of use of one or more Units (or portions of a Unit) to any person or entity, including to Sublessee or a Restricted Subsidiary (i) no Unit may be subleased, assigned or used by another person for a period beyond the Term of this Lease, (ii) the sublease, assignment or use agreement must be in writing and expressly subject and subordinate to this Lease, and (iii) Lessee shall remain primarily liable hereunder for the performance of all of the terms of this Lease to the same extent as if such agreement had not been entered into, and, except as set forth in the Sublease, such agreement will expressly prohibit any further assignment, sublease or transfer.

- 4.3. Lessee will cause the Units to be maintained, managed and monitored, ordinary wear and tear excepted, in accordance with the Services Agreement and the Purchase Agreement, and in compliance with all management and monitoring obligations, all applicable requirements of law and of any governmental authority or agency having jurisdiction over such Units or Lessee, and in accordance with standards no lower than the highest of those applied by Lessee or any of its Affiliates to similar satellite transponders owned or leased by it, ordinary wear and tear excepted. Lessee will maintain or cause to be maintained, and shall permit Owner Trustee or any Owner Participant to inspect, any records, logs and other materials required by any governmental authority having jurisdiction to be maintained or filed in respect of any Unit.
- 4.4. Lessee will not create, incur, assume or suffer to exist any claim, lien, security interest or encumbrance upon any of its interest under the Services Agreement or any other Collateral, other than Permitted Liens. Lessee agrees that it will, within sixty (60) days after the

12

last day of each calendar quarter occurring during the term of this Lease, furnish to Owner Trustee and each Owner Participant a certificate of a Responsible Officer of Lessee certifying that, as of such date, Lessee is in compliance with the requirements of this Section.

4.5. In connection with any exercise of remedies, Owner Trustee and each of the Owner Participants agrees that it will not take any action to control, maintain, operate or use any Unit or to cause or permit such Unit to be controlled, maintained, operated or used by any person other than Lessee or Sublessee until all necessary approvals of the Federal Communications Commission and the Regulatory Commission of Alaska have been obtained.

Section 5. General Tax Indemnity.

5.1. Lessee agrees to pay or reimburse each Owner Participant and Owner Trustee for, and to indemnify and hold each Owner Participant and Owner Trustee harmless from, all fees (including, but not limited to, license, documentation, recording or registration fees), and taxes (including but not limited to all sales, use, gross receipts, personal property, occupational, value added or other taxes, levies, imposts, duties, assessments, charges or withholdings of any nature whatsoever), together with any penalties, fines or additions to tax, or interest thereon (all of the foregoing being hereafter referred to as "Impositions"), arising at or relating to any time prior to or during the term of this Lease, or upon any termination of this Lease or prior to or upon the return of the Units to any Owner Participant or Owner Trustee in accordance with this Lease, and levied or imposed upon any Owner Participant or Owner Trustee, directly or otherwise, by any Federal, state or local government or taxing authority in the United States or by any foreign country or foreign or international taxing authority upon or with respect to (a) any Unit, (b) the exportation, importation, registration, purchase, ownership, delivery, condition, lease, assignment, storage, transportation, possession, use, operation, maintenance, repair, return, sale (including to Owner Trustee or Lessee pursuant to this Lease), transfer of title or other disposition thereof, (c) the rentals, receipts, or earnings arising from any Unit, or (d) this Lease or any other Pertinent Document or any payment made hereunder or thereunder; provided that this Section 5 shall not apply to: (i) Impositions which are based upon or measured by any Owner Participant's or Owner Trustee's net income, or which are in substitution for, or relieve any Owner Participant or Owner Trustee from, any actual Imposition based upon or measured by such Owner Participant's or Owner Trustee's net income (excluding, however, (A) any Imposition based on,

or measured by the net income of each Owner Participant or Owner Trustee imposed by the United States or any state or local government taxing authority in the United States to the extent they would not have been imposed on an Owner Participant if on the Delivery Date such Owner Participant had advanced funds in the form of a loan secured by the Units to Lessee equal to the aggregate Purchase Price of the Units, debt service for such loan equal to scheduled rent was paid from time to time and a principal balance in the amount of the Balloon Payment remaining at the end of the term of this Lease was due at the end of such term, which treatment is consistent with the intent of Owner Participant and Lessee that Lessee be the beneficiary of the tax benefits for the Units for tax purposes, including, without limitation, the ability to claim depreciation on the Units and to deduct the interest component of the rentals, and (B) Impositions imposed with respect to the payment, receipt or accrual of any indemnity

13

payment hereunder); (ii) Impositions based upon the voluntary transfer, assignment or disposition by any Owner Participant or Owner Trustee of any interest in any of the Units other than a transfer pursuant to the exercise of remedies hereunder or a transfer to the Lessee; and (iii) except where there exists an Event of Default, Impositions in respect of any Unit arising after the expiration or earlier termination of the Lease in respect of such Unit and the return or other disposition of such Unit in full compliance with the terms of this Lease, provided that such Impositions do not relate to acts or events arising or occurring prior to or coincident with such time.

- 5.2. Lessee agrees to pay on or before the time or times prescribed by law any Impositions (except any Impositions excluded by Section 5.1); provided, however, that Lessee shall be under no obligation to pay any such Imposition so long as (i) such Imposition is being contested in good faith and by appropriate legal proceedings, with appropriate reserves being maintained therefor in accordance with generally accepted accounting principles, consistently applied; and (ii) the nonpayment thereof does not, in the reasonable opinion of any Owner Participant or Owner Trustee, materially adversely affect the right of any Owner Participant or Owner Trustee to receive payment of rentals or the Balloon Payment or involve a material risk of sale, forfeiture or loss of the Units or any interest therein or materially adversely affect the security interest created by this Lease. If any Impositions (except any Impositions excluded by Section 5.1) shall have been charged or levied against any Owner Participant or Owner Trustee directly and paid by such Owner Participant or Owner Trustee, Lessee shall immediately reimburse such Owner Participant or Owner Trustee, as appropriate, on presentation of an invoice therefor; provided, however, if such payment of any Imposition is made by any Owner Participant or Owner Trustee for which such Owner Participant or Owner Trustee shall have been reimbursed later becomes the subject of any refund (or like adjustment) paid to such Owner Participant or Owner Trustee, such Owner Participant or Owner Trustee shall promptly remit the amount of such refund (or like adjustment) to Lessee.
- 5.3. If the applicable Owner Participant or Owner Trustee shall not be entitled to a corresponding and equal deduction with respect to any Imposition which Lessee is required to pay or reimburse under Section 5.1 or 5.2 and which payment or reimbursement constitutes income to any Owner Participant or Owner Trustee, then Lessee shall also pay to such Owner Participant or Owner Trustee, as appropriate, on demand the amount of such Imposition on a gross-up basis such that, after subtracting all Impositions imposed on such Owner Participant or Owner Trustee with respect to such payment by Lessee (including any Impositions otherwise excluded by Section 5.1 and assuming for this purpose that such Owner Participant or Owner Trustee was subject to taxation at the applicable federal, state or local marginal rates) such Owner Participant or Owner Trustee shall be fully reimbursed for the Imposition with respect to which such Owner Participant or Owner Trustee is entitled to be paid or reimbursed.
- 5.4. Lessee shall properly prepare and file any reports or returns which may be required with respect to the Units or this Lease. If any Owner Participant or Owner Trustee is obligated to file any such reports or returns, then Lessee shall at least 10 days before the same are due prepare the same and forward them to such Owner Participant or Owner Trustee, as

14

appropriate, with detailed instructions as to how to comply with all applicable filing requirements and funds in the amount of any payment required pursuant thereto.

Section 6. Increased Costs, Etc.

(a) Illegality. Notwithstanding any other provision herein, if any Regulatory Change shall make it unlawful for any Owner Participant to fund its portion of the Balance Due at the Adjusted LIBOR Rate, as contemplated by this Lease, interest on the Balance Due shall automatically be calculated at the Base Rate on the last day of the then current Interest Period or such earlier date as required by law. If any such conversion occurs on a day which is not the last day of the then current Interest Period with respect thereto, Lessee shall pay to

the Owner Participants such amounts, if any, as may be required pursuant to Paragraph (d) below.

- (b) Requirements of Law. In the event that any Regulatory Change or compliance by any Owner Participant with any request or directive (whether or not having the force of law) from any central bank or other governmental authority made subsequent to the date hereof:
 - (i) shall subject any Owner Participant to any tax of any kind whatsoever with respect to this Lease or the funding made by it, or change the basis of taxation of payments to such Owner Participant in respect thereof (except for taxes covered by Section 5.1);
 - (ii) shall impose, modify or hold applicable any reserve, special deposit, compulsory loan or similar requirement against assets held by, deposits or other liabilities in or for the account of, advances, loans or other extensions of credit by, or any other acquisition of funds by, any office of such Owner Participant which is not otherwise included in the determination of the Adjusted LIBOR Rate; or
 - (iii) shall impose on such Owner Participant
 condition;

and the result of any of the foregoing is to increase the cost to such Owner Participant, by an amount which such Owner Participant deems to be material, of making, continuing or maintaining its portion of the Balance Due or to reduce any amount receivable hereunder in respect thereof then, in any such case, Lessee shall promptly pay such Owner Participant, upon its demand, any additional amounts necessary to compensate such Owner Participant for such increased cost or reduced amount receivable. If any Owner Participant becomes entitled to claim any additional amounts pursuant to this subsection, it shall promptly notify the Lessee of the event by reason of which it has become so entitled. A certificate as to any additional amounts payable pursuant to this subsection submitted by such Owner Participant to Lessee in good faith and setting forth in reasonable detail the calculation of such amounts shall be conclusive in the

15

absence of manifest error. The provisions of this Paragraph (b) shall survive the termination of this Lease and the payment of the Balance Due and all other amounts payable hereunder.

- (c) Capital Adequacy. In the event that any Owner Participant or corporation controlling such Owner Participant shall have determined that any Regulatory Change regarding capital adequacy or in the interpretation or application thereof or compliance by such Owner Participant or such corporation with any request or directive regarding capital adequacy (whether or not having the force of law) from any governmental authority made subsequent to the date hereof does or shall have the effect of reducing the rate of return on such Owner Participant's capital as a consequence of its obligations or investment hereunder to a level below that which such Owner Participant could have achieved but for such change or compliance (taking into consideration such Owner Participant's policies with respect to capital adequacy) by an amount deemed by such Owner Participant to be material, then from time to time, after submission by such Owner Participant in good faith to Lessee (with a copy to the Owner Trustee) of a written request therefor setting forth in reasonable detail the calculation of such amount (which request shall be conclusive in the absence of manifest error), Lessee shall pay to such Owner Participant such additional amount or amounts as will compensate such Owner Participant for such reduction. The provisions of this Paragraph (c) shall survive the termination of this Lease and the payment of the Balance Due and all other amounts payable hereunder.
- (d) Breakage Costs. Lessee agrees to indemnify each Owner Participant and to hold each Owner Participant harmless from any loss or expense which such Owner Participant may sustain or incur as a consequence of (i) any rental payment or the Balance Due or any portion thereof, being paid on any day other than the last day of an Interest Period or (ii) the making of a prepayment or payment of any portion of the Balance Due on a day which is not the last day of an Interest Period, including, without limitation, in each case, any such loss (other than non-receipt of anticipated profits) or expense arising from the reemployment of funds obtained by it or from fees payable to terminate the deposits from which such funds were obtained (it being

understood that any such calculation will be made on notional amounts as the Funding Parties are not required to show that they matched deposits specifically). A certificate as to any additional amounts payable pursuant to this subsection submitted by such Owner Participant to Lessee in good faith shall be conclusive in the absence of manifest error. The provisions of this Paragraph (d) shall survive the termination of this Lease and the payment of the Balance Due and all other amounts payable hereunder.

Section 7. Risk of Loss; Replacement; Waiver and Indemnity.

7.1. (a) If a Casualty Occurrence with respect to a Unit takes place, the Lessee shall provide to Owner Trustee a Casualty Notice on or before the next Payment Date but in any event within fifteen (15) days of the date thereof, and, on the Payment Date next following the date of such Casualty Occurrence, the Lessee shall purchase such Unit on an as is, where is basis,

16

without recourse, representation or warranty (except that such Unit is free and clear of liens caused by actions of the Owner Participants), express or implied, from the Owner Trustee or the Owner Participants, at a price equal to the Casualty Amount with respect to such Unit determined as of such Payment Date, together with any rental payment then due, provided, however, that if such Casualty Occurrence occurs thirty (30) days or less prior to such Payment Date, the Lessee shall purchase such Unit on the second succeeding Payment Date after such Casualty Occurrence at a price equal to the Casualty Amount with respect to such Unit, determined as of such Payment Date and the portion of the next scheduled rental payment allocable to the terminated Units. Upon paying such Casualty Amount in respect thereof and making all rental payments due and owing with respect thereto, Owner Trustee shall transfer title to such Unit to the Lessee, and the Lessee's obligation to pay further rent for such Unit shall cease.

Any payments received at any time by Owner Trustee or Owner Participants pursuant to the Services Agreement or the Purchase Agreement as a result of the occurrence of any Casualty Occurrence for which a Casualty Notice has been received shall be applied as follows: first, in reduction of the Lessee's obligation to pay the Casualty Amount (to the extent not already paid to the Owner Trustee at the time any such payments are received) and rental payments required to be made by the Lessee pursuant to this Section, to the extent not already paid by the Lessee; second, to the extent already paid by the Lessee, to reimburse the Lessee for its payment of the Casualty Amount; and third, the balance, if any, of such payments remaining thereafter shall be paid to the Lessee. Any payments received at any time by Owner Trustee or Owner Participants pursuant to the Services Agreement or the Purchase Agreement with respect to which Owner Trustee has received written confirmation from Lessee that such payment is not a result of a Casualty Occurrence, shall be promptly paid to Lessee.

Until the payment of such Casualty Amount by Lessee, as described herein, Lessee shall continue to make all rental payments due hereunder. Upon the making of such payment by Lessee in respect of any Unit, each remaining scheduled rental payment and the Balloon Payment shall be reduced by an amount equal to the product of the amount of such scheduled rental payment or such Balloon Payment (prior to the receipt of such payment by Owner Participants) multiplied by the Unit Value Fraction.

In the case of a Requisition of Use with respect to a Unit which does not constitute a Casualty Occurrence, such Requisition of Use shall not terminate this Lease with respect to such Unit and each and every obligation of the Lessee with respect thereto shall remain in full force and effect. So long as no Default or Event of Default shall have occurred and be continuing under this Lease, Lessee shall be entitled to all sums, attributable to the period such Unit is subject to this Lease, received by reason of any such Requisition of Use referred to in the preceding sentence.

(b) Except as hereinabove in this Section 7.1 provided, Lessee shall not be released from its obligations hereunder in the event of, and shall bear the risk of, any Casualty

17

Occurrence to any Unit prior to or during the Term of this Lease and thereafter until all of Lessee's obligations hereunder are fully performed.

(c) Definitions.

"Balance Due" shall mean on any date the Aggregate Purchase Price, minus the aggregate Amortization Amounts paid by Lessee on or prior to such date.

"Casualty Amount" shall mean the product of (x) the then Balance Due multiplied by (y) the Unit Value Fraction.

"Casualty Occurrence" with respect to a Unit shall mean any of the following events: (i) loss of such Unit or of the use thereof, including, without limitation, disappearance, loss, theft, destruction, or damage beyond repair of such Unit, or rendition of such Unit permanently unfit for commercial use or operation; (ii) an insurance settlement with respect to such Unit on the basis of a total loss; (iii) a Requisition of Use for an indefinite period or for a stated period extending for six (6) consecutive months or more or extending beyond the end of the Term of the Lease; (iv) the use of such Unit or the Satellite in the normal course of business shall have been prohibited for a period of six (6) consecutive months or more or extending beyond the end of the Term of the Lease as a result of any rule, regulation, order or other action by any governmental body having jurisdiction; (v) such Unit becomes a "Failed Transponder" pursuant to the Purchase Agreement; (vi) a Satellite Failure or the Satellite shall at any time have less than four (4) pounds of fuel on board and available for use; or (vii) one or more Owner Participants shall be subject to regulation under the Public Utility Holding Company Act of 1935, as amended, or be deemed to be a "common carrier" subject to the jurisdiction of the Federal Communications Commission (except for nonburdensome direct or derivative obligations to furnish information, which the Owner Participants agree to furnish so long as the costs and expenses thereof are to be paid by the Lessee) under the Communications Act of 1934 by reason of their participation in the transactions contemplated by the Pertinent Documents. The date of such Casualty Occurrence shall be the date of such disappearance, loss, theft, destruction, damage, Requisition of Use for the stated period, prohibition of use for the stated period, failure or regulation.

"Requisition of Use" with respect to a Unit shall mean the taking, condemnation, confiscation or seizure of, requisition of title to or use of, such Unit by the act of the United States or any foreign government or any state or local authority within the United States or any instrumentality or agency of any of the foregoing.

"Satellite Failure" shall mean the failure of one or more of the basic subsystems on the Satellite, or any other event, rendering the Satellite commercially unusable.

"Unit Value Fraction" shall mean a fraction determined as of the payment date the numerator of which is the Purchase Price for the Unit as to which there has been a Casualty Occurrence (which Purchase Price shall vary depending on whether such Unit was a C-Band or a

18

Ku-Band Transponder) and the denominator of which is the Aggregate Purchase Price of all Units then subject to this Lease, including the Unit having suffered the Casualty Occurrence.

- (d) In the event that PanAmSat provides a "Transponder Spare" or a "Reserve Transponder", as each such term is defined in the Purchase Agreement, as a substitute for any Unit, such Transponder Spare or Reserve Transponder shall be substituted for such Unit under this Lease, and shall be considered a Unit for all purposes hereof, without any further act. Lessee shall promptly notify Owner Trustee of any such substitution, and shall take such action, and shall execute and deliver such documents, as Owner Trustee may request in order to protect or perfect Owner Trustee's interest in and to such substitute Unit.
- 7.2. Lessee waives and releases any claim now or hereafter existing against any Owner Participant or Owner Trustee on account of, and shall indemnify, reimburse and hold each Owner Participant and Owner Trustee harmless from, any and all claims by third parties (including, but not limited to, claims relating to trademark or patent infringement and claims based upon negligence, strict liability in tort, violation of laws, statutes, rules, codes or orders or claims arising out of any loss or damage to any property or death or injury to any person), and any and all losses, damages or obligations owing to third parties, and any and all penalties, liabilities, demands, suits, judgments and causes of action, and all legal proceedings, and any and all costs or expenses in connection therewith (including attorneys' fees and expenses incurred by any Owner Participant and Owner Trustee), which may be imposed on, incurred by or asserted against any Owner Participant or Owner Trustee in any way relating to or arising in any manner out of (i) the registration, purchase, taking or foreclosure of a security interest in, ownership, delivery, condition, lease, sublease, assignment, storage, transportation, possession, use, operation, return or other disposition of any Collateral, including any Unit, or any defect in any such Collateral, arising from the material or any article used therein or from the design, testing or use thereof, or from any maintenance, service, repair, overhaul or testing of any such Collateral regardless of when such defect shall be discovered, whether or not such Collateral is in the possession of Lessee and no matter where it is located; (ii) any Pertinent Document, the enforcement thereof or the consummation of the transactions contemplated thereby; or (iii) any investigation, litigation or proceeding involving or connected with any Unit or the use thereof which is related to any environmental cleanup, audit, compliance or other matter relating to the protection of the environment; provided that Lessee shall not be obligated to indemnify any Owner Participant or Owner Trustee for any such claim, loss, damage or liability which a court of competent jurisdiction shall have determined resulted from such Owner

Participant's or Owner Trustee's own gross negligence or willful misconduct or from acts or events that occur with respect to any Unit after the expiration of the Term of this Lease with respect to such Unit and return of such Unit in accordance with the terms hereof (other than product liability claims (whether in contract or tort) and tort claims); provided, however, that nothing in the preceding proviso shall be deemed to exclude or limit any claim that any indemnified person may have under this Lease or applicable laws for damages from the Lessee for breach of the Lessee's representations, warranties or covenants contained herein.

19

Section 8. [Reserved.]

Section 9. Default.

9.1. The following shall constitute events of default (each an "Event of Default") hereunder:

- (a) default, and continuance thereof for three Business Days, in the payment when due of any amount owing by Lessee under Schedule II hereto; or default, and continuance thereof for five Business Days, in the payment when due of any other amount owing by Lessee hereunder or under any other Pertinent Document; or
- (b) any representation or warranty on the part of Lessee contained in any Pertinent Document or in any Officer's Certificate, estoppel certificate or other writing or instrument delivered pursuant to this Lease shall at any time prove to have been incorrect in any material respect when made, or reaffirmed, as the case may be; or
- (c) Lessee shall default in any material respect in the performance or observance of any term, covenant, condition or agreement on its part to be performed or observed under Section 4.2, or Section 24.2; or
- (d) Lessee shall default in any material respect in the performance or observance of any other term, covenant, condition or agreement on its part to be performed or observed hereunder (and not constituting an Event of Default under any other clause of this Section 9.1), and such default shall continue unremedied for 30 days after the earlier of (i) the date on which written notice thereof shall have been given to Lessee by Owner Trustee or any Owner Participant, and (ii) the date on which a Responsible Officer of Lessee has actual knowledge of such default; or
- (e) either (i) Lessee, Sublessee or any other GCI Entity shall generally fail to pay, or admit in writing its inability to pay, its debts as they become due, or shall voluntarily commence any case or proceeding or file any petition under any bankruptcy, insolvency or similar law or seeking dissolution, liquidation or reorganization or the appointment of a receiver, trustee, custodian or liquidator for itself or a substantial portion of its property, assets or business or to effect a plan or other arrangement with its creditors, or shall file any answer admitting the jurisdiction of the court and the material allegations of any involuntary petition filed against it in any bankruptcy, insolvency or similar case or proceeding, or shall be adjudicated bankrupt, or shall make a general assignment for the benefit of creditors, or shall consent to, or acquiesce in the appointment of, a receiver, trustee, custodian or liquidator for itself or a substantial portion of its property, assets or business, or (ii) corporate action shall be taken by Lessee, Sublessee or any other GCI Entity for the purpose of effectuating any of the foregoing; or

20

- (f) involuntary proceedings or an involuntary petition shall be commenced or filed against the Lessee, Sublessee or any other GCI Entity under any bankruptcy, insolvency or similar law or seeking the dissolution, liquidation or reorganization of Lessee, Sublessee or such GCI Entity (as the case may be) or the appointment of a receiver, trustee, custodian or liquidator for the Lessee, Sublessee or such GCI Entity (as the case may be) or of a substantial part of the property, assets or business of Lessee, Sublessee or such GCI Entity (as the case may be), or any writ, judgment, warrant of attachment, execution or similar process shall be issued or levied against a substantial part of the property, assets or business of Lessee, Sublessee or any other GCI Entity, and such proceedings or petition shall not be dismissed, or such writ, judgment, warrant of attachment, execution or similar process shall not be released, vacated or fully bonded, within 60 days after commencement, filing or levy, as the case may be; or
- $\mbox{(g)}$ any of the Collateral shall be subject to a lien, claim or encumbrance, other than a Permitted Lien; or

- (h) a default shall occur in the payment when due (subject to any applicable grace period), whether by acceleration or otherwise, of any Indebtedness (other than obligations described in clause (a) of this Section 9.1) or in the payment of any Contingent Obligation of Lessee or any GCI Entity having a principal amount, individually or in the aggregate, equal to or in excess of \$1,000,000 and/or a default shall occur in the performance or observance of any other obligation or condition with respect to such Indebtedness if (i) the effect of such default is (x) to accelerate the maturity of any such Indebtedness or (y) to permit the holder or holders of such Indebtedness, or any trustee or agent for such holders, to cause such Indebtedness to become or be declared due prior to its stated maturity or (ii) the holder or holders of such Indebtedness to become due and payable prior to its expressed maturity; or
- (i) one or more judgments or decrees shall be entered against Lessee and/or any GCI Entity involving in the aggregate a liability of \$1,000,000 or more and all such judgments or decrees shall not have been vacated, discharged, stayed or bonded pending appeal within 30 days from the entry thereof; or
- (j) any "Event of Default" shall occur under the Sublease, the Credit Agreement or the Indenture; or
- (k) any of the Pertinent Documents shall become invalid or unenforceable for any reason, or any party to any Pertinent Document shall repudiate its obligations thereunder, or question the validity or enforceability thereunder, or Owner Trustee shall cease to have a first priority perfected security interest in the Collateral or any part thereof.

21

- 9.2. If any Event of Default has occurred and is continuing, Owner Trustee shall have the rights, options and remedies of a secured party under the U.C.C. (regardless of whether the U.C.C. or a law similar thereto has been enacted in a jurisdiction wherein the rights or remedies are asserted), and, without limiting the foregoing, Owner Trustee also may exercise in any order one or more or all of the remedies hereinafter set forth (it being understood that no remedy herein conferred is intended to be exclusive of any other remedy or remedies, but each and every remedy shall be cumulative and shall be in addition to every other remedy given herein or now or hereafter existing at law or in equity or by statute):
 - (a) Owner Trustee may proceed by appropriate court action or actions either at law or in equity, to enforce performance by Lessee of the applicable covenants of this Lease or to recover damages for the breach thereof; or
 - (b) Owner Trustee may, by notice in writing to Lessee terminate this Lease (provided that, in the case of an Event of Default described in Section 9.1(e) or 9.1(f), this Lease shall automatically terminate), but Lessee shall remain liable as hereinafter provided; and Owner Trustee may, at its option, do any one or more of the following: (i) declare the Balance Due with respect to the Units, and any and all other amounts which under any Pertinent Document are payable and unpaid, to be immediately due and payable (provided that, in the case of an Event of Default described in Section 9.1(e) or 9.1(f), the Balance Due and all such other amounts shall automatically and without any notice or declaration, become immediately due and payable), and recover any damages and expenses (including the costs and expenses described in Sections 2.5 and 7.2) in addition thereto which any Owner Participant or Owner Trustee shall have sustained by reason of the breach of any covenant, representation or warranty contained in this Lease; (ii) enforce the security interest given hereunder pursuant to the U.C.C. or any other law; (iii) enter upon the premises where any of the Collateral may be and take possession of all or any of such Collateral; and (iv) require Lessee to assemble and return the Units as provided in Section 10.

In connection with any exercise of remedies, Lessee shall, at the request of the Owner Trustee, take all actions within Lessee's power to cause to be transferred to the Owner Trustee all Permits or other rights or privileges necessary to use the Units.

9.3. In addition to the remedies set forth in Section 9.2, if any Event of Default shall occur and be continuing, Owner Trustee may, but is not required to, sell the Collateral in one or more sales. Any Owner Participant and Owner Trustee may purchase all or any part of the Collateral at such sale. Lessee acknowledges that sales for cash or credit to a wholesaler, retailer or user of such Collateral, or at public or private auction, are all commercially reasonable. Any notice required by law of intended disposition by Owner Trustee shall be deemed reasonably and properly given if given at least 10 days before such disposition. If the sale of all or any part of the Collateral is made on

credit or for future delivery, Owner Trustee shall not be required to apply any portion of the sale price to the obligations of Lessee hereunder or under the related documents to which it is a party until such amount actually is received by Owner Trustee, and

2

any Collateral so sold may be retained by Owner Trustee until the sale price is paid in full by the purchaser or purchasers thereof.

Upon the payment in full by or on behalf of Lessee of all obligations of Lessee due and payable under this Lease, Owner Trustee shall execute and deliver to Lessee or its assignee or nominee, a bill of sale (without representations or warranties except that the Units are free and clear of all claims, liens, security interests and other encumbrances by or in favor of any person claiming by, through or under Owner Trustee or Owner Participants) with respect to any Collateral that has not been sold or otherwise transferred in accordance with this Section 9.3.

- 9.4. The proceeds of such sale or exercise of other remedies shall be applied in the following order:
 - (a) First, to the payment of reasonable costs and expenses of each Owner Participant and Owner Trustee in administering the Lease and in exercising remedies, including expenses of foreclosure or suit, if any, and of any sale, and of all other proper fees, expenses, liabilities and advances (including reasonable legal expenses and attorneys' fees, whether of special counsel or in respect of allocated time charges of internal counsel) of each Owner Participant and Owner Trustee and of all taxes, assessments or liens superior to the lien granted hereunder, except any taxes, assessments or superior lien subject to which any sale of Collateral may have been made;
 - (b) Second, to the other amounts, except those specified in clause (c) below, which under the terms of this Lease have accrued;
 - (c) Third, to Owner Trustee to the extent of the aggregate Balance Due, plus any unpaid interest accruing because of the late payment of the Balance Due to the date of distribution; and
 - (d) Fourth, to the payment of the surplus, if any, to Lessee.

If there is a deficiency in any amounts due hereunder after Owner Trustee has exercised such remedies as it shall, in its sole discretion, elect to pursue, Lessee will, promptly upon Owner Trustee's demand, pay the same to Owner Trustee.

- 9.5. If Lessee fails to perform any of its agreements contained herein, whether or not an Event of Default has occurred and is continuing, Owner Trustee may perform such agreement, and the reasonable fees and expenses incurred by Owner Trustee in connection with such performance together with interest thereon shall be payable by Lessee upon demand. Interest on fees and expenses so incurred by Owner Trustee shall accrue as provided in Paragraph D of Schedule II hereto from the date such expense is incurred until paid in full.
- 9.6. Lessee unconditionally and irrevocably appoints Owner Trustee as its true and lawful attorney-in-fact, with full power of substitution, to the extent permitted by applicable law,

23

in its name and stead and on its behalf, for the purpose of effectuating any sale, assignment, transfer or delivery hereunder, if an Event of Default occurs and is continuing, whether pursuant to foreclosure or power of sale or otherwise, and in connection therewith to execute and deliver all such deeds, bills of sale, assignments, releases (including releases of this Lease on the records of any governmental body) and other proper instruments as Owner Trustee may reasonably consider necessary or appropriate. Lessee ratifies and confirms all that such attorney or any substitute shall lawfully do by virtue hereof. If requested by Owner Trustee or any purchaser, Lessee shall ratify and confirm any such lawful sale, assignment, transfer or delivery by executing and delivering to Owner Trustee or such purchaser, all deeds, bills of sale, assignments, releases and other proper instruments to effect such ratification and confirmation as may be designated in any such request.

Section 10. Return of Units.

10.1. If Owner Trustee shall rightfully demand possession of any Unit pursuant to this Lease or otherwise, Lessee, at its expense, shall forthwith (i) deliver possession of such Unit to Owner Trustee by delivering whatever data, information, records, reports, software, and other property is necessary to own and operate the Unit, all of which shall be in proper working order, to Owner Trustee, (ii) at the request of Owner Trustee, take all actions within Lessee's power to cause to be transferred to the Owner Trustee all Permits and other rights and privileges necessary to use the Units and (iii) take the actions

Section 11. Distributions to Owner Participants.

- All amounts of money received or realized by Owner Trustee pursuant to this Lease which are to be distributed to any Owner Participant (as distinguished from Lessee or any other person) shall be distributed as follows:
- 11.1. So long as the Balance Due has not been accelerated, Owner Trustee shall distribute each payment of rent and any Balloon Payment and any payment of interest pursuant to Section 15 on any overdue installment of rent or any other payment, received by it to each Owner Participant, on a pro rata basis, without preference or priority of any Owner Participant over another in accordance with such Owner Participant's then allocable share of the Balance Due, and in case moneys are insufficient to pay in full all amounts to be then distributed pursuant to this Section 11.1, then application shall be made first, to unpaid accrued interest pursuant to Section 15 on any overdue installment of rent or other payment, and second, to the payment of rental and any Balloon Payment.
- 11.2. Whether or not an Event of Default has occurred and is continuing, so long as the Balance Due has not been accelerated, Owner Trustee shall distribute all payments (including, without limitation, payments of warranty claims under the Purchase Agreement) it receives under or pursuant to this Lease in respect of a Casualty Occurrence (except to the extent otherwise required by Section 7.1) or in respect of early termination of this Lease, and any payment it receives directly or through Lessee from any insurer, governmental body or any other

24

person (to the extent that Lessee is not entitled thereto) with respect to a Casualty Occurrence, to each Owner Participant on a pro rata basis, without preference or priority of any Owner Participant over another in accordance with such Owner Participant's then allocable share of the Balance Due.

- 11.3. After the acceleration of the Balance Due and payable as a result of an Event of Default, Owner Trustee shall, after complying with paragraphs (a) and (b) of Section 9.4, distribute all moneys received and amounts realized by it after such date (including any insurance proceeds with respect to a Casualty Occurrence and amounts realized by Owner Trustee pursuant to the exercise of any of the remedies pursuant to this Lease) to the payment to Owner Participants, pro rata, without preference or priority of any Owner Participant over another in accordance with the Balance Due, plus any unpaid interest accruing because of the late payment of the Balance Due to the date of distribution; and in case moneys are insufficient to pay in full the whole amount due, owing or unpaid to Owner Participants pursuant to this Lease and the other Pertinent Documents then application shall be made first to unpaid accrued interest thereon, and second, to the Balance Due.
- 11.4. Owner Trustee shall distribute each payment not described in Section 11.1, 11.2, or 11.3, to each Owner Participant entitled to receive such payment pursuant to the terms of this Lease.
- 11.5. All distributions by Owner Trustee to Owner Participants pursuant to Section 11.1, 11.2 or 11.3 shall be made pro rata, without preference or priority of any Owner Participant over another, in accordance with the amounts due each Owner Participant at the time of such payment in respect of the types of obligations described in the Section pursuant to which such distribution is being made.
- 11.6. The amounts payable by Owner Trustee to Owner Participants pursuant to this Lease will be payable upon Owner Trustee receiving the same pursuant to this Lease at the Owner Trustee's Office, in immediately available funds, by transferring on the Business Day received (if received prior to 10:00 a.m., New York time; if received after such time, then on the next Business Day), by wire, in immediately available federal funds, to an account, designated to Owner Trustee in writing by such Owner Participant, maintained by such Owner Participant with any bank located in the United States maintaining wire transfer facilities, the amounts to be distributed to such Owner Participant.
- Section 12. Limitation of Liability; Compensation.
- 12.1. It is expressly understood and agreed by and among Lessee, each Owner Participant, and their respective successors and permitted assigns, that, subject to the proviso contained in this Section 12.1, all representations, warranties, covenants and undertakings of the Owner Trustee hereunder shall be binding upon the Owner Trustee only in its capacity as Owner Trustee under the Trust Agreement, and (except as otherwise expressly provided herein) the Owner Trustee shall not be liable in its individual capacity for any breach thereof, except in the

in handling its own funds, and, in all other cases for its gross negligence or willful misconduct; provided, however, that nothing in this Section 12.1 shall be construed to limit in scope or substance those representations, warranties, covenants and undertakings of the Owner Trustee made expressly in its individual capacity set forth herein or in the Trust Agreement. The term "Owner Trustee" as used in this Lease shall include any successor trustee under the Trust Agreement, or the Owner Participant if the trust created thereby is revoked.

12.2. Lessee shall pay Owner Trustee (or any successor thereto) its fees, costs and expenses for the performance of Owner Trustee's obligations hereunder and under the Trust Agreement in accordance with the terms of any agreement between Lessee and Owner Trustee (or any successor thereto) and, absent such agreement, Lessee shall pay Owner Trustee (or any successor thereto) its reasonable and customary fees, costs and expenses for the performance of Owner Trustee's obligations hereunder and under the Trust Agreement.

Section 13. Assignments.

- 13.1. All or any of the right, title or interest and obligations of any Owner Participant in and to this Lease and the rights, benefits, advantages and obligations of any Owner Participant hereunder, including the rights to receive payment of rental or any other payment hereunder, and its rights, titles and interests in and to the Trust Estate, may be assigned or transferred by such Owner Participant at any time; provided that (i) each assignment or transfer complies with all applicable securities laws, (ii) except in the case of an assignment or transfer to an Affiliate of such Owner Participant, each assignment or transfer is in a minimum amount equal to the lesser of \$5,000,000 or all of such Owner Participant's interest hereunder and under the Trust Agreement, (iii) any assignee or transferee first (A) acknowledges that the obligations to be performed from and after the date of such transfer or assignment are its obligations, including the obligations imposed by this Section 13.1 (and the transferor Owner Participant shall deliver to Lessee a written instrument executed by the assignee or transferee demonstrating that the transfer or assignment complies with this Section 13.1(iii)), and (B) further represents and warrants to Owner Trustee, each Owner Participant and Lessee in writing that:
 - (a) it will not be acquiring its interest with the assets of a separate account or an employee benefit plan (or its related trust) as defined in Section 3 of ERISA, or with the assets of a plan (or its related trust) as defined in Section 4975(e)(1) of the Internal Revenue Code of 1986 as amended (the "Code"), or with any assets which are "plan assets" within the meaning of Department of Labor regulation Section 2510.3-101, unless the acquisition qualifies for an exemption from the prohibited transaction rules under Section 406 of ERISA and Section 4975 of the Code, other than assets allocated to an insurance company pooled separate account as defined in ERISA Section 3(17) maintained by such Owner Participant which satisfies the requirements of U.S. Department of Labor Prohibited Transaction Class Exemption 90-1 with respect to the transactions contemplated by the Lease in order for such transactions to be exempt from the prohibitions of Section 406 of ERISA and Section 4975 of the Code;

26

- (b) except in the case of an assignee or transferee which is an Affiliate of the assigning or transferring Owner Participant, it is (i) a commercial bank, savings and loan association, savings bank, credit union, pension plan, depository institution, insurance company, finance company or other financially responsible entity, in each case, having a minimum consolidated net worth (on a consolidated basis for such institution and all of its consolidated affiliates) of \$50,000,000, (ii) a pension plan with investment assets in excess of \$50,000,000 or (iii) an investment manager controlling assets in excess of \$50,000,000;
- (c) it will not transfer its interests hereunder or under the Trust Agreement unless the proposed transferee makes the foregoing representations and covenants; and
- (d) it will not take any action with respect to this Lease, any other Pertinent Document or the transactions contemplated hereby and thereby that would violate any applicable securities laws.
- 13.2. Lessee shall not sell, assign, transfer or otherwise dispose of its rights or delegate its obligations under this Lease to any other person, except as permitted or required by Section 4.2, and any attempt to do so shall be void.
- Section 14. Ownership, Grant of Security Interest and Further Assurances.
- 14.1. Title to the Units shall remain in Owner Trustee as security for the obligations of Lessee hereunder and under the related documents to which it is a party until Lessee has fulfilled all of its obligations hereunder and under the related documents to which it is a party. Lessee hereby assigns,

hypothecates, transfers and pledges to Owner Trustee, and grants to Owner Trustee a security interest, in (i) the Units, including any substitute Units as set forth in Section 7.1(d), (ii) the Sublease, (iii) the Purchase Agreement, (iv) the Services Agreement and (v) all proceeds of the foregoing, in each case, whether now owned or hereafter acquired (all such property in which a security interest is granted by this Lease is referred to herein as "Collateral"), to secure the payment of all sums due hereunder and under the Pertinent Documents to which Lessee is a party and the performance of all other obligations hereunder and under the Pertinent Documents to which Lessee is a party and agrees, at its expense, to do any further reasonable act and execute, acknowledge, deliver, file, register and record any further documents which Owner Trustee or a Deciding Interest of Owner Participants may reasonably request in order to protect Owner Trustee's title to and perfected security interest in the Collateral, subject to no liens other than Permitted Liens, and Owner Participants' and Owner Trustee's rights and benefits under this Lease and the Collateral. Lessee shall promptly and duly execute and deliver to Owner Participants and the Owner Trustee such documents and assurances and take such further action as the Owner Trustee may from time to time reasonably request in order to carry out more effectively the intent and purpose of this Lease and the Pertinent Documents and to establish and protect the rights and remedies created or intended to be created in favor of the Owner Trustee hereunder and thereunder, to establish, perfect and

27

maintain the right, title and interest of the Owner Trustee in and to the Units, subject to no Lien other than Permitted Liens, at the expense of the Lessee, the recording or filing of counterparts or appropriate memoranda hereof, or of such financing statements or other documents with respect hereto as Owner Participants or the Owner Trustee may from time to time reasonably request, and the Lessee agrees to execute and deliver promptly such of the foregoing financing statements or other documents as may require execution by the Lessee. To the extent permitted by applicable laws, the Lessee hereby authorizes any such financing statements to be filed without the necessity of the signature of the Lessee.

- 14.2. If Lessee would be entitled to any amount or title to any Unit hereunder but for the existence of any Event of Default or Default, Owner Trustee shall hold such amount or Unit as part of the Collateral and shall be entitled to apply such amounts against any amounts due hereunder; provided that Owner Trustee shall distribute such amount or transfer such Unit in accordance with the other terms of this Lease if and when no Event of Default or Default exists.
- 14.3. Upon the final payment and performance in full of all obligations of Lessee hereunder, the security interest granted herein shall terminate and all rights to the Collateral shall revert to Lessee. Upon any such termination, Owner Trustee will, at Lessee's sole expense, execute and deliver to Lessee such documents as Lessee shall reasonably request to evidence such termination.

Section 15. Late Payments, Etc.

Lessee shall pay to Owner Trustee, on demand, interest at the rate set forth in Paragraph D of Schedule II hereto on the amount of any payment not made when due hereunder from the date due (not taking into account any grace period) until payment is made.

Section 16. Effect of Waiver.

No delay or omission to exercise any right, power or remedy accruing to Owner Trustee or any Owner Participant upon any breach or default of Lessee hereunder shall impair any such right, power or remedy nor shall it be construed to be a waiver of any such breach or default, or an acquiescence therein or of or in any similar breach or default thereafter occurring, nor shall any single or partial exercise of any right, power or remedy preclude other or further exercise thereof, or the exercise of any other right, power or remedy, nor shall any waiver of any single breach or default be deemed a waiver of any other breach or default theretofore or thereafter occurring. Any waiver, permit, consent or approval of any kind or character on the part of Owner Participants or Owner Trustee of any breach or default under this Lease must be specifically set forth in writing and must satisfy the requirements set forth in Section 19 with respect to approval by Owner Participants and Owner Trustee.

28

Section 17. Survival of Covenants.

All claims pertaining to the representations, warranties and covenants of Lessee under this Lease shall survive the expiration or termination of this Lease.

THIS LEASE SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICT OF LAW PRINCIPLES.

Section 19. Effect and Modification of Lease.

This Lease exclusively and completely states the rights of Owner Trustee, Owner Participants and Lessee with respect to the leasing of the Units and supersedes all prior agreements, oral or written, with respect thereto. No variation, modification, amendment or waiver of this Lease shall be valid unless in writing and signed by Owner Trustee, with the consent of a Deciding Interest of Owner Participants, and by Lessee. No variation, modification, amendment or waiver of this Lease purporting to (i) postpone, reduce or forgive, in whole or in part, any rental payment, Balloon Payment, Balance Due, interest or other amount payable hereunder, or modify the definition, or method of calculation, of any rental payment, Balloon Payment, Balance Due, interest or other amount payable hereunder, (ii) release any Collateral granted hereunder (except as expressly provided in this Lease) or modify or delete this sentence or the definition of "Deciding Interest of Owner Participants", (iii) waive any condition precedent to the obligation of any Owner Participant without the Owner Participant's written consent, or (iv) modify or delete Schedule II or any annex thereto, shall be valid unless in writing and signed by Owner Trustee with the consent of all Owner Participants.

Section 20. Reports; Confidentiality.

20.1. Lessee shall deliver, or shall cause to be delivered to, Owner Participants and Owner Trustee forthwith upon any Responsible Officer of Lessee obtaining knowledge of the occurrence of an Event of Default or an Default, or any default under the Services Agreement or the Purchase Agreement, an Officer's Certificate signed by such Responsible Officer specifying the nature and period of existence thereof and stating what action Lessee has taken or is taking or proposes to take with respect thereto.

"Officer's Certificate" shall mean, with respect to any party, a certificate duly executed on behalf of such party by its President or one of its Vice Presidents, its Treasurer or its Controller.

20.2. Owner Trustee and each Owner Participant may at its own expense upon reasonable prior notice inspect, during the normal business hours of Lessee, Lessee's books and records relating to the maintenance and performance of the Units and make copies and extracts

29

therefrom, and may discuss such matters with its officers. Lessee shall provide, with reasonable promptness, such other information and data with respect to Sublessee or Lessee, or the Units, as from time to time may be reasonably requested by any Owner Participant or Owner Trustee.

20.3. Owner Trustee and each Owner Participant shall maintain in confidence and not publish, disseminate or disclose in any manner to any person and shall not use (x) any material, nonpublic information relating to Lessee or (y) any technical, nonfinancial information, data or know-how which is identified in writing as confidential by Lessee, in either case, which may be furnished pursuant to this Lease (hereinafter collectively called "Confidential Information") subject to Owner Trustee's and each Owner Participant's (a) obligation to disclose any such Confidential Information pursuant to a request or order under applicable laws and regulations or pursuant to a subpoena or other legal process, (b) right to disclose any such technical or financial Confidential Information to bank examiners, the National Association of Insurance Commissioners, insurance commissioners, Lessee's affiliates, auditors, counsel, other professional advisors, other Owner Participants, and other banks or other entities in connection with an offer by any Owner Participant to make an assignment pursuant to Section 14 hereof, (c) right to use and disclose any such Confidential Information in connection with the transactions set forth herein, and (d) right to disclose any such Confidential Information in connection with the transactions set forth herein or in connection with any litigation or dispute involving Owner Trustee or any Owner Participant and the Lessee or Sublessee; provided, however, that Confidential Information disclosed pursuant to clause (b) or (d) of this sentence shall be so disclosed subject to such procedures as are reasonably calculated to maintain the confidentiality thereof; provided, further, that Confidential Information disclosed pursuant to applicable laws, regulations, subpoenas or other legal process shall be so disclosed subject to such confidentiality provisions, if any, as may be provided under applicable law. Owner Trustee and Owner Participants agree, to the extent permitted by applicable law, to use their reasonable efforts promptly to notify Lessee in writing of each order, subpoena or other legal process providing for the disclosure and/or production of Confidential Information and shall, to the extent permitted by applicable law, use their reasonable efforts promptly to supply Lessee with a copy of such order, subpoena or other legal process, in order that Lessee may intervene in the relevant administrative or legal proceeding or take other appropriate legal action to protect the confidentiality of such Confidential Information. Notwithstanding the foregoing, (i) the foregoing obligation of confidentiality shall not apply to any such Confidential

Information that was known to Owner Trustee or any Owner Participant or any of its affiliates prior to the time it received such Confidential Information from Lessee or Sublessee pursuant to this Lease, other than as a result of the disclosure thereof by a person who, to the knowledge or reasonable belief of Owner Trustee or such Owner Participant, was prohibited from disclosing it by any duty of confidentiality arising (under this Lease or otherwise) by contract or law, and (ii) the foregoing obligation of confidentiality shall not apply to any such Confidential Information that becomes part of the public domain independently of any act of Owner Trustee or such Owner Participant not permitted hereunder (through publication, the issuance of a patent disclosing such information or otherwise) or when identical or substantially similar information is received by Owner Trustee or such Owner Participant without restriction as to its disclosure or use, from a person who, to the knowledge or reasonable belief of Owner Trustee or such Owner Participant,

3.0

was not prohibited from disclosing it by any duty of confidentiality arising (under this Lease or otherwise) by contract or law.

20.4. Owner Trustee and Owner Participants shall have the right, by their agents, to discuss the affairs of Lessee and the status of any Unit with officers of Lessee and, if there shall be an Default or Event of Default, with representatives of its certified public accountants, all at such reasonable times as a Deciding Interest of Owner Participants may request during the term of this Lease. Lessee shall, if a Default or Event of Default has occurred, take such reasonable actions as any Owner Participant may request to arrange for such Owner Participant or Owner Trustee to have discussions with any employee of Lessee and/or any third party that provides services and support with respect to the Units.

Section 21. Notices.

All demands, notices and other communications hereunder shall be in writing and shall be deemed to have been duly given when personally delivered or telecopied (and confirmed telephonically with the recipient or confirmed in writing by overnight courier) or one Business Day after being sent by overnight delivery service, addressed to: (A) Lessee or Owner Trustee at the address set forth on Schedule 21, or at such other address as may hereafter be furnished in accordance with this Section 21 by any party to all other parties, and (B) each Owner Participant at its address set forth in Schedule I hereto, or at such other address as may hereafter be furnished in accordance with this Section 21 by any party to all other parties.

Section 22. Counterparts.

This Lease has been executed in several counterparts. One counterpart has been prominently marked "Owner Trustee's Copy". Only the counterpart marked "Owner Trustee's Copy" shall evidence a monetary obligation of Lessee or shall be deemed to be an original or to be chattel paper for purposes of the Uniform Commercial Code, and such copy shall be held by Owner Trustee.

Section 23. Severability.

Whenever possible, each provision of this Lease shall be interpreted in such manner as to be effective and valid under applicable law; but if any provision of this Lease shall be prohibited by or invalid under applicable law, such provision shall be ineffective to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of this Lease.

Section 24. Successors and Assigns; Merger.

24.1. This Lease shall be binding upon the parties hereto and their respective successors and assigns and shall inure to the benefit of the parties hereto and their respective successors and permitted assigns.

31

24.2. Lessee shall not enter into any transaction of merger or consolidation with any other entity unless (a) either (i) Lessee shall be the surviving entity or (ii) if Lessee is not the surviving entity, then the surviving entity shall be a corporation organized under the laws of the United States of America, a state thereof or the District of Columbia, and shall expressly assume in a writing the obligations of Lessee hereunder. (b) immediately after such merger or consolidation, no Event of Default or Default shall have occurred and be continuing hereunder, (c) immediately after such merger or consolidation, Owner Trustee shall have an enforceable security interest of record in all Collateral subject to no liens other than Permitted Liens, and (d) Lessee shall have delivered to Owner Participants and Owner Trustee in form and substance satisfactory to them (i) an Officer's Certificate of Lessee confirming that all requirements of this Section 24.2 have been met and that no Event of Default or Default shall have occurred and be continuing, and (ii) an opinion of counsel to Lessee stating that (among other things that may be reasonably requested by any Owner Participant or Owner Trustee) such

merger or consolidation has been completed in accordance with the requirements of law of each relevant jurisdiction and that, if Lessee is not the surviving entity, the surviving entity has expressly assumed all of the obligations of Lessee hereunder. If all of the foregoing requirements have been met, the surviving entity shall have all the rights of Lessee hereunder.

Section 25. Brokers.

None of the parties has engaged or authorized any broker, finder, investment banker or other third party to act on its behalf, directly or indirectly, as a broker, finder, investment banker, agent or any other like capacity in connection with this Lease or the transactions contemplated hereby, except that Lessee has engaged GECC and/or an affiliate thereof, the fees of which will be paid by Lessee.

Section 26. Jury Trial.

EACH OF LESSEE, EACH OWNER PARTICIPANT AND OWNER TRUSTEE WAIVES ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHTS UNDER THIS LEASE OR ANY RELATED DOCUMENT OR UNDER ANY AMENDMENT, INSTRUMENT, DOCUMENT OR AGREEMENT DELIVERED OR WHICH MAY IN THE FUTURE BE DELIVERED IN CONNECTION HEREWITH OR THEREWITH OR ARISING FROM ANY RELATIONSHIP EXISTING IN CONNECTION WITH THIS LEASE OR ANY RELATED DOCUMENT AND AGREES THAT ANY SUCH ACTION OR PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

Section 27. Captions; Table of Contents.

Section captions and the table of contents used in this Lease (including Schedule II hereto) are for convenience of reference only and shall not affect the construction of this Lease.

32

Section 28. Schedules and Exhibits.

Schedule I and Schedule II and each of the exhibits hereto, along with all attachments referenced in any of such items are incorporated herein by reference and made a part thereof.

Section 29. Certain Definitions.

"Acceptance Certificate" is defined in Section 1.2(o).

"Adjusted LIBOR Rate" is defined in Annex II to Schedule II hereto.

"Affiliate" of any Person shall mean any other Person directly or indirectly controlling, controlled by or under common control with, such Person.

"Aggregate Purchase Price" is defined in Paragraph B of Schedule II hereto.

"Amortization Amount" is defined in Annex II to Schedule II hereto.

"Appraisal" is defined in Section 1.2(b).

"Appraiser" means GE American Communications, Inc.

"Appraised Value" is defined in Section 1.2(b).

"Balance Due" is defined in Section 7.1(c).

"Balloon Payment" is defined in Paragraph E of Schedule II hereto.

"Base Lessee Residual Guarantee" is defined in Paragraph E of Schedule II hereto.

"Base Rate" shall mean the prime rate as published in The Wall Street Journal from time to time.

"Base Term" is defined in Section 2.1.

"Bill of Sale" is defined in Section 1.2(a).

"Business Day" is defined in Paragraph C of Schedule II hereto.

"Casualty Amount" is defined in Section 7.1(c).

"Casualty Notice" shall mean written notice delivered by Lessee to Owner Trustee in accordance with Section 7.1(a) which specifies the Unit or Units suffering a Casualty Occurrence, the nature and scope of the Casualty Occurrence, the rental payment date or other

date on which the Casualty Amount with respect to such Casualty Occurrence is due and payable in accordance with Section 7.1(a) and the Casualty Amount, setting forth in reasonable detail the calculation thereof.

- "Casualty Occurrence" is defined in Section 7.1(c).
- "Closing Notice" is defined in Section 1.2(p).
- "Code" is defined in Section 13.1(a).
- "Collateral" is defined in Section 14.1.
- "Confidential Information" is defined in Section 20.3.

"Contingent Obligation": as to any person, any obligation of such person guaranteeing or in effect guaranteeing any Indebtedness, leases, dividends or other obligations ("primary obligations") of any other person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of such person, whether or not contingent (a) to purchase any such primary obligation or any property constituting direct or indirect security therefor, (b) to advance or supply funds (i) for the purchase or payment of any such primary obligation or (ii) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (c) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation or (d) otherwise to assure or hold harmless the owner of any such primary obligation against loss in respect thereof; provided, however, that the term Contingent Obligation shall not include (w) endorsements of instruments for deposit or collection in the ordinary course of business, (x) any obligation arising under any agreement between members of a group consolidated or combined for purposes of computing income tax liability requiring a member to make capital contributions or cash advances based upon assumed tax benefits, (y) any obligation of such person or its subsidiaries for primary obligations of any other person otherwise reflected in the consolidated balance sheet of such person and its subsidiaries, and (z) any obligation with respect to a lease accounted for in accordance with GAAP as an operating lease.

"Credit Agreement" shall mean that certain Amended and Restated Credit Agreement, dated as of November 14, 1997, by and among GCI Holdings, Inc., the lenders identified therein, Bank of America, N.A. (as successor to Nationsbank of Texas, N.A.), as administrative agent, Credit Lyonnais New York Branch, as documentation agent, and TD Securities (USA) Inc., as syndication agent, as heretofore or hereafter amended, supplemented or otherwise modified, and any credit agreement, loan agreement or indenture entered into by one or more GCI Entities in replacement thereof.

34

"Deciding Interest of Owner Participants" shall mean Owner Participants holding an interest in more than 50% of the Balance Due at the time of giving the relevant instruction or taking action.

"Default" shall mean any condition, event or act which, with notice or lapse of time or both, would become an Event of Default.

"Delivery Date" is defined in Section 1.1.

"Disclosure Document" is defined in Section 1.2(f).

"Environmental Laws" means and includes the Resource Conservation and Recovery Act of 1976, (RCRA) 42 U.S.C. ss.ss. 6901-6987, as amended by the Hazardous and Solid Waste Amendments of 1984, the Comprehensive Environmental Response, Compensation and Liability Act, as amended by the Superfund Amendments and Reauthorization Act of 1986, 42 U.S.C. ss.ss. 9601-9657, (CERCLA), the Hazardous Materials Transportation Act of 1975, 49 U.S.C. ss.ss. 1801-1812, the Toxic Substances Control Act, 15 U.S.C. ss.ss. 2601-2671, the Clean Air Act, 42 U.S.C. ss.ss. 7401 et seq., the Federal Insecticide, Fungicide and Rodenticide Act, 7 U.S.C. ss.ss. 136 et seq., and all similar federal, state and local environmental laws, ordinances, rules, orders, statutes, decrees, judgments, injunctions, codes and regulations, and any other federal, state or local laws, ordinances, rules, codes and regulations, and any other federal, state or local laws, ordinances, rules, codes and regulations relating to the environment, human health or natural resources or the regulation or control of or imposing liability or standards of conduct concerning human health, the environment, Hazardous Materials or the clean-up or other remediation thereof, or any part thereof, as any of the foregoing may have been from time to time amended, supplemented or supplanted.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time and in effect.

"Eurocurrency Reserve Rate" means, for any day with respect to a

Eurodollar Rate Advance, the maximum rate (expressed as a decimal) at which any lender subject thereto would be required to maintain reserves under Regulation D of the Board of Governors of the Federal Reserve System (or any successor or similar regulations relating to such reserve requirements) against "Eurocurrency Liabilities" (as that term is used in Regulation D), if such liabilities were outstanding. The Eurocurrency Reserve Rate shall be adjusted automatically on and as of the effective date of any change in the Eurocurrency Reserve Rate.

"Event of Default" is defined in Section 9.1.

"Execution Date" is defined in the preamble.

"Fair Market Value" means, with respect to any Unit, the fair market sales value as determined by an independent appraiser chosen by Owner Trustee and, so long as no Event of

3.5

Default has occurred and is continuing, approved by the Lessee (such approval not to be unreasonably withheld) that would be obtained in an arm's-length transaction between an informed and willing buyer (other than a lessee currently in possession) and an informed and willing seller, under no compulsion, respectively, to buy or sell and neither of which is related to the Lessee, for the purchase of the Units. Such fair market sales value shall be calculated as the value for the use of the Units, assuming, in the determination of such fair market sales value, that the Units are in the condition and repair required to be maintained by the terms of this Lease.

"GCI" means General Communication, Inc., an Alaska corporation.

"GCI Communication" is defined in the Recitals.

"GCI Entity" shall mean Lessee, Sublessee, GCI and each Restricted Subsidiary.

"GECC" is defined in the preamble.

"Hazardous Material" means any hazardous waste, as defined by 42 U.S.C. ss. 6903(5), any hazardous substances as defined by 42 U.S.C. ss. 6901(14), any pollutant or contaminant as defined by 42 U.S.C. ss. 9601(33) and any other toxic substances, oil or hazardous materials or other chemicals or substances regulated by any Environmental Laws.

"Impositions" is defined in Section 5.1.

"Indebtedness" shall mean, with respect to a person, at a particular date, the sum (without duplication) at such date of (a) indebtedness for borrowed money or for the deferred purchase price of property or services (except trade accounts payable arising in the ordinary course of business) in respect of which such person is liable, as obligor, (b) obligations of such person under capitalized leases and (c) any reimbursement obligations of such person in respect of letters of credit, acceptances, or similar obligations issued or created for the account of such person.

"Indenture" mean the Indenture dated as of August 1, 1997, between GCI, Inc, and The Bank of New York, as trustee, providing for the $$180,000,000\ 9-3/4\%$ Senior Notes due 2007, as amended, supplemented or otherwise modified from time to time.

"Interest Period" is defined in Annex II to Schedule II hereto.

"Inspector" is defined in Annex III to Schedule II hereto.

"Lease" is defined in the preamble.

"Lessee" is defined in the preamble.

36

"LIBOR Rate" is defined in Annex II to Schedule II hereto.

"Material Adverse Effect" shall mean a material adverse effect upon (a) the financial condition, operations, business, assets or prospects of (i) GCI and its consolidated subsidiaries, taken as a whole, (ii) Lessee or (iii) Sublessee; (b) the ability of either Lessee or Sublessee to perform its obligations under the Pertinent Documents, the Purchase Agreement or the Services Agreement, (c) the value, utility or useful life of any Unit, (d) the validity or enforceability of any Pertinent Document or (e) the status, priority or perfection of Owner Trustee's interest in the Collateral, or any rights, claims or remedies of the Owner Trustee or any Owner Participant.

"Officer's Certificate" is defined in Section 20.1.

"Owner Participant" shall mean GECC and its successors and assigns as holders of a beneficial interest in the ${\it Trust}$ Estate.

- "Owner Trustee" is defined in the preamble.
- "Owner Trustee's Office" is defined in Section 2.3.
- "PanAmSat" is defined in the Recitals.
- "Payment Date" is defined in Annex II to Schedule II hereto.
- "PBGC" shall mean the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA.
 - "Permits" is defined in Section 1.4(d).

"Permitted Liens" means (i) any rights of the parties pursuant to this Lease and any rights of any persons entitled to use of the Units in accordance with Section 4.2; (ii) any lien arising out of any act of, or any failure to act by, or any claim (including any claim for taxes) against, any Owner Participant or Owner Trustee which is unrelated to the transactions contemplated hereby; and (iii) any lien for current taxes, assessments or other governmental charges which are not delinquent or the validity of which is being contested in good faith by appropriate proceedings promptly initiated and diligently prosecuted so long as such proceedings do not involve any reasonable danger of sale, forfeiture or loss of all or any part of any Unit and do not materially adversely affect the lien created by this Lease;

"Pertinent Documents" shall mean the Bill of Sale, the Acceptance Certificate, this Lease, the Sublease, the Trust Agreement and each document executed and delivered in connection with this Lease or the Sublease (as any of the foregoing may be amended, modified, restated or supplemented from time to time).

"Purchase Agreement" is defined in the Recitals.

37

- "Purchase Price" is defined in Paragraph B of Schedule II hereto.
- "Redelivery Period" is defined in Paragraph E of Schedule II hereto.
- "Regulatory Change" means, relative to any Owner Participant
- (a) any change in (or the adoption, implementation, change in phase-in or commencement of effectiveness of) any
 - (i) United States federal or state law or foreign law applicable to such Owner Participant;
 - (ii) regulation, interpretation, directive, requirement or request (whether or not having the force of law) applicable to such Owner Participant of (A) any court, government authority charged with the interpretation or administration of any law referred to in clause (a)(i) or of (B) any fiscal, monetary or other authority having jurisdiction over such Owner Participant; or
 - (iii) generally accepted accounting principles or regulatory accounting principles applicable to such Owner Participant and affecting the application to such Owner Participant of any law, regulation, interpretation, directive, requirement or request referred to in clause (a)(i) or (a)(ii) above; or
- (b) any change in the application to such Owner Participant of any existing law, regulation, interpretation, directive, requirement, request or accounting principles referred to in clause (a)(i), (a)(ii) or (a)(iii) above.
- "Renewal Term" is defined in Section 2.1.
- "Requisition of Use" is defined in Section 7.1(c).
- "Responsible Officer" is defined in Paragraph E of Schedule II hereto.
- "Sales Period" is defined in Paragraph E of Schedule II hereto.
- "Satellite" is defined in Section 1.4(c).
- "Satellite Failure" is defined in Section 7.1(c).
- "Services Agreement" is defined in Section 1.2(1).
- "State Street" is defined in the preamble.
- "Sublease" is defined in the Recitals.

38

"Sublessee" is defined in the Recitals.

"Supplemental Lessee Residual Guaranty" is defined in Paragraph ${\tt E}$ of Schedule II hereto.

"Term" is defined in Section 2.1.

"Trust Agreement" means the Trust Agreement (GCI Transponder 2000-1), dated as of March 31, 2000, between State Street and GECC, as it may be amended, supplemented or otherwise modified from time to time.

"Trust Estate" means the Units, the lease, the other Collateral and any and all other property held by Owner Trustee for the benefit of the Owner Participants pursuant to the Trust Agreement.

"U.C.C." shall mean the Uniform Commercial Code, as in effect from time to time in the State of Alaska.

"Unit" is defined in the Recitals.

"Unit Value Fraction" is defined in Section 7.1(c).

Unless otherwise specified herein, all accounting terms used herein shall be interpreted, and all accounting determinations and computations shall be made, in accordance with generally accepted accounting principles consistently applied.

Any capitalized terms used herein that are not otherwise defined herein shall have the meanings assigned thereto in the Sublease.

39

IN WITNESS WHEREOF, the parties hereto have executed this Lease as of the day and year first above written.

GCI SATELLITE CO., INC., Lessee

By: /s/

Name: John M. Lowber

Title: Senior Vice President and Chief Financial Officer

STATESTREET BANK AND TRUST COMPANY
OF CONNECTICUT, NATIONAL
ASSOCIATION, not in its individual
capacity except as expressly set
forth herein, but solely as Owner
Trustee

By: /s/

Name: Rinette Elovecky Title: Vice President

GENERAL ELECTRIC CAPITAL CORPORATION, Owner Participant

By: /s/

Name: Molly S. Fergusson Title: Manager, Operations

LEASE INTENDED S-1 FOR SECURITY

<TABLE>

SCHEDULE I TO LEASE INTENDED FOR SECURITY

OWNER PARTICIPANTS

<CAPTION>

Name: Address: Wire Instructions: Commitment
Amount:

<S>
General Electric Capital
Corporation

120 Long Ridge Road Stamford, Connecticut 06927 CC>
Bankers Trust Company - New York
ABA #021001033
Account# 50-205-776

\$53,631,000

GECC T&I Depositary Contact: Doris Adams 212/250-8383

Reference: GCI Satellite Co., Inc.

</TABLE>

SCHEDULE II TO LEASE INTENDED FOR SECURITY

A. Description of Units.

As set forth on Annex I hereto (including the Purchase Price for each thereof), which by this reference is made a part hereof and of the Lease referred to above.

B. Aggregate Purchase Price.

"Purchase Price" for a Unit shall mean the purchase price of the Unit set forth in Annex I hereto (which shall include an allocated portion of legal expense and warranty cost) and the "Aggregate Purchase Price" shall mean the sum of the Purchase Price for all Units.

C. Rental.

During the term, Lessee shall pay rental for each Unit, on the dates, in the amounts and otherwise as set forth in Annex II hereto, which by this reference is made a part hereof and of the Lease referred to above. All such payments, as well as all other payments due to Owner Trustee under the Lease, shall be received by Owner Trustee not later than 1:00 p.m., New York time, on the date due; if such payments are received by Owner Trustee not later than 10:00 a.m., New York time, Owner Trustee shall disburse such funds (including for purposes of Sections 2 and 11.6 of the Lease) to Owner Participants by the end of the same Business Day (if such payments are received after such time, Owner Trustee shall deliver such funds to Owner Participants by 11:00 a.m., New York time, on the next Business Day). Funds received by any Owner Participant after 2:30 p.m., New York time, shall be deemed to be received by such Owner Participant for purposes of Paragraph D on the next succeeding Business Day. The foregoing provisions are subject to the last sentence of Section 2.3. As used herein, "Business Day" shall mean a day, other than a Saturday or a Sunday on which Owner Trustee and commercial banks located in Connecticut and New York are open for the purpose of conducting commercial banking business and, if related to LIBOR, banks were open in London, England.

D. Interest on Late Payments.

The interest rate on late payments shall be (i) the Adjusted LIBOR Rate then applicable pursuant to Annex II to Schedule II, plus 5.00% per annum, computed daily on the basis of a 360 day year for actual days elapsed, or (ii) the maximum amount permitted by law, whichever is less.

E. Purchase/Remarketing Provision.

(i) Lessee's Option. At least six months before the end of the lease term (the "Sales Period"), a Responsible Officer of Lessee shall notify Owner Trustee of its election of (1) either exercising option (a) or renewing this Lease pursuant to Section 2.1 (it being understood that

Lessee shall notify Owner Trustee at least three months before the end of the lease term whether it will purchase or renew, which notice shall be irrevocable) or (2) exercising option (b) below, and at the end of the term of the Lease, provided that the Lease has not been earlier terminated, Lessee shall, in accordance with its election:

- (a) Purchase for cash all Units then subject to the Lease for an amount equal to the Balance Due on the last day of the Term (which sum is referred to herein as the "Balloon Payment"), plus all other amounts then due pursuant to this Lease (unless Lessee has renewed this Lease pursuant to Section 2.1); or
- (b) Sell the Units then subject to the Lease to a purchaser not affiliated with Lessee and pay to Owner Trustee on the day the Lease terminates for the Units (i) the proceeds of sale of each Unit without deductions (which sale shall occur on the date the Lease terminates and shall be for cash) and (ii) if the proceeds are less than the Balloon Payment for such Units or if not all Units are sold and the proceeds of those Units which are sold are less than the Balloon Payment, Lessee shall pay Owner Trustee (from funds other than the sales proceeds) the difference between (x) the amount of the Balloon Payment and (y) the amount of the sale proceeds, if any; provided, however, Lessee shall not be obligated to pay to Owner Trustee with respect to the amount calculated in this clause (ii) an amount which exceeds the sum of (i) the applicable amount shown on Schedule A hereto (the "Base Lessee Residual Guarantee"), plus (ii) if applicable, the amount shown on Schedule B hereto (the "Supplemental Lessee Residual Guarantee"). If the proceeds received for all such

Units exceed the amount of the Balloon Payment, Owner Trustee shall remit to Lessee the excess. The Supplemental Lessee Residual Guarantee shall be zero if (i) Lessee has exercised a Renewal Term and, in connection therewith has requested a re-appraisal of the Units and (ii) such re-appraisal is from the Appraiser or other person acceptable to the Owner Participants, is in form and substance satisfactory to Owner Participant and shows that the Fair Market Value of the Units as of the last day of such Renewal Term is at least equal to 101% of the value for the Units set forth for the last day of such Renewal Term in the Appraisal delivered on the Delivery Date.

Lessee's election shall be irrevocable at the time made, but if Lessee fails to make a timely election (including a timely election as to whether Lessee will purchase or renew), or if an Event of Default or a Default has occurred and is continuing on the last day of the term of the Lease, or if a Requisition for Use has occurred and is continuing with respect to one or more Units, Lessee shall be deemed to have elected option (a) (irrespective of whether theretofore Lessee shall have elected option (b)).

For the purposes hereof, "Responsible Officer" with respect to a party shall mean its President or any of its Vice Presidents.

(ii) Option (b) Procedures. If Lessee elects option (b), Lessee shall use its best efforts throughout the Sales Period to obtain a purchaser or purchasers for the Units which are not

Schedule II-2

affiliated in any way with Lessee. Owner Trustee may direct Lessee to hire and pay for sales agents as directed by Owner Participant. Except as otherwise provided below, any sale by Lessee shall be for the highest cash bid submitted to Lessee, including any bid submitted by Owner Trustee or any Owner Participant. The determination of the highest bid shall be made by Owner Trustee with the consent of the Owner Participant at the end of the Sales Period, except as otherwise provided below. Owner Trustee may accept any bid solicited by Lessee or its agent, in which case Lessee agrees to maintain the Units in accordance with the condition herein provided. Neither Owner Trustee nor any Owner Participant shall have any responsibility for procuring any purchaser. If, nevertheless, Owner Trustee or any Owner Participant undertakes any sales efforts, Lessee shall promptly reimburse Owner Trustee or such Owner Participant for any charges, costs and expenses incurred in such effort, including any allocated time charges, costs and expenses of internal counsel or other attorneys' fees.

To facilitate the sale of the Units and to assure the best possible sales price for the Units, Lessee, at its expense, shall do, at least, all of the following (which, shall be in addition to, and not in limitation of, Lessee's obligations under the Lease, including, without limitation, its obligations to insure the Units and make rental payments).

By the last day of the first calendar month after the commencement of the Sales Period (the "Redelivery Period"), Lessee at its expense, (i) shall have removed all of the Units from service in the Lessee's business, (ii) shall have tested all of the Units to ensure that they comply with the requirements of Annex III, and (iii) shall have certified to each Owner Participant and Owner Trustee that the Units at that time comply with such requirements.

Lessee shall maintain the Units in the condition required by Section 4 during the Sales Period and thereafter until such Units are returned pursuant hereto. Lessee shall allow and pay for unlimited inspection of telemetry logs and other records and testing by potential purchasers, Owner Trustee and any Owner Participant, and shall pay the travel costs for all inspectors and testers, as well as any fees charged by governmental or industry testing agencies and testing companies. Lessee shall promptly provide any information, records or computer printouts (other than U.S. Government classified information) requested by Owner Trustee, any Owner Participant or potential purchasers, including records of all maintenance of or repair to each Unit prior to or during the term of the Lease and original manufacturers' operating specifications.

Lessee shall be responsible for all costs of sale and in order to facilitate a sale shall provide, or cause to be provided, at Lessee's expense, such equipment and ancillary facility configurations, repairs, corrections and modifications as shall be necessary or desirable to Owner Trustee, any Owner Participant or any purchaser. If requested, Lessee shall provide financing to one or more qualified creditworthy purchasers at a rate not higher than the purchasers' standard commercial borrowing rates and on standard terms and conditions to enable the purchasers to purchase the Units and pay cash to the Owner Trustee. On behalf of purchasers, Lessee, at its expense, shall continue to maintain and insure Units until the purchaser's revenue start date, as indicated by the purchaser. Lessee shall inspect and certify that each Unit at the date the Lease terminates meets all such requirements and all requirements of the Lease. During the last year of

Schedule II-3

the term of the Lease, Lessee shall not acquire, by purchase, lease or otherwise, and shall prevent any person or entity directly or indirectly controlling, controlled by or in common control with Lessee from acquiring, by purchase, lease or otherwise, more units (or the use of more units) of a type or types similar to the Units unless and only to the extent that Lessee is properly compelled by a court of competent jurisdiction to do so (and Lessee acknowledges and agrees that the Owner Participants may be unwilling to make the Units available to Lessee for such purposes). Lessee shall warrant to Owner Trustee, Owner Participants and any purchaser absolutely and without condition that the Units will be in the condition described in Annex III to this Schedule upon their sale to such purchaser, and that the Units will operate in accordance with their original manufacturers' specifications for at least a period equal to 12 years minus the Term.

In addition, Lessee shall warrant to Owner Participants and any purchaser that the fuel on the satellite in which the Units are contained is sufficient to provide for a geosynchronous orbit of such satellite at its orbital position of 123(0) West Longitude (or such other orbital slot as the Federal Communications Commission has approved or required) for at least a period equal to 12 years minus the Term. Lessee shall provide access to the Units for purchasers and Owner Participants, including codes and encryption devices, and take all actions within Lessee's power to cause to be transferred to such purchaser all permits, licenses or other rights or privileges necessary to operate the Units, including but not limited to any Federal Communications Commission position, operating or frequency licenses.

If (x) option (b) is in effect at the end of the Term but Lessee has been unable to sell a Unit to an unaffiliated person during the Sales Period and (y) the total amount paid by Lessee to Owner Trustee (including any proceeds of sales of Units) pursuant to option (b) at the end of the Lease term is less than the amount specified in option (a), at the option of the Owner Trustee, the absolute title to each unsold Unit at the end of the Term shall vest in Owner Trustee, free and clear of any rights of redemption or other rights of Lessee, and Lessee shall, at Lessee's expense, if requested by Owner Trustee or Owner Participants, do any further acts and execute, acknowledge, deliver, file, register and record any further documents which Owner Trustee or any Owner Participant may reasonably request in order to establish and confirm Owner Trustee's or such Owner Participant's title and rights.

(iii) Procedures After Payment Made Under Option (a) or (b). Upon (x) payment of the Balloon Payment (pursuant to option (a)) or sales price (pursuant to option (b)), as the case may be, in United States dollars, and (y) the payment by Lessee in full of all other obligations of Lessee due and payable under the Lease, Owner Trustee shall execute and deliver to Lessee (in the case of option (a)) or to the purchaser, which may be an Owner Participant (in the case of option (b)) a quitclaim bill of sale (without representations or warranties except that each Unit is free and clear of all claims, liens, security interests and other encumbrances by or in favor of Owner Trustees, Owner Participants or any person claiming by, through or under Owner Trustee or Owner Participants) for the Units (other than any unsold Units which are subject to the preceding paragraph), and such other documents as may be required to release such Units from the terms and scope of the Lease, in such form as may reasonably be requested by Lessee, all at Lessee's expense.

Schedule II-4

<TABLE>

SCHEDULE A TO SCHEDULE II TO LEASE INTENDED FOR SECURITY

Base Lessee Residual Guarantee

Lease Balance, Lessor Risk and Lessee Risk (in \$ of original cost)

<CAPTION>

	Lease Balance/		
End of Year	Purchase Option	Maximum Base Lessee	Maximum Lessor
(February 28th)	Amount	Residual Guarantee	Residual Risk
<s></s>	<c></c>	<c></c>	<c></c>
2001	\$52,022,070.00	\$45,502,882.92	\$6,519,187.08
2002	50,413,140.00	44,654,086.29	5,759,053.71
2003	48,536,055.00	42,955,116.35	5,580,938.65
2004	46,658,970.00	41,285,832.26	5,373,137.74
2005	40,223,250.00	35,057,913.17	5,165,336.83
2006	33,787,530.00	29,334,653.42	4,452,876.58
2007	27,351,810.00	23,611,393.67	3,740,416.33

2008 21,452,400.00 18,424,443.93 3,027,956.07

\$ 7,060,000

13,714,432.49

2,374,867.51

</TABLE>

Schedule II-5 SCHEDULE B TO SCHEDULE II TO LEASE INTENDED FOR SECURITY

16,089,300.00

Supplemental Lessee Residual Guarantee

2009

Supplemental Lessee Residual Guarantee, which, if in force, equals 1% of Equipment Cost during the Base Lease Term and any Lease Renewal.

Schedule II-6 ANNEX I TO SCHEDULE II TO LEASE INTENDED FOR SECURITY

DESCRIPTION OF UNITS

C-Band Transponders Nos. 1, 2, 3, 4, 6 and 8 and Ku-Band Transponder No. 10 on the Galaxy XR satellite located at 123(Degree) West Longitude.

C-Band Transponders

Cost of Transponder

Cost of Warranty (a): Allocation of Legal Expense (b): Allocation of Owner-Trustee fees (c): Allocation of Structuring Fee (d):	650,000 23,333 1,667 6,667
Purchase Price for C-Band Transponder: Total Number of C-Band Transponders Purchased:	\$ 7,741,933 x 6
Aggregate Purchase Price for C-Band Transponders:	\$46,450,000
Ku-Band Transponder:	
Cost of Transponder Cost of Warranty (a): Allocation of Legal Expense (b): Allocation of Owner-Trustee fees (c): Allocation of Structuring Fee (d):	\$ 6,500,000 650,000 20,000 2,000 9,000
Purchase Price for Ku-Band Transponder: Total Number of Ku-Band Transponders Purchased:	\$ 7,181,000 x 1
Aggregate Purchase Price for Ku-Band Transponders:	\$ 7,181,000
Aggregate Purchase Price for All Transponders:	\$53,631,000

- (a) Warranty Fee is equal to 10% of \$6,500,000 for each transponder
- (b) Total Legal Expense is equal to \$160,000
- (c) Owner-Trustee expenses include \$2,000 acceptance fee and \$10,000 for out-of-pocket expenses and counsel fees. An additional \$2,000 annual administrative fee will be billed to the lessee.
- (d) Structuring Fee of 1% of the total purchase price or \$536,310 will be paid at date of closing. A portion of the structuring fee equal to \$100,000 was paid upon signing the Commitment Letter, \$49,000 will be included in the Aggregate Purchase Price for All Transponders while the remaining \$387,310 will be payable at closing.

Schedule II-7 ANNEX II TO SCHEDULE II TO LEASE INTENDED FOR SECURITY

RENTAL PAYMENTS

On the last day of each March, June, September and December or, if such day is not a Business Day, on the next Business Day (each, a "Payment Date"), Lessee shall make a rental payment to Owner Trustee equal to the sum of (i) interest accrued on the Balance Due during the period from, and including, the previous Payment Date (or, in the case of the first Payment Date, from and including the Delivery Date) to, but excluding, such Payment Date (each such period, an "Interest Period") at a rate equal to the sum of (A) the Adjusted

LIBOR Rate for such period, plus (B) 3.25% per annum, plus (ii) the Amortization Amount for such Payment Date. The interest described in the foregoing clause (i) shall be calculated on the basis of a 360 day year for the actual number of days elapsed. GECC shall notify Lessee and the Owner Trustee of the rental payment due on each Payment Date at least three (3) Business Days prior to such Payment Date, provided that the failure to so notify Lessee shall not affect Lessee's obligations to pay rent or any other amount hereunder.

"Adjusted LIBOR Rate" shall mean, for any Interest Period, (i) the LIBOR Rate for such Interest Period divided by (ii) one minus the Eurocurrency Reserve Rate in effect for such period.

"LIBOR Rate" for any Interest Period shall mean the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) appearing on Telerate Page 3750 (or successor page) as the London interbank offered rate for deposits in Dollars at approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period for a term of three (3) months. If for any reason such rate is not available, the term LIBOR Rate shall mean the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) appearing on Reuters Screen LIBO Page as the London interbank offered rate for deposits in Dollars at approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period for a term of three (3) months; provided, however, if more than one rate is specified on the Reuters Screen LIBO Page, the applicable rate shall be the arithmetic mean of all such rates.

"Amortization Amount" for any Payment Date shall mean the amount set forth opposite such Payment Date on Schedule A hereto.

Schedule II-8

<TABLE>

SCHEDULE A TO ANNEX II TO SCHEDULE II TO LEASE INTENDED FOR SECURITY

Rental Payment Schedule

<CAPTION>

Renta	al Pa Date	ayment	Total Rental Payment	Interest	Principal
<s></s>			<c></c>	<c></c>	<c></c>
Jun	30,	2000	\$1,646,471.70	\$1,244,239.20	\$402,232.50
Sep	30,	2000	1,637,139.91	1,234,907.41	402,232.50
Dec	31,	2000	1,627,808.11	1,225,575.61	402,232.50
Mar	31,	2001	1,618,476.32	1,216,243.82	402,232.50
Jun	30,	2001	1,609,144.52	1,206,912.02	402,232.50
Sep	30,	2001	1,599,812.73	1,197,580.23	402,232.50
Dec	31,	2001	1,590,480.94	1,188,248.44	402,232.50
Mar	31,	2002	1,581,149.14	1,178,916.64	402,232.50
Jun	30,	2002	1,638,856.10	1,169,584.85	469,271.25
Sep	30,	2002	1,627,969.00	1,158,697.76	469,271.25
Dec	31,	2002	1,617,081.91	1,147,810.66	469,271.25
Mar	31,	2003	1,606,194.82	1,136,923.57	469,271.25
Jun	30,	2003	1,595,307.73	1,126,036.48	469,271.25
Sep	30,	2003	1,584,420.63	1,115,149.38	469,271.25
Dec	31,	2003	1,573,533.54	1,104,262.29	469,271.25
Mar	31,	2004	1,562,646.45	1,093,375.20	469,271.25
Jun	30,	2004	2,691,418.10	1,082,488.10	1,608,930.00
Sep	30,	2004	2,654,090.93	1,045,160.93	1,608,930.00
Dec	31,	2004	2,616,763.75	1,007,833.75	1,608,930.00
Mar	31,	2005	2,579,436.58	970,506.58	1,608,930.00
Jun	30,	2005	2,542,109.40	933,179.40	1,608,930.00
Sep	30,	2005	2,504,782.22	895,852.22	1,608,930.00
Dec	31,	2005	2,467,455.05	858,525.05	1,608,930.00

Schedule TI-9

<TABLE>

Decision 1	D I			
	. Payment	al Dantal Danmant	T	D
<s></s>	Date Tot	2	Interest	Principal
		<c></c>	<c></c>	<c></c>
Mar 3	31, 2006	2,430,127.87	821,197.87	1,608,930.00
Jun 3	30, 2006	2,329,800.70	783,870.70	1,608,930.00
	,	, ,	·	
Sep 3	30, 2006	2,355,473.52	746,543.52	1,608,930.00
Dec 3	31, 2006	2,318,146.34	709,216.34	1,608,930.00
Mar 3	31, 2007	2,280,819.17	671,889.17	1,608,930.00
		0.400.444.40	504 554 00	4 454 050 50
Jun 3	30, 2007	2,109,414.49	634,561.99	1,474,852.50
g 2	7007	2 075 107 01	COO 345 41	1 474 050 50
sep s	30, 2007	2,075,197.91	600,345.41	1,474,852.50
Dec 3	31, 2007	2,040,981.34	566,128.84	1,474,852.50
DCC 3	2007	2,040,001.04	300,120.04	1,474,032.30
Mar 3	31, 2008	2,006,764.76	531,912.26	1,474,852.50
	-,		,	_,,
Jun 3	30, 2008	1,838,470.68	497,695.68	1,340,775.00
Sep 3	30, 2008	1,807,364.70	466,589.70	1,340,775.00
Dec 3	31, 2008	1,776,285.72	435,483.72	1,340,775.00
	31, 2009	1,745,152.74	404,377.74	1,340,775.00

 | | | |

Schedule II-10 ANNEX III TO SCHEDULE II TO LEASE INTENDED FOR SECURITY

RETURN REQUIREMENTS

Upon any return of the Units to Owner Trustee or Owner Participants, at Lessee's sole expense:

- 1. All Units and the Satellite in which they are contained (a) will have been maintained at all times in accordance with the Lease and, whether or not specified in the Lease, in accordance with the original manufacturer's suggested maintenance recommendations, (b) will be in the same condition as when delivered by the original manufacturer, regular and reasonable wear, tear and ordinary fuel use projected upon insertion into orbit excepted, (c) will be positioned properly in geosynchronous orbit at its orbital position 123(0) West longitude (or such other orbital slot as the Federal Communications Commission has approved or required), and (d) will be in full compliance with all domestic and international satellite regulations, and with recommendations or requirements of any insurance rating or classification agency or any government agency or department having jurisdiction necessary for maintenance of the Units' and satellite's classification or certification at the time of redelivery (it being agreed that Lessee, if taking advantage of any applicable grace period will nevertheless put the Units into such condition before redelivery hereunder).
- 2. At Lessee's expense, telemetry logs and other records will be inspected by and the Units and satellite tested a qualified person (the "Inspector") selected by the Owner Participants to verify that the Units and satellite are in the condition required by this Annex. If as a result of any such inspection and/or testing requested by the Inspector any item or component of any Unit or satellite is determined to be defective in any respect, then at the option of the Owner Participants, Lessee will pay to the Owner Participants an amount equal to (a) the then current market value of such item or component in place, in use and in the condition that would have been required if such item or component were redelivered, together with (b) additional rent at a per diem rate equal to the daily equivalent of the most recent rent payable under the Lease for the number of days beyond the end of the Term which, in the Owner Participants' reasonable opinion, will be required to replace such item or component.

The Term will be extended for any period necessary to conduct the inspection referred to above and to make repairs or replacements, appraisals and adjustments, if any, and, in the case of extension of the Term, Lessee will pay extended rent hereunder in respect of such period at the per diem rate referred to in the preceding paragraph. During any such period of extension, all of Lessee's obligations under the Lease will continue in full force and effect.

3. All regulatory and compliance certificates, permits and the like, if any, will be current and in full force and effect for transfer of ownership.

4. All systems will be operational and the Units will be in a condition suitable for continuous operation at rated capacity in commercial service.

Schedule II-12

<ARTICLE>

<LEGEND>

THIS SCHEDULE CONTAINS SUMMARY FINANCIAL INFORMATION EXTRACTED FROM THE INTERIM CONDENSED CONSOLIDATED STATEMENT OF INCOME FOR THE THREE AND SIX MONTH PERIODS ENDED JUNE 30, 2000 AND THE INTERIM CONDENSED CONSOLIDATED BALANCE SHEET AS OF JUNE 30, 2000 AND IS QUALIFIED IN ITS ENTIRETY BY REFERENCE TO SUCH FINANCIAL STATEMENTS.

</LEGEND> <CIK> 0000808461 <NAME> GENERAL COMMUNICATION, INC. <MULTIPLIER> 1,000 <C> <PERIOD-TYPE> 6-MOS <FISCAL-YEAR-END> DEC-31-2000 JAN-01-2000 <PERIOD-START> <PERIOD-END> JUN-30-2000 <CASH> 12,609 <SECURITIES> 47,159 <RECETVABLES> <ALLOWANCES> 3,452 4,602 <INVENTORY> <CURRENT-ASSETS> 77,157 479,320 <DEPRECIATION> 130,244 <TOTAL-ASSETS> 673,092 <CURRENT-LIABILITIES> 58,632 <BONDS> 377,254 <PREFERRED-MANDATORY> 21,658 <PREFERRED> <COMMON> 186,407 <OTHER-SE> (103)<TOTAL-LIABILITY-AND-EQUITY> 673,092 <SALES> <TOTAL-REVENUES> 139,703 <CGS> <TOTAL-COSTS> 59,295 <OTHER-EXPENSES> 74,402 <LOSS-PROVISION> 1,579 <INTEREST-EXPENSE> 19,412 <INCOME-PRETAX> (14,627)<INCOME-TAX> (5,603) <INCOME-CONTINUING> (9,024)<DISCONTINUED> Ω <EXTRAORDINARY> <CHANGES> <NET-INCOME> (9,024)<EPS-BASIC> (0.19)<EPS-DILUTED> (0.19)

5

</TABLE>